UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

Form 10-Q

(Mark One)

☑ QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended March 31, 2012

OR

□ TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the transition period from to

Commission File Number 001-12488

Powell Industries, Inc.

(Exact name of registrant as specified in its charter)

Delaware (State or other jurisdiction of incorporation or organization)

8550 Mosley Drive, Houston, Texas (Address of principal executive offices) 88-0106100 (I.R.S. Employer Identification No.)

77075-1180 (Zip Code)

Accelerated filer

Smaller reporting company

X

Registrant's telephone number, including area code: (713) 944-6900

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. 🛛 Yes \Box No

Indicate by check mark whether the registrant has submitted electronically and posted on its corporate Web site, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T (§232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit and post such files). 🛛 Yes 🗌 No

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, or a smaller reporting company. See the definitions of "large accelerated filer," "accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act. (Check one):

Large accelerated filer

 \square

Non-accelerated filer

Indicate by check mark whether the registrant is a shell company (as defined in Exchange Act Rule 12b-2). 🗆 Yes 🗵 No

At May 1, 2012, there were 11,773,920 outstanding shares of the registrant's common stock, par value \$0.01 per share.

 \Box (Do not check if a smaller reporting company)

POWELL INDUSTRIES, INC. AND SUBSIDIARIES TABLE OF CONTENTS

	Page
Part I — Financial Information	3
Item 1. Condensed Consolidated Financial Statements (unaudited)	3
Condensed Consolidated Balance Sheets	3
Condensed Consolidated Statements of Operations	4
Condensed Consolidated Statements of Cash Flows	5
Notes to Condensed Consolidated Financial Statements	6
Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations	18
Item 3. Quantitative and Qualitative Disclosures About Market Risk	23
Item 4. Controls and Procedures	24
Part II — Other Information	25
Item 1. Legal Proceedings	25
Item 1A. Risk Factors	26
Item 5. Other Information	26
Item 6. Exhibits	27

<u>Signatures</u>

2

PART I - FINANCIAL INFORMATION

Item 1. Condensed Consolidated Financial Statements

POWELL INDUSTRIES, INC. AND SUBSIDIARIES **Condensed Consolidated Balance Sheets** (In thousands, except share and per share data)

	March 31, 2012 (Unaudited)	September 30, 2011
ASSETS	(,	
Current Assets:		
Cash and cash equivalents	\$ 128,065	\$ 123,466
Cash held in escrow	75	1,000
Accounts receivable, less allowance for doubtful accounts of \$1,043 and \$391, respectively	97,694	109,317
Costs and estimated earnings in excess of billings on uncompleted contracts	77,768	51,568
Inventories, net	42,760	36,640
Income taxes receivable	553	4,071
Deferred income taxes	4,011	3,580
Prepaid expenses and other current assets	4,208	7,040
Total Current Assets	355,134	336,682
Property, plant and equipment, net	73,774	59,637
Goodwill	1,003	1,003
Intangible assets, net	14,494	15,847
Other assets	7,511	8,507
Total Assets	\$ 451,916	\$ 421,676
LIABILITIES AND EQUITY		
Current Liabilities:		
Current maturities of long-term debt and capital lease obligations	\$ 916	\$ 1,140
Income taxes payable	2,360	881
Accounts payable	60,234	56,893
Accrued salaries, bonuses and commissions	19,176	22,314
Billings in excess of costs and estimated earnings on uncompleted contracts	68,617	44,523
Accrued product warranty	4,783	4,603
Other accrued expenses	5,607	7,370
Total Current Liabilities	161,693	137,724
Long-term debt and capital lease obligations, net of current maturities	3,712	4,301
Deferred compensation	2,955	3,242
Other liabilities	1,082	1,066
Total Liabilities	169,442	146,333
Commitments and Contingencies (Note I)		
Equity:		
Stockholders' Equity:		
Preferred stock, par value \$.01; 5,000,000 shares authorized; none issued		
Common stock, par value \$.01; 30,000,000 shares authorized; 11,772,320 and 11,752,393 shares issued and		
outstanding, respectively	118	117
Additional paid-in capital	35,187	34,343
Retained earnings	247,921	242,254
Accumulated other comprehensive income (loss)	(752)	(1,371)
Total Stockholders' Equity	282,474	275,343
Total Liabilities and Equity	\$ 451,916	\$ 421,676

The accompanying notes are an integral part of these condensed consolidated financial statements.

POWELL INDUSTRIES, INC. AND SUBSIDIARIES Condensed Consolidated Statements of Operations (Unaudited) (In thousands, except per share data)

		Three Months Ended				Six Mont	hs Ende	d
	Mar	ch 31, 2012	Ma	rch 31, 2011	Ma	rch 31, 2012	Ma	rch 31, 2011
Revenues	\$	181,486	\$	125,111	\$	338,942	\$	249,785
Cost of goods sold		147,249		100,234		284,327		199,043
Gross profit		34,237		24,877		54,615		50,742
Selling, general and administrative expenses		21,523		21,538		41,286		42,466
Amortization of intangible assets		704		1,254		1,407		2,421
Operating income		12,010		2,085		11,922		5,855
Gain on sale of investment		_		(1,229)		_		(1,229)
Interest expense		68		94		144		208
Interest income		(34)		(62)		(63)		(107)
Income before income taxes		11,976		3,282		11,841		6,983
Income tax provision		4,565		1,549		6,175		2,818
Net income	\$	7,411	\$	1,733	\$	5,666	\$	4,165
Earnings per share:								
Basic	\$	0.63	\$	0.15	\$	0.48	\$	0.36
Diluted	\$	0.63	\$	0.15	\$	0.48	\$	0.35
Weighted average shares:								
Basic		11,762		11,666		11,765		11,653
Diluted		11,835		11,799	_	11,836		11,787

The accompanying notes are an integral part of these condensed consolidated financial statements.

POWELL INDUSTRIES, INC. AND SUBSIDIARIES Condensed Consolidated Statements of Cash Flows (Unaudited) (In thousands)

	Six Mon March 31, 2012	ths Ended <u>March 31, 2011</u>
Operating Activities:		
Net income	\$ 5,666	\$ 4,165
Adjustments to reconcile net income to net cash provided by operating activities:		
Depreciation	5,020	5,037
Amortization	1,413	2,456
Stock-based compensation	386	1,088
Bad debt expense	636	606
Deferred income taxes	304	(782)
Gain on sale of investment in joint venture	—	(1,229)
Changes in operating assets and liabilities:		
Accounts receivable	11,755	(10,324)
Costs and estimated earnings in excess of billings on uncompleted contracts	(25,911)	2,172
Inventories	(5,964)	(739)
Prepaid expenses and other current assets	6,391	3,176
Accounts payable and income taxes payable	4,632	656
Accrued liabilities	(4,925)	(12,927)
Billings in excess of costs and estimated earnings on uncompleted contracts	24,015	15,884
Other	(41)	(749)
Net cash provided by operating activities	23,377	8,490
Investing Activities:		
Proceeds from sale of fixed assets	70	60
Purchases of property, plant and equipment	(18,879)	(2,667)
Increase in cash held in escrow	(75)	
Decrease in cash held in escrow	1,000	
Proceeds from sale of investment in joint venture	—	1,229
Net cash used in investing activities	(17,884)	(1,378)
Financing Activities:		
Borrowings on Canadian revolving line of credit	7,352	2,619
Payments on Canadian revolving line of credit	(7,352)	(2,627)
Payments on industrial development revenue bonds	(400)	(400)
Short-term and other financing	19	(189)
Net cash used in financing activities	(381)	(597)
Net increase in cash and cash equivalents	5,112	6,515
Effect of exchange rate changes on cash and cash equivalents	(513)	188
Cash and cash equivalents at beginning of period	123,466	115,353
Cash and cash equivalents at end of period	\$ 128,065	\$ 122,056

The accompanying notes are an integral part of these condensed consolidated financial statements

POWELL INDUSTRIES, INC. AND SUBSIDIARIES

Notes to Condensed Consolidated Financial Statements (Unaudited)

A. OVERVIEW AND SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

Overview

Powell Industries, Inc. (we, us, our, Powell or the Company) was incorporated in the state of Delaware in 2004 as a successor to a Nevada company incorporated in 1968. The Nevada corporation was the successor to a company founded by William E. Powell in 1947, which merged into the Company in 1977. Our major subsidiaries, all of which are wholly-owned, include: Powell Electrical Systems, Inc.; Transdyn, Inc.; Powell Industries International, Inc.; Switchgear & Instrumentation Limited (S&I) and Powell Canada Inc.

We develop, design, manufacture and service custom engineered-to-order equipment and systems for the management and control of electrical energy and other critical processes. Headquartered in Houston, Texas, we serve the transportation, environmental, energy, industrial and utility industries.

Basis of Presentation

These unaudited condensed consolidated financial statements include the accounts of Powell and its wholly-owned subsidiaries. All significant intercompany accounts and transactions have been eliminated in consolidation.

These unaudited condensed consolidated financial statements have been prepared pursuant to the rules of the Securities and Exchange Commission (SEC). Certain information and footnote disclosures, normally included in annual financial statements prepared in accordance with accounting principles generally accepted in the United States (U.S. GAAP), have been condensed or omitted pursuant to those rules and regulations. We believe that the disclosures made are adequate to make the information presented not misleading. In the opinion of management, all adjustments, consisting only of normal recurring adjustments, necessary to fairly state the financial position, results of operations and cash flows with respect to the interim consolidated financial statements have been included. The results of operations for the interim periods are not necessarily indicative of the results for the entire fiscal year. The year-end condensed balance sheet data was derived from audited financial statements, but does not include all disclosures required by U.S. GAAP.

These unaudited condensed consolidated financial statements should be read in conjunction with the audited consolidated financial statements and notes thereto of Powell and its subsidiaries included in Powell's Annual Report on Form 10-K for the year ended September 30, 2011, which was filed with the SEC on December 12, 2011.

Use of Estimates

The preparation of financial statements in conformity with U.S. GAAP requires management to make estimates and assumptions that affect the amounts reported in the condensed consolidated financial statements and accompanying footnotes. The most significant estimates used in our financial statements affect revenue and cost recognition for construction contracts, the allowance for doubtful accounts, provision for excess and obsolete inventory, goodwill and other intangible assets, self-insurance, warranty accruals, income taxes and estimates related to acquisition valuations. The amounts recorded for insurance claims, warranties, legal, income taxes and other contingent liabilities require judgments regarding the amount of expenses that will ultimately be incurred. We base our estimates on historical experience and on various other assumptions, as well as the specific circumstances surrounding these contingent liabilities, in evaluating the amount of liability that should be recorded. Estimates may change as new events occur, additional information becomes available or operating environments change. Actual results may differ from our estimates.

New Accounting Standards

In January 2010, the Financial Accounting Standards Board (FASB) issued updated guidance to amend the disclosure requirements related to recurring and nonrecurring fair value measurements. This update requires new disclosures about significant transfers of assets and liabilities between Level 1 and Level 2 of the fair value hierarchy (including the reasons for these transfers) and the reasons for any transfers in or out of Level 3. This update also requires a reconciliation of recurring Level 3 measurements about purchases, sales, issuances and settlements on a gross basis. In addition to these new disclosure requirements, this update clarifies certain existing disclosure requirements. For example, this update clarifies that reporting entities are required to provide fair value

measurement disclosures for each class of assets and liabilities, rather than each major category of assets or liabilities. This update also clarifies the requirement for entities to disclose information about both the valuation techniques and inputs used in estimating Level 2 and Level 3 fair value measurements. This update became effective for us with the interim and annual reporting period beginning after December 15, 2009, our fiscal year 2011, except for the requirement to provide the Level 3 activity of purchases, sales, issuances and settlements on a gross basis, which became effective for us with the interim and annual reporting period beginning after December 15, 2010, our fiscal year 2012. We are not required to provide the amended disclosures for any previous periods presented for comparative purposes. Other than requiring additional disclosures, adoption of this update did not have a material impact on our consolidated financial statements.

In May 2011, the FASB issued accounting guidance related to fair value measurement, which amends current guidance to achieve common fair value measurement and disclosure requirements in U.S. GAAP and International Financial Reporting Standards. This guidance generally represents clarification of fair value measurement standards, but also includes instances where a particular principle or requirement for measuring fair value or disclosing information about fair value measurements has changed. This guidance is effective for fiscal years, and interim periods within those years, beginning after December 15, 2011. We will adopt this guidance for our fiscal year beginning October 1, 2012. We do not expect this pronouncement to have a material effect on our consolidated financial statements.

In June 2011, the FASB issued new accounting guidance on the presentation of comprehensive income in financial statements. Entities are required to present total comprehensive income either in a single, continuous statement of comprehensive income or in two separate, but consecutive, statements. Under the single-statement approach, entities must include the components of net income, a total for net income, the components of other comprehensive income and a total for comprehensive income. Under the two-statement approach, entities must report an income statement and, immediately following, a statement of other comprehensive income. This new guidance will be effective for us beginning October 1, 2012, and will have financial presentation changes only.

In September 2011, the FASB issued new accounting guidance which simplifies how an entity is required to test goodwill for impairment. Under this guidance, an entity would be allowed to first assess qualitative factors to determine whether it is necessary to perform the two-step quantitative goodwill impairment test. An entity would not be required to calculate the fair value of a reporting unit unless the entity determines, based on a qualitative assessment, that it is more likely than not that its fair value is less than its carrying amounts. This new guidance includes a number of factors to consider in conducting the qualitative assessment. This guidance is effective for annual and interim goodwill impairment tests performed for fiscal years beginning after December 15, 2011, our Fiscal 2013. Early adoption is permitted; however, we will not adopt this guidance until October 1, 2012. This guidance is not expected to have a material impact on our reported results of operations or financial position.

Subsequent Events

We evaluated subsequent events through the time of filing this Quarterly Report on Form 10-Q. No significant events occurred subsequent to the balance sheet or prior to the filing of this report that would have a material impact on our consolidated financial statements or results of operations.

B. FAIR VALUE MEASUREMENTS

We measure certain financial assets and liabilities at fair value. Fair value is defined as an "exit price" which represents the amount that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants as of the measurement date. As such, fair value is a market-based measurement that should be determined based on assumptions that market participants would use in valuing an asset or liability. The accounting guidance requires the use of valuation techniques to measure fair value that maximize the use of observable inputs and minimize the use of unobservable inputs. As a basis for considering such assumptions and inputs, a fair value hierarchy has been established which identifies and prioritizes three levels of inputs to be used in measuring fair value.

The three levels of the fair value hierarchy are as follows:

Level 1 — Observable inputs such as quoted prices (unadjusted) in active markets for identical assets or liabilities.

Level 2 — Inputs other than the quoted prices in active markets that are observable either directly or indirectly, including: quoted prices for similar assets and liabilities in active markets; quoted prices for identical or similar assets and liabilities in markets that are not active or other inputs that are observable or can be corroborated by observable market data.

Level 3 — Unobservable inputs that are supported by little or no market data and require the reporting entity to develop its own assumptions.

The following table summarizes the fair value of our assets and liabilities that were accounted for at fair value on a recurring basis as of March 31, 2012 (in thousands):

	Fair Value Measurements at March 31, 2012									
	Active Ident	Quoted Prices in Significant Other Active Markets for Observable Identical Assets Inputs (Level 1) (Level 2)		Inputs		Observable Inputs		ificant servable puts vel 3)		r Value at ch 31, 2012
Assets		<u> </u>								
Cash equivalents	\$	45,648	\$		\$		\$	45,648		
Total	\$	45,648	\$		\$	_	\$	45,648		
Liabilities										
Foreign currency forward contracts	\$		\$	1	\$		\$	1		
Total	\$		\$	1	\$		\$	1		

The following table summarizes the fair value of our assets and liabilities that were accounted for at fair value on a recurring basis as of September 30, 2011 (in thousands):

		Fair Value Measurements at September 30, 2011							
	Active Iden	ed Prices in Markets for tical Assets Level 1)	Significant Other Observable Inputs (Level 2)		Significant Unobservable Inputs (Level 3)			Value at Der 30, 2011	
Assets	<u> </u>	<u> </u>							
Cash equivalents	\$	65,792	\$		\$		\$	65,792	
Total	\$	65,792	\$	_	\$		\$	65,792	
Liabilities									
Foreign currency forward contracts	\$	—	\$	—	\$	—	\$	—	
Total	\$		\$		\$		\$		

Cash equivalents, primarily funds held in money market savings instruments, are reported at their current carrying value which approximates fair value due to the short-term nature of these instruments and are included in cash and cash equivalents in our Condensed Consolidated Balance Sheets.

Foreign currency forward contracts are valued using an income approach which consists of a discounted cash flow model that takes into account the present value of future cash flows under the terms of the contracts using observable market spot and forward rates as of our reporting date, and are included in Level 2 inputs in the above table. We use these derivative instruments to mitigate non-functional currency transaction exposure on certain contracts with customers and vendors. We mitigate derivative credit risk by transacting with highly rated counterparties. We have evaluated the credit and non-performance risks associated with our derivative counterparties and believe them to be insignificant at March 31, 2012. All contracts are recorded at fair value and marked-to-market at the end of each reporting period, with unrealized gains and losses being included in accumulated other comprehensive income on our Condensed Consolidated Balance Sheets for that period. See Note H for further discussion regarding our derivative instruments.

C. EARNINGS PER SHARE

The following table sets forth the computation of basic and diluted earnings per share (in thousands, except per share data):

	Three Mor Marc	nths Ended ch 31,		ths Ended ch 31,
	2012	2011	2012	2011
Numerator:				
Net income attributable to Powell Industries, Inc.	\$ 7,411	\$ 1,733	\$ 5,666	\$ 4,165
Denominator:				
Weighted average basic shares	11,762	11,666	11,765	11,653
Dilutive effect of stock options, restricted stock and restricted stock units	73	133	71	134
Weighted average diluted shares with assumed conversions	11,835	11,799	11,836	11,787
Net earnings per share:				
Basic	\$ 0.63	\$ 0.15	\$ 0.48	\$ 0.36
Diluted	\$ 0.63	\$ 0.15	\$ 0.48	\$ 0.35

All options were included in the computation of diluted earnings per share for the three and six months ended March 31, 2012 and 2011, respectively, as the options' exercise price was less than the average market price of our common stock.

D. DETAIL OF SELECTED BALANCE SHEET ACCOUNTS

Allowance for Doubtful Accounts

Activity in our allowance for doubtful accounts receivable consisted of the following (in thousands):

	Three Mo Mare	nths Ended h 31,		ths Ended ch 31,
	2012	2011	2012	2011
Balance at beginning of period	\$ 588	\$ 1,188	\$ 391	\$ 907
Increase to bad debt expense	451	318	636	606
Deductions for uncollectible accounts written off, net of recoveries		(4)	—	(3)
Increase (decrease) due to foreign currency translation	4	6	16	(2)
Balance at end of period	\$1,043	\$ 1,508	\$1,043	\$1,508

Warranty Accrual

Activity in our product warranty accrual consisted of the following (in thousands):

		nths Ended ch 31,	Six Mont Marc	hs Ended h 31,
	2012	2011	2012	2011
Balance at beginning of period	\$ 4,396	\$ 5,845	\$ 4,603	\$ 5,929
Warranty expense	917	917 452		863
Deductions for warranty charges	(590)	(628)	(1,240)	(1,209)
Increase due to foreign currency translation	60	94	34	180
Balance at end of period	\$ 4,783	\$ 5,763	\$ 4,783	\$ 5,763

Inventories

The components of inventories are summarized below (in thousands):

	March 31, 2012	Sep	tember 30, 2011
Raw materials, parts and subassemblies	\$44,274	\$	38,400
Work-in-progress	7,065		5,892
Provision for excess and obsolete inventory	(8,579)		(7,652)
Total inventories	\$42,760	\$	36,640

Cost and Estimated Earnings on Uncompleted Contracts

The components of costs and estimated earnings and related amounts billed on uncompleted contracts are summarized below (in thousands):

	March 31, 2012	September 30, 2011
Costs incurred on uncompleted contracts	\$465,436	\$ 475,525
Estimated earnings	139,024	131,367
	604,460	606,892
Less: Billings to date	595,309	599,847
Net underbilled position	\$ 9,151	\$ 7,045
Included in the accompanying balance sheets under the following captions:		
Costs and estimated earnings in excess of billings on uncompleted contracts —		
underbilled	\$ 77,768	\$ 51,568
Billings in excess of costs and estimated earnings on uncompleted contracts —		
overbilled	(68,617)	(44,523)
Net underbilled position	\$ 9,151	\$ 7,045

E. OTHER INTANGIBLE ASSETS

Intangible assets balances, subject to amortization, at March 31, 2012 and September 30, 2011 consisted of the following (in thousands):

	March 31, 2012		September 30, 2011					
	Gross Carrying Value		cumulated 1ortization	Net Carrying Value	Gross Carrying Value		cumulated 1ortization	Net Carrying Value
Supply agreement								
Balance, beginning of period	\$17,580	\$	(6,052)	\$11,528	\$17,580	\$	(4,881)	\$12,699
Amortization			(586)	(586)			(1,171)	(1,171)
Balance, end of period	\$17,580	\$	(6,638)	\$10,942	\$17,580	\$	(6,052)	\$11,528
Purchased technology								
Balance, beginning of period	\$11,747	\$	(7,759)	\$ 3,988	\$10,272	\$	(6,318)	\$ 3,954
Acquisition	—			—	1,500		—	1,500
Amortization	—		(899)	(899)	_		(1,689)	(1,689)
Foreign currency translation	106		87	193	(25)		248	223
Balance, end of period	\$11,853	\$	(8,571)	\$ 3,282	\$11,747	\$	(7,759)	\$ 3,988
Non-compete agreements								
Balance, beginning of period	\$ 4,170	\$	(4,170)	\$ —	\$ 5,365	\$	(3,666)	\$ 1,699
Amortization	—		—	—	—		(920)	(920)
Foreign currency translation	—		—	—	(35)			(35)
Impairment (a)					(1,160)		416	(744)
Balance, end of period	\$ 4,170	\$	(4,170)	\$ —	\$ 4,170	\$	(4,170)	\$ —
Trade name								
Balance, beginning of period	\$ 1,098	\$	(767)	\$ 331	\$ 5,437	\$	(938)	\$ 4,499
Amortization	—		(86)	(86)	—		(583)	(583)
Foreign currency translation	26		(1)	25	(124)		(1)	(125)
Impairment (a)					(4,215)		755	(3,460)
Balance, end of period	\$ 1,124	\$	(854)	\$ 270	\$ 1,098	\$	(767)	\$ 331
Customer relationships								
Balance, beginning of period	\$ —	\$	_	\$ —	\$ 3,479	\$	(198)	\$ 3,281
Amortization					_		(205)	(205)
Foreign currency translation	—			—	(103)			(103)
Impairment (a)					(3,376)		403	(2,973)
Balance, end of period	\$ —	\$	_	\$ —	\$ —	\$	_	\$ —
Total	\$34,727	\$	(20,233)	\$14,494	\$34,595	\$	(18,748)	\$15,847

(a) Represents the \$7.2 million impairment charge recorded in fiscal 2011 related to the intangible assets of Powell Canada.

All goodwill and intangible assets disclosed above are reported in our Electrical Power Products business segment.

Amortization of intangible assets recorded for the six months ended March 31, 2012 and 2011 was approximately \$1.4 million and \$2.4 million, respectively.

F. COMPREHENSIVE INCOME

Comprehensive income was as follows (in thousands):

		Three Months Ended Six Month March 31, Marc		
	2012 2011		2012	2011
Net income attributable to Powell Industries, Inc.	\$ 7,411	\$ 1,733	\$5,666	\$4,165
Unrealized gain on foreign currency translation, net of tax	826	1,712	618	2,072
Unrealized loss on derivative contracts, net of tax	—	(10)		(3)
Comprehensive income	\$8,237	\$ 3,435	\$6,284	\$6,234

G. LONG-TERM DEBT

Long-term debt consisted of the following (in thousands):

	March 31, 2012	September 30, 2011
Industrial development revenue bonds	\$ 4,000	\$ 4,400
Capital lease obligations	628	1,041
Subtotal long-term debt and capital lease obligations	4,628	5,441
Less current portion	(916)	(1,140)
Total long-term debt and capital lease obligations	\$ 3,712	\$ 4,301

US Revolver

In March 2012, we amended our existing credit agreement (Amended Credit Agreement) with a major domestic bank. This amendment to our credit facility was made to increase the dollar limit on capital expenditures to allow us to support our continued expansions, including the Canadian Oil Sands and offshore production markets. The Amended Credit Agreement provides for a \$75.0 million revolving credit facility (US Revolver). Obligations are collateralized by the stock of certain of our subsidiaries.

The interest rate for amounts outstanding under the Amended Credit Agreement for the US Revolver is a floating rate based upon the higher of the Federal Funds Rate plus 0.5%, or the bank's prime rate. Once the applicable rate is determined, a margin ranging from negative 0.5% to 1.75%, as determined by our consolidated leverage ratio, is added to the applicable rate.

The US Revolver provides for the issuance of letters of credit which reduce the amounts which may be borrowed under the revolver. The amount available under the US Revolver was reduced by \$42.7 million for our outstanding letters of credit at March 31, 2012.

There were no borrowings outstanding under the US Revolver as of March 31, 2012. Amounts available under the US Revolver were \$32.3 million at March 31, 2012. The US Revolver expires on December 31, 2016.

The Amended Credit Agreement contains certain restrictive and maintenance-type covenants, including restrictions on our ability to pay dividends, as well as restrictions on the amount of capital expenditures allowed. It also contains financial covenants defining various financial measures and the levels of these measures with which we must comply, as well as a "material adverse change" clause. A "material adverse change" is defined as a material change in our operations, business, properties, liabilities or condition (financial or otherwise) or a material impairment of our ability to perform our obligations under our credit agreements.

The Amended Credit Agreement is collateralized by a pledge of 100% of the voting capital stock of each of our domestic subsidiaries and 66% of the voting capital stock of each non-domestic subsidiary, excluding Powell Canada. The Amended Credit Agreement provides for customary events of default and carries cross-default provisions with other existing debt agreements. If an event of default (as defined in the Amended Credit Agreement) occurs and is continuing, on the terms and subject to the conditions set forth in the Amended Credit Agreement, amounts outstanding under the Amended Credit Agreement may be accelerated and may become immediately due and payable. As of March 31, 2012, we were in compliance with all of the financial covenants of the Amended Credit Agreement.

Canadian Revolver

On December 15, 2009, we entered into a credit agreement with a major international bank (the Canadian Facility) to finance the acquisition of Powell Canada and provide additional working capital support for our operations in Canada. In March 2012, we reduced the Canadian Facility from a \$20 million CAD (approximately \$20.1 million) revolving credit facility (the Canadian Revolver) to a \$10 million CAD (approximately \$10.0 million), and eliminated the restrictions on amounts which may be borrowed based on a borrowing base calculation.

The Canadian Revolver provides for the issuance of letters of credit which reduce the amounts which may be borrowed under the Canadian Revolver. As of March 31, 2012, there were no letters of credit outstanding under the Canadian Revolver.

There were no borrowings outstanding under the Canadian Revolver, and \$10.0 million was available at March 31, 2012. The Canadian Facility expires on February 28, 2015. The interest rate for amounts outstanding under the Canadian Revolver is a floating interest rate based upon either the Canadian Prime Rate, or the lender's US Bank Rate. Once the applicable rate is determined, a margin of 0.375% to 1.125%, as determined by our consolidated leverage ratio, is added to the applicable rate.

The principal financial covenants are consistent with those in our Amended Credit Agreement. The Canadian Facility contains a "material adverse effect" clause. A "material adverse effect" is defined as a material change in the operations of Powell or Powell Canada in relation to our financial condition, property, business operations, expected net cash flows, liabilities or capitalization.

The Canadian Facility is secured by the assets of our Canadian operations and provides for customary events of default and carries cross-default provisions with our existing debt agreements. If an event of default (as defined in the Canadian Facility) occurs and is continuing, on the terms and subject to the conditions set forth in the Canadian Facility, amounts outstanding under the Canadian Facility may be accelerated and may become immediately due and payable. As of March 31, 2012, we were in compliance with all of the financial covenants of the Canadian Facility, and no borrowings were outstanding under this facility at March 31, 2012.

Industrial Development Revenue Bonds

We borrowed \$8.0 million in October 2001 through a loan agreement funded with proceeds from tax-exempt industrial development revenue bonds (Bonds). These Bonds were issued by the Illinois Development Finance Authority and were used for the completion of our Northlake, Illinois facility. Pursuant to the Bond issuance, a reimbursement agreement between us and a major domestic bank required an issuance by the bank of an irrevocable direct-pay letter of credit (Bond LC), as collateral, to the Bonds' trustee to guarantee payment of the Bonds' principal and interest when due. The Bond LC is subject to both early termination and extension provisions customary to such agreements, as well as various covenants, for which we were in compliance at March 31, 2012. While the Bonds mature in 2021, the reimbursement agreement requires annual redemptions of \$400,000 that commenced on October 25, 2002. A sinking fund is used for the redemption of the Bonds. At March 31, 2012, the balance in the restricted sinking fund was approximately \$234,000 and was recorded in cash and cash equivalents. The Bonds bear interest at a floating rate determined weekly by the Bonds' remarketing agent, which was the underwriter for the Bonds and is an affiliate of the bank. This interest rate was 0.55% per year on March 31, 2012.

H. DERIVATIVE INSTRUMENTS AND HEDGING STRATEGIES

We operate in various countries and have operations in the United Kingdom and Canada. These international operations expose us to market risk associated with foreign currency exchange rate fluctuations. We have entered into certain forward contracts to hedge the risk of certain foreign currency rate fluctuations. To the extent we choose to manage volatility associated with the net exposures, we enter into various financial transactions which we account for using the applicable accounting guidance for derivative instruments and hedging activities. Our objective is to hedge the variability in forecasted cash flow due to the foreign currency risk associated with certain long-term sales. As of March 31, 2012, we held only derivatives that were designated as cash flow hedges related to the U.S. Dollar/British Pound Sterling exchange rate.

All derivatives are recognized on the Condensed Consolidated Balance Sheets at their fair value and classified based on the instrument's maturity date. The total notional amount of outstanding derivatives as of March 31, 2012 was \$0.9 million.

The following table presents the fair value of derivative instruments included within the Condensed Consolidated Balance Sheets as of March 31, 2012:

Asset Derivatives	Asset Derivatives		
Balance Sheet Location	Fair Value	Balance Sheet Location	Fair Value
	(in the	ousands)	
Prepaid expenses and other current assets	\$ —	Other accrued expenses	\$ 1
	<u>\$</u>	outer accurace expenses	\$ 1
	Balance Sheet Location	Balance Sheet Location Fair Value (in the Value) Prepaid expenses and other current assets \$—	Balance Sheet Location Fair Value Balance Sheet Location (in thousands) Prepaid expenses and other current assets \$—Other accrued expenses

There were no derivatives outstanding at September 30, 2011.

The following table presents the amounts affecting the Condensed Consolidated Statements of Operations for the three and six month periods ended March 31, 2012:

	Amount of C Recognized Comprehensiv Deriva	in Other e Income on		Amount of C Reclassifi Accumulat Comprehens into Inc	ed from ed Other ive Income
	Three Months	Six Months	Location of Gain (Loss)	Three Months	Six Months
	Ended	Ended	Reclassified from Accumulated	Ended	Ended
	March 31,	March 31,	Other comprehensive Income	March 31,	March 31,
Derivatives designated:	2012	2012	into Income	2012	2012
	(in thou	sands)		(in thou	sands)
Derivatives designated as cash flow hedges:					
Foreign exchange forwards	<u>\$ </u>	<u>\$ </u>	Revenues	<u>\$ (30)</u>	<u>\$ (1</u>)
Total designated cash flow hedges	<u>\$ </u>	<u>\$ </u>		\$ (30)	<u>\$ (1)</u>

The following table presents the amounts affecting the Condensed Consolidated Statements of Operations for the three and six month periods ended March 31, 2011:

	Recogniz Comprehen	f Gain (Loss) ed in Other sive Income on vatives ¹		Amount of C Reclassifi Accumulat Comprehens into Inc	ed from ed Other ive Income
Derivatives designated:	Three Months Ended March 31, 2011	Six Months Ended March 31, 2011	Location of Gain (Loss) Reclassified from Accumulated Other comprehensive Income into Income	Three Months Ended March 31, 2011	Six Months Ended March 31, 2011
<u> </u>		ousands)	into income	(in thou	
Derivatives designated as cash flow hedges:					
Foreign exchange forwards	<u>\$ (16)</u>	<u>\$ (5</u>)	Revenues	<u>\$ 22</u>	\$ 42
Total designated cash flow hedges	<u>\$ (16)</u>	<u>\$ (5)</u>		\$ 22	\$ 42

For the three and six month periods ended March 31, 2011, we recorded in revenues an immaterial amount of ineffectiveness from cash flow hedges.

Refer to Note B for a description of how the above financial instruments are valued in accordance with the fair value measurement accounting guidance for the three and six month periods ended March 31, 2012.

Cash Flow Hedges

The purpose of our foreign currency hedging activities is to protect us from the risk that the eventual cash flows resulting from transactions that are denominated in currencies other than the U.S. Dollar will be adversely affected by changes in exchange rates. We are currently hedging our exposure to the reduction in value of forecasted foreign currency cash flows through foreign currency forward agreements through October 31, 2012, for transactions denominated in the British Pound Sterling.

All changes in the fair value of outstanding cash flow hedge derivatives, except the ineffective portion, are recorded in accumulated other comprehensive income, until net income is affected by the variability of cash flows of the hedged transaction, or until it is probable that the forecasted transaction will not occur. In most cases, amounts recorded in accumulated other comprehensive income will be released to net income some time after the maturity of the related derivative. The Condensed Consolidated Statements of Operations' classification of effective hedge results is the same as that of the underlying exposure. Results of hedges of revenue and product costs are recorded in revenue and costs of sales, respectively, when the underlying hedged transaction affects net income. Results of hedges of selling and administrative expense, if any, are recorded together with those costs when the related expense is recorded. In addition, any ineffective portion of the changes in the fair value of the derivatives designated as cash flow hedges are reported in the Condensed Consolidated Statements of Operations as the changes occur.

As of March 31, 2012, there were no deferred net losses (net of tax) on outstanding derivatives recorded in accumulated other comprehensive income expected to be reclassified to net income during the next 12 months as a result of underlying hedged transactions being recorded in net income. Actual amounts ultimately reclassified to net income are dependent on the exchange rates in effect when the derivative contracts that are currently outstanding mature. As of March 31, 2012, the maximum term over which we are hedging exposure to the variability of cash flows for our forecasted and recorded transactions is seven months. For the three and six months ended March 31, 2012, we recorded in selling, general and administrative expense an immaterial amount of ineffectiveness from cash flow hedges.



Credit Risk

We are exposed to credit-related losses in the event of non-performance by counterparties to hedging instruments. In recent years, the ability of financial counterparties to perform under financial instruments has become less certain. We attempt to take into account the financial viability of counterparties in both valuing the instruments and determining their effectiveness as hedging instruments. If a counterparty was unable to perform, our ability to qualify for hedging certain transactions would be compromised and the realizable value of the financial instruments would be uncertain. As a result, our results of operations and cash flows would be impacted.

I. COMMITMENTS AND CONTINGENCIES

Letters of Credit and Bonds

Certain customers require us to post bank letter of credit guarantees or performance bonds issued by a surety. These guarantees and performance bonds assure that we will perform under the terms of our contract. In the event of default, the counterparty may demand payment from the bank under a letter of credit or performance by the surety under a performance bond. To date, there have been no significant expenses related to either for the periods reported. We were contingently liable for secured and unsecured letters of credit of \$42.7 million as of March 31, 2012, under our US Revolver. We also had performance and maintenance bonds totaling \$217.5 million that were outstanding with our sureties, with additional bonding capacity of \$382.5 million available with our sureties, at March 31, 2012.

We have a facility agreement (Facility Agreement) between S&I and a large international bank. This \$12.0 million Facility Agreement provides S&I the ability to enter into forward exchange contracts, currency options and performance bonds. At March 31, 2012, we had outstanding a total of \$10.7 million of guarantees under this Facility Agreement.

The Facility Agreement provides for financial covenants, customary events of default and carries cross-default provisions with our Amended Credit Facility. If an event of default (as defined in the Facility Agreement) occurs and is continuing, on the terms and subject to the conditions set forth in the Facility Agreement, obligations outstanding under the Facility Agreement may be accelerated and may become or be declared immediately due and payable.

Litigation

We are involved in various legal proceedings, claims and other disputes arising in the ordinary course of business which, in general, are subject to uncertainties and the outcomes are not predictable. Although we can give no assurance about the outcome of pending or threatened litigation and the effect such outcomes may have on us, management believes that any ultimate liability resulting from the outcome of such proceedings, to the extent not otherwise provided or covered by insurance, will not have a material adverse effect on our consolidated financial position or result of operations or liquidity.

J. STOCK-BASED COMPENSATION

Refer to our Annual Report on Form 10-K for the fiscal year ended September 30, 2011 for a full description of our existing stock-based compensation plans.

Restricted Stock Units

In October 2010, we granted 34,566 restricted stock units (RSUs) with a fair value of \$30.79 per unit to certain officers and key employees of the Company. An additional 4,482 RSUs were granted in October 2010, with a fair value of \$32.12 per unit. The RSUs vest over a three-year period from their date of issuance. The fair value of the RSUs was based on the closing price of our common stock as reported on the NASDAQ Global Market (NASDAQ) on the grant dates. Sixty-percent of the actual amount of the RSUs earned will be based on the cumulative earnings as reported relative to the three-year performance cycle which began October 1 of the year granted, and ranges from 0% to 150% of the target RSUs granted. The remaining forty-percent of the RSUs are time-based and vest over a three-year period. The RSUs do not have voting rights of common stock, and the shares of common stock underlying the RSUs are not considered issued and outstanding until actually issued.

During the first quarter of fiscal 2012, we granted 32,894 performance-based RSUs with a fair value of \$31.18 per unit to certain officers and key employees of the Company. The RSUs vest over a three-year period from their date of issuance, and are earned over a three-year performance cycle.

During the first quarter of fiscal 2012, we also granted 21,931 time-based RSUs with a fair value of \$31.18 per unit to certain officers and key employees of the Company. The RSUs vest over a three-year period from their date of issuance, and are time-based.



RSU activity (number of shares) for us was as follows:

	Number of Restricted Stock Units	Weighted Average Grant Date Fair Value Per Share
Outstanding at September 30, 2010	87,454	\$ 38.96
Granted	39,048	30.94
Expired or cancelled	(35,746)	36.50
Vested/exercised	(21,378)	37.68
Outstanding at September 30, 2011	69,378	36.10
Granted	54,825	31.18
Expired or cancelled	(19,729)	40.48
Vested/exercised		—
Outstanding at March 31, 2012	104,474	\$ 36.24

During the six months ended March 31, 2012, we recorded compensation expense of approximately \$0.2 million related to the RSUs. We recorded compensation expense of approximately \$0.3 million related to the RSUs for the six months ended March 31, 2011.

Restricted Stock

Under the 2006 Equity Compensation Plan (the 2006 Plan), any employee of the Company and its subsidiaries and consultants are eligible to participate in the plan and receive awards. Awards can take the form of options, stock appreciation rights, stock awards and performance unit awards.

During the first quarter of fiscal 2011, 26,000 shares of restricted stock were issued to certain officers and key employees of the Company with a fair value ranging from \$30.79 to \$32.12 per share under the 2006 Plan. The restricted stock grant vests over a three-year period on each anniversary of the grant date. Compensation expense is recognized over a three-year period based on the price per share on the grant date. In conjunction with the separation of our former President and Chief Executive Officer (CEO) in September 2011, the remaining 7,601 shares issued to him that were unvested became immediately vested. During the first half of 2012, there was no restricted stock granted.

During the six months ended March 31, 2012, we recorded compensation expense of \$0.3 million related to restricted stock grants. We recorded compensation expense of approximately \$0.6 million related to restricted stock grants for the six months ended March 31, 2011.

Stock Options

Stock option activity for the six months ended March 31, 2012 was as follows:

	Stock Options	Weighted Average Exercise Price	Remaining Weighted Average Contractual <u>Term (Years)</u>	In	gregate trinsic Value iousands)
Outstanding at September 30, 2011	97,550	\$ 18.44			
Granted	—	—			
Exercised	(18,800)	18.44			
Forfeited / Cancelled					
Outstanding at March 31, 2012	78,750	\$ 18.44	0.23	\$	1,452
Exercisable at March 31, 2012	78,750	\$ 18.44	0.23	\$	1,452



K. BUSINESS SEGMENTS

We manage our business through operating segments, which are comprised of two reportable business segments: Electrical Power Products and Process Control Systems. Electrical Power Products includes equipment and systems for the distribution and control of electrical energy. Process Control Systems consists principally of instrumentation, computer controls, communications and data management systems to control and manage critical processes.

The table below reflects certain information relating to our operations by reportable segment. All revenues represent sales from unaffiliated customers. The accounting policies of the business segments are the same as those described in the summary of significant accounting policies. Corporate expenses are allocated to the operating business segments primarily based on revenues. The corporate assets are mainly cash, cash equivalents and marketable securities.

Detailed information regarding our business segments is shown below (in thousands):

		Three Months Ended Six Months Ended March 31, March 31,		
	2012	2011	2012	2011
Revenues:				
Electrical Power Products	\$173,974	\$118,576	\$324,638	\$235,719
Process Control Systems	7,512	6,535	14,304	14,066
Total	\$181,486	\$125,111	\$338,942	\$249,785
Gross profit:				
Electrical Power Products	\$ 32,533	\$ 23,157	\$ 51,526	\$ 47,449
Process Control Systems	1,704	1,720	3,089	3,293
Total	\$ 34,237	\$ 24,877	\$ 54,615	\$ 50,742
Depreciation and amortization:				
Electrical Power Products	\$ 3,229	\$ 3,879	\$ 6,414	\$ 7,374
Process Control Systems	25	42	13	84
Total	\$ 3,254	\$ 3,921	\$ 6,427	\$ 7,458
Income before income taxes:				
Electrical Power Products	\$ 11,665	\$ 3,428	\$ 11,644	\$ 7,513
Process Control Systems	311	(146)	197	(530)
Total	\$ 11,976	\$ 3,282	\$ 11,841	\$ 6,983

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

The following discussion and analysis of our financial condition and results of operations should be read in conjunction with the accompanying condensed consolidated financial statements and related notes included elsewhere in this Quarterly Report on Form 10-Q and with our Annual Report on Form 10-K for the year ended September 30, 2011, which was filed with the Securities and Exchange Commission on December 12, 2011 and is available on the SEC's website at www.sec.gov.

CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS

We are including the following discussion to inform our existing and potential shareholders generally of some of the risks and uncertainties that can affect our Company and to take advantage of the "safe harbor" protection for forward-looking statements that applicable federal securities law affords.

From time to time, our management or persons acting on our behalf make forward-looking statements to inform existing and potential shareholders about our Company. These statements may include projections and estimates concerning the timing and success of specific projects and our future backlog, revenues, income and capital spending. Forward-looking statements are generally accompanied by words such as "estimate," "project," "predict," "believe," "expect," " anticipate," "plan," "goal" or other words that convey the uncertainty of future events or outcomes. In addition, sometimes we will specifically describe a statement as being a forward-looking statement and refer to this cautionary statement.

In addition, various statements in this Quarterly Report on Form 10-Q, including those that express a belief, expectation or intention, as well as those that are not statements of historical fact, are forward-looking statements. These forward-looking statements speak only as of the date of this report; we disclaim any obligation to update these statements unless required by securities law, and we caution you not to rely on them unduly. We have based these forward-looking statements on our current expectations and assumptions about future events. While our management considers these expectations and assumptions to be reasonable, they are inherently subject to significant business, economic, competitive, regulatory and other risks, contingencies and uncertainties, most of which are difficult to predict and many of which are beyond our control. These risks, contingencies and uncertainties relate to, among other matters, the following:

- The ongoing economic uncertainty and financial market conditions have negatively impacted and may continue to impact our customer base, suppliers and backlog.
- Our backlog is subject to unexpected adjustments and cancellations and, therefore, may not be a reliable indicator of our future earnings.
- Our volume of fixed-price contracts and use of percentage-of-completion accounting could result in volatility in our results of operations.
- Fluctuations in the price and supply of raw materials used to manufacture our products may reduce our profits.
- We face risks relating to material weaknesses in our internal control over financial reporting.
- Our industry is highly competitive.
- Our operations could be adversely impacted by the continuing effects from the U.S. government regulations on offshore deepwater and Outer Continental Shelf (OCS) drilling projects.
- International and political events may adversely affect our operations.
- Our acquisition strategy involves a number of risks.
- Our operating results may vary significantly from quarter to quarter.
- We may be unsuccessful at generating profitable internal growth.
- The departure of key personnel could disrupt our business.
- Our business requires skilled labor, and we may be unable to attract and retain qualified employees.

- Actual and potential claims, lawsuits and proceedings could ultimately reduce our profitability and liquidity and weaken our financial condition.
- Unforeseen difficulties with the maintenance or operation of our enterprise resource planning system could adversely affect our internal controls and our business.
- We carry insurance against many potential liabilities, and our management of risk may leave us exposed to unidentified or unanticipated risks.
- We may incur additional healthcare costs arising from federal healthcare reform legislation.
- Technological innovations by competitors may make existing products and production methods obsolete.
- Catastrophic events could disrupt our business.

We believe the items we have outlined above are important factors that could cause estimates included in our financial statements to differ materially from actual results and those expressed in a forward-looking statement made in this report or elsewhere by us or on our behalf. We have discussed many of these factors in more detail in our Annual Report on Form 10-K for the year ended September 30, 2011. These factors are not necessarily all of the factors that could affect us. Unpredictable or unanticipated factors we have not discussed in this report could also have material adverse effects on actual results of matters that are the subject of our forward-looking statements. We do not intend to update our description of important factors each time a potential important factor arises, except as required by applicable securities laws and regulations. We advise our shareholders that they should (1) be aware that factors not referred to above could affect the accuracy of our forward-looking statements and (2) use caution when considering our forward-looking statements.

Overview

We develop, design, manufacture and service custom engineered-to-order equipment and systems for the management and control of electrical energy and other critical processes. Headquartered in Houston, Texas, we serve the transportation, environmental, energy, industrial and utility industries. Our business operations are consolidated into two business segments: Electrical Power Products and Process Control Systems. Revenues and costs are primarily related to engineered-to-order equipment and systems which precludes us from providing detailed price and volume information.

The global markets in which Powell participates are capital-intensive and cyclical in nature. Cyclicality is driven by customer demand, global economic markets and potential environmental or regulatory impacts which affect the manner in which our customers proceed with large capital projects. Our customers analyze various factors including the short-term demand for oil and electrical energy, the overall banking environment, federal and state budgets, the outlook for offshore drilling and related regulatory actions and the drive towards environmental controls over the type and way energy is produced and utilized. These factors have contributed to decisions by customers to delay or to change where they place new capital projects, which decreased our backlog of orders to \$282.3 million entering fiscal 2011, down \$83.5 million from the beginning of fiscal 2010. However, during fiscal 2011, orders received were \$725.2 million compared to \$466.8 million during fiscal 2010 and our backlog increased to \$443.0 million at the beginning of fiscal 2012 (our current fiscal year). This resulted in an increase in backlog of 57% compared to the backlog at the beginning of fiscal 2011. Some of our orders received in fiscal 2011 are for large complex petrochemical and offshore oil and gas construction projects which will take several months to produce, most of which were awarded in competitive bid situations. This competitive-bid environment will continue to place downward pressure on gross profit margins as we work to fulfill these orders in fiscal 2012 and 2013. Orders received during the first six months of fiscal 2012 were \$392.1 million, a decrease of approximately \$10 million from the same period of the prior year.

Execution challenges on certain large projects at Powell Canada have negatively impacted net income in the first half of fiscal 2012. These execution challenges resulted from scope changes and cost overruns on certain projects. The Company is currently pursuing recovery of certain of these costs. However, there is no assurance these costs can be recovered and costs recovered are recorded when change orders are approved by the customers.

Results of Operations

Revenue and Gross Profit

Consolidated revenues increased \$56.4 million to \$181.5 million in the second quarter of fiscal 2012 compared to \$125.1 million in the second quarter of fiscal 2012, domestic revenues increased by 27.3% to \$105.3 million compared to the second quarter of 2011. Total international revenues increased to \$76.2 million in the second quarter of 2012 compared to \$42.4 million in the second quarter of 2011. This increase in domestic and international revenues in fiscal 2012 is directly related to the increase in the backlog of orders coming into fiscal 2012 compared to fiscal 2011, as discussed above. This increase in revenue also increased gross profit for the second quarter of fiscal 2012, as compared to the second quarter of fiscal 2011, by approximately \$9.4 million. Gross profit as a percentage of revenues decreased to 18.9% in the second quarter of 2012, compared to 19.9% in the second quarter of fiscal 2011 primarily as a result of project execution challenges on certain large projects at Powell Canada and margins were negatively impacted due to certain large projects awarded for fiscal 2012 being at lower margins than projects that were in process during the second quarter of fiscal 2011. We anticipate that gross margin levels, compared to fiscal 2011, will continue to be depressed over the balance of fiscal 2012 given the overall mix of jobs currently in our backlog.

For the six months ended March 31, 2012, consolidated revenues increased \$89.2 million to \$338.9 million compared to \$249.8 million for the six months ended March 31, 2011. For the first six months of fiscal 2012, domestic revenues increased by 22.3% to \$202.4 million compared to the first six months of fiscal 2011. Total international revenues increased to \$136.5 million in the first six months of 2012 compared to \$84.3 million in the first six months of fiscal 2011. The increase in domestic and international revenues in fiscal 2012 was directly related to the increase in the backlog of orders coming into fiscal 2012 compared to fiscal 2011, as discussed above. This increase in revenues also increased gross profit for the first six months of fiscal 2012, as compared to the first six months of fiscal 2011 by \$3.9 million. Gross profit as a percentage of revenues decreased to 16.1% for the first six months of fiscal 2012, compared to 20.3% for the first six months of fiscal 2011 primarily as a result of project execution challenges on certain large projects at Powell Canada and margins were negatively impacted due to certain large projects awarded for fiscal 2012 being at lower margins than projects that were in process during the first half of fiscal 2011.

Electrical Power Products

Our Electrical Power Products business segment recorded revenues of \$174.0 million in the second quarter of fiscal 2012, compared to \$118.6 million for the second quarter of fiscal 2011. The increase in revenues is a result of the increase in the backlog of orders coming into fiscal 2012. In the second quarter of 2012, revenues from public and private utilities were approximately \$24.7 million, compared to \$31.3 million in the second quarter of fiscal 2011. Revenues from industrial and commercial customers totaled \$139.6 million in the second quarter of 2012, an increase of \$59.2 million compared to the second quarter of fiscal 2011. Municipal and transit projects generated revenues of \$9.7 million in the second quarter of fiscal 2012 compared to \$6.9 million in the second quarter of fiscal 2011.

Business segment gross profit, as a percentage of revenues, was 18.7% in the second quarter of fiscal 2012, compared to 19.5% in the second quarter of fiscal 2011. This decrease in gross profit as a percentage of revenues resulted primarily from the competitive margins realized on certain projects during the second quarter of fiscal 2012.

For the six months ended March 31, 2012, our Electrical Power Products segment recorded revenues of \$324.6 million, compared to \$235.7 million for the six months ended March 31, 2011. The increase in revenues is a result of the increase in the backlog of orders coming into fiscal 2012. In the first six months of fiscal 2012, revenues from public and private utilities were approximately \$57.4 million, compared to \$64.5 million in the first six months of fiscal 2011. Revenues from commercial and industrial customers totaled \$246.0 million in the first six months of fiscal 2012, an increase of \$88.2 million compared to the first six months of fiscal 2011. Municipal and transit projects generated revenues of \$21.2 million in the first six months of fiscal 2012, compared to \$13.4 million in the first six months of fiscal 2011.

For the six months ended March 31, 2012, gross profit from the Electrical Power Products business segment, as a percentage of revenues, was 15.9%, compared to 20.1% for the six months ended March 31, 2011. This decrease in gross profit as a percentage of revenues resulted primarily from the project execution challenges on certain large projects at Powell Canada and the competitive margins realized on certain projects during the first half of fiscal 2012.

Process Control Systems

Our Process Control Systems business segment recorded revenues of \$7.5 million in the second quarter of fiscal 2012, an increase from \$6.5 million in the second quarter of fiscal 2011. Business segment gross profit, as a percentage of revenues, decreased to 22.7% in the second quarter of fiscal 2012 compared to 26.3% in the second quarter of fiscal 2011 due to the overall mix of projects.

For the six months ended March 31, 2012, our Process Control Systems business segment recorded revenues of \$14.3 million, an increase from \$14.1 million for the six months ended March 31, 2011. Business segment gross profit decreased as a percentage of revenues to 21.6% for the first six months of fiscal 2012, compared to 23.4% for the first six months of fiscal 2011 due to the overall mix of projects.

For additional information related to our business segments, see Note K of Notes to Condensed Consolidated Financial Statements.

Consolidated Selling, General and Administrative Expenses

Selling, general and administrative expenses were \$21.5 million for the second quarter of fiscal 2012, compared to \$21.5 million for the second quarter of fiscal 2012. Consolidated selling, general and administrative expenses decreased to 11.9% of revenues in the second quarter of fiscal 2012 compared to 17.2% of revenues in the second quarter of fiscal 2011. Selling, general and administrative expenses decreased as a percentage of revenues as a result of the increase in revenues.

Selling, general and administrative expenses were \$41.3 million for the first six months of fiscal 2012, compared to \$42.5 million for the first six months of fiscal 2011. For the six months ended March 31, 2012, consolidated selling, general and administrative expenses decreased to 12.2% of revenues, compared to 17.0% of revenues for the six months ended March 31, 2011. Selling, general and administrative expenses decreased due to decreases in legal fees, contract services and long-term incentive compensation expenses.

Gain on Sale of Investment

Gain on sale of investment recorded in the second quarter of fiscal 2011 consisted of a \$1.2 million gain which resulted from cash received from the sale of our 50% equity investment in a joint venture in Kazakhstan, which was previously a part of the acquisition of Powell Canada in fiscal 2010.

Interest Expense and Income

Interest expense was approximately \$68,000 and \$144,000 for the three and six months ended March 31, 2012, respectively, a decrease of approximately \$26,000 and \$64,000 compared to the three and six months ended March 31, 2011, respectively. The decrease in interest expense was primarily due to lower amounts outstanding under our credit facilities during the first half of fiscal 2012.

Interest income was approximately \$34,000 and \$63,000 for the three and six months ended March 31, 2012, respectively, compared to approximately \$63,000 and \$107,000 for the three and six months ended March 31, 2011, respectively, because cash available for investments decreased as the cash required to offset bank fees increased.

Provision for Income Taxes

Our provision for income taxes reflects an effective tax rate on earnings before income taxes of 38.1% in the second quarter of fiscal 2012, compared to 47.2% in the second quarter of fiscal 2011. For the first six months of fiscal 2012, our effective tax rate was 52.1%, compared to 40.4% for the first six months of fiscal 2011. The effective tax rates for the second quarter of fiscal 2012 and 2011, and the first six months of fiscal 2012 and 2011 have been negatively impacted by our inability to record a tax benefit related to pre-tax losses in Canada.

Net Income

In the second quarter of fiscal 2012, we generated net income of \$7.4 million, or \$0.63 per diluted share, compared to \$1.7 million, or \$0.15 per diluted share, in the second quarter of fiscal 2011. For the six months ended March 31, 2012, we recorded net income of \$5.7 million, or \$0.48 per diluted share, compared to \$4.2 million, or \$0.35 per diluted share, for the six months ended March 31, 2011. Net income for the second quarter of fiscal 2011 was favorably impacted by approximately \$1.2 million related to the gain on the sale of our investment in our previous 50% owned joint venture in Kazakhstan.

Backlog

The order backlog at March 31, 2012, was \$497.0 million, compared to \$443.0 million at September 30, 2011, and \$436.6 million at March 31, 2011. New orders placed during the second quarter of fiscal 2012 totaled \$203.3 million compared to \$216.8 million in the second quarter of fiscal 2011. Backlog has increased primarily due to strong activity in oil and gas production and petrochemical projects, as well as various projects in the Canadian Oil Sands region.

Liquidity and Capital Resources

Cash and cash equivalents increased to approximately \$128.1 million at March 31, 2012, primarily as a result of cash flow provided by operations of approximately \$23.4 million during the first six months of fiscal 2012. As of March 31, 2012, current assets exceeded current liabilities by 2.2 times and our debt to total capitalization was 1.6%.

At March 31, 2012, we had cash and cash equivalents of approximately \$128.1 million, compared to approximately \$123.5 million at September 30, 2011. We have a \$75.0 million revolving credit facility in the U.S., which expires in December 2016. As of March 31, 2012, there were no amounts borrowed under this line of credit. We also have a \$10.0 million revolving credit facility in Canada. At March 31, 2012, there was no balance outstanding under the Canadian revolving credit facility. Total long-term debt and capital lease obligations, including current maturities, totaled \$4.6 million at March 31, 2012, compared \$5.4 million at September 30, 2011. Letters of credit outstanding were \$42.7 million at March 31, 2012, compared to \$13.2 million at September 30, 2011, which reduce our availability under our US Revolver. Amounts available under the U.S. revolving credit facility were \$32.3 million at March 31, 2012. Amounts available under the Canadian revolving credit facility were \$10.0 million at March 31, 2012. For further information regarding our debt, see Notes G and I of Notes to Condensed Consolidated Financial Statements.

Approximately \$6.2 million of our cash at March 31, 2012, was held internationally for international operations. It is our intention to indefinitely reinvest all current and future foreign earnings internationally in order to ensure sufficient working capital and support and expand these operations. In the event that the Company elects to repatriate some or all of the foreign earnings that were previously deemed to be indefinitely reinvested outside the U.S., under current tax laws we would incur additional tax expense upon such repatriation.

We believe that cash available and borrowing capacity under our existing credit facility should be sufficient to finance anticipated operational activities, capital improvements and expansions, as well as debt repayments for the foreseeable future. We will continue to monitor the factors that drive our markets and strive to maintain our leadership and competitive advantage in the markets we serve while aligning our cost structures with market conditions.

Operating Activities

Cash provided by operating activities was \$23.4 million during the six months of fiscal 2012 and \$8.5 million during the first six months of fiscal 2011. Cash flow from operations is primarily influenced by demand for our products and services and is impacted as our progress payment terms with our customers are matched with the payment terms with our suppliers. Cash flow from operations increased during the first six months of fiscal 2012 compared to the same period in the prior year, primarily due to higher levels of business activity.

Investing Activities

Investments in property, plant and equipment during the first six months of fiscal 2012 totaled approximately \$18.9 million, compared to \$2.7 million during the first six months of fiscal 2011. A significant portion of these investments has been to acquire land in the United States and Canada to support our continued expansion in the Canadian Oil Sands and offshore production markets. During the first six months of fiscal 2011, we received cash of approximately \$1.2 million from the sale of our 50% equity investment in a joint venture in Kazakhstan.

Financing Activities

Net cash used in financing activities was approximately \$0.4 million for the first six months of fiscal 2012 and approximately \$0.6 million for the first six months of fiscal 20101

New Accounting Standards

See Note A to our condensed consolidated financial statements included in this report for information on new accounting standards.

Critical Accounting Policies

The discussion and analysis of our financial condition and results of operations are based on our condensed consolidated financial statements, which have been prepared in accordance with U.S. GAAP. The preparation of these condensed consolidated financial

statements requires us to make estimates and assumptions that affect the reported amounts of assets and liabilities, disclosures of contingent assets and liabilities known to exist at the date of the condensed consolidated financial statements and the reported amounts of revenues and expenses during the reporting period. We evaluate our estimates on an ongoing basis, based on historical experience and on various other assumptions that are believed to be reasonable under the circumstances. There can be no assurance that actual results will not differ from those estimates.

There have been no material changes to our critical accounting policies as disclosed in our Annual Report on Form 10-K for the year ended September 30, 2011.

Outlook

The global markets in which Powell participates are capital-intensive and cyclical in nature. Cyclicality is driven by customer demand, global economic markets and potential environmental or regulatory impacts which affect the manner in which our customers proceed with large capital projects. Our customers analyze various factors including the short-term demand for oil and electrical energy, the overall banking environment, federal and state budgets, the outlook for offshore drilling and related regulatory actions and the drive towards environmental controls over the type and way energy is produced and utilized. These factors have contributed to decisions by customers to delay or to change where they place new capital projects, which decreased our backlog of orders to \$282.3 million entering fiscal 2011, down \$83.5 million from the beginning of fiscal 2010. However, during fiscal 2011, orders received were \$725.2 million compared to \$466.8 million during the same twelve-month period of fiscal 2010 and our backlog increased to \$443.0 million at the beginning of fiscal 2011. This resulted in an increase in backlog of 57% compared to the backlog at the beginning of fiscal 2011. Some of the orders received in fiscal 2011 are for large complex petrochemical and offshore oil and gas construction projects which will take several months to produce, most of which were awarded in competitive bid situations. This competitive-bid environment could continue to place pressure on gross profit margins as we work to fulfill these orders in fiscal 2012 and 2013. Execution challenges on certain large projects at Powell Canada have negatively impacted and could continue to negatively impact our net income for fiscal 2012. These execution challenges resulted from scope changes and cost overruns on certain projects. The Company is currently pursuing recovery of certain of these costs. However, there is no assurance these costs can be recovered and costs recovered are recorded when change orders are approved by the customers. Orders

Growth in demand for energy is expected to continue over the long term. New infrastructure investments will be needed to ensure the available supply of petroleum products. New power generation and distribution infrastructure will also be needed to meet the growing demand for electrical energy. New power generation plants will also be needed to replace the aging facilities across the U.S., as those plants reach the end of their life cycle. A heightened concern for environmental damage, together with the uncertainty of gasoline prices, has expanded the popularity of urban transit systems, which should drive demand for investment in transit infrastructure, contingent upon available financing. Opportunities for future projects continue; however, the timing and pricing of many of these projects is difficult to predict. The demand for our products and services should continue as investment in large capital-intensive infrastructure projects receive funding and support. The increase in our backlog to \$497.0 million at March 31, 2012 resulted from several large complex projects awarded related to petrochemical and offshore oil and gas construction projects during fiscal 2011 and \$392.1 million of additional orders received in the first six months of fiscal 2012.

We believe that cash available and borrowing capacity under our existing credit facility should be sufficient to finance anticipated operational activities, capital improvements, market expansions and debt repayments for the foreseeable future. We will continue to monitor the factors that drive our markets and strive to maintain our leadership and competitive advantage and invest in the markets we serve while aligning our cost structures with market conditions.

We have recently acquired land in the United States and Canada to support our continued expansion in the Canadian Oil Sands and offshore production markets.

Item 3. Quantitative and Qualitative Disclosures About Market Risk

We are exposed to certain market risks arising from transactions we have entered into in the normal course of business. These risks primarily relate to fluctuations in interest rates, foreign exchange rates and commodity prices.

Interest Rate Risk

If we determine to borrow under one of our credit facilities, we will be subject to market risk resulting from changes in interest rates related to our floating rate bank credit facility. If we were to make such borrowings, a hypothetical 100 basis point increase in variable interest rates would not result in a material impact to our financial statements. While we do not currently have any derivative contracts to hedge our exposure to interest rate risk, we have in the past and may in the future enter into such contracts. During each of the past three years, we have not experienced a significant effect on our business due to changes in interest rates.

Foreign Currency Transaction Risk

We have operations that expose us to currency risk in the British Pound Sterling, the Canadian Dollar and to a lesser extent the Euro. Amounts invested in our foreign operations are translated into U.S. Dollars at the exchange rates in effect at the balance sheet date. The resulting translation adjustments are recorded as accumulated other comprehensive income (loss), a component of stockholders' equity in our consolidated balance sheets. We believe the exposure to the effects that fluctuating foreign currencies have on our consolidated results of operations is limited because the foreign operations primarily invoice customers and collect obligations in their respective currencies or U.S. Dollars. Our international operations are financed utilizing local credit facilities denominated in local currencies. Additionally, expenses associated with these transactions are generally contracted and paid for in the same local currencies. A 10% unfavorable change in the U.S. Dollar exchange rate, relative to other functional currencies in which we operate, would not materially impact our consolidated balance sheet at March 31, 2012.

During fiscal 2011 and the first half of fiscal 2012, we entered into eight foreign currency forward contracts to manage the volatility of future cash flows on certain long-term contracts that are denominated in the British Pound Sterling. The contracts are designated as cash flow hedges for accounting purposes. The changes in fair value related to the effective portion of the hedges are recognized as a component of accumulated other comprehensive income on our Condensed Consolidated Balance Sheets. At March 31, 2012, we recorded a net liability of \$1,000 on our Condensed Consolidated Balance Sheets related to these transactions.

Commodity Price Risk

We are subject to market risk from fluctuating market prices of certain raw materials. While such materials are typically available from numerous suppliers, commodity raw materials are subject to price fluctuations. We attempt to pass along such commodity price increases to our customers on a contract-by-contract basis to avoid a negative effect on profit margin. While we may do so in the future, we have not currently entered into any derivative contracts to hedge our exposure to commodity risk. We continue to experience price volatility with some of our key raw materials and components. Fixed price contracts may limit our ability to pass cost increases to our customers, thus negatively impacting our earnings. Fluctuations in commodity prices may have a material impact on our future earnings and cash flows.

Market Risk

We are also exposed to general market and other risk and its potential impact on accounts receivable or costs and estimated earnings in excess of billings on uncompleted contracts. The amounts recorded may be at risk if our customers' ability to pay these obligations is negatively impacted by economic conditions. Our customers and their industries are typically engineering procurement and construction firms, oil and gas producers, oil and gas pipelines, refineries, petrochemical plants, electrical power generators, public and private utilities, co-generation facilities, mining/metals operations, pulp and paper plants, transportation authorities, governmental agencies and other large industrial customers. We maintain ongoing discussions with customers regarding contract status with respect to payment status, change orders and billing terms in an effort to monitor collections of amounts billed.

Item 4. Controls and Procedures

Evaluation of Disclosure Controls and Procedures

A material weakness is a deficiency, or combination of deficiencies, in internal control over financial reporting, such that there is a reasonable possibility that a material misstatement of our annual or interim financial statements will not be prevented or detected on a timely basis. The matters that management identified in our Annual Report on Form 10-K (Annual Report) filed with the Securities and Exchange Commission (SEC) on December 12, 2011, continued to exist and were still considered material weaknesses in our internal control over financial reporting at March 31, 2012.

We have established and maintain a system of disclosure controls and procedures that are designed to provide reasonable assurance that information required to be disclosed in our reports filed with the SEC pursuant to the Securities Exchange Act of 1934, as amended, is recorded, processed, summarized and reported within the time periods specified in the rules and forms of the Commission and that such information is accumulated and communicated to our management, including our Chief Executive Officer (CEO) and Chief Financial Officer (CFO), as appropriate, to allow timely decisions regarding required disclosures.

Management, with the participation of our CEO and CFO, evaluated the effectiveness of the design and operation of our disclosure controls and procedures (as defined in Rules 13a-15(e) and 15d-15(e) of the Securities Exchange Act of 1934, as amended) as of the

end of the period covered by this report. In designing and evaluating disclosure controls and procedures, our management recognizes that any disclosure controls and procedures, no matter how well designed and operated, can only provide reasonable assurance of achieving the desired control objective. In light of the material weaknesses found in our internal controls over financial reporting previously disclosed in our Annual Report filed with the SEC on December 12, 2011 that continue to exist, our CEO and CFO have concluded that our disclosure controls and procedures were not effective.

We have performed additional analysis and other post-closing procedures to ensure our condensed consolidated financial statements as of and for the period ended March 31, 2012, were prepared in accordance with generally accepted accounting principles. Accordingly, management concluded that the financial statements fairly present in all material respects our financial condition, results of operations and cash flows as of, and for, the periods presented in this quarterly report.

Remediation Plan

As of March 31, 2012, there were control deficiencies which constituted material weaknesses in our internal control over financial reporting. Management has taken, and is taking steps to strengthen our internal control over financial reporting. Specifically,

- A new Controller and other accounting staff members have been hired at Powell Canada where the control deficiencies exist.
- Members of project management and the accounting staff at Powell Canada have received additional training related to policies, procedures and internal controls, including Powell's policies regarding monthly reconciliations and supervisory review procedures for all significant accounts.
- Additional training has and will be provided related to the ERP business system that was implemented at Powell Canada in April 2011 to foster utilization of tools available for timely review of projects in progress.
- User access and segregation of duties will be reviewed to determine and implement the appropriate steps necessary to prevent or mitigate potential conflicts.
- A consultant, along with the management team, has evaluated, made recommendations and implemented corrective actions over the contract revenue and cost accumulation processes of Powell Canada's field operations.
- Our internal audit department has reviewed and will continue to assess progress on the remediation plan noted above.
- Various members of the Company's management team have been on-site in Canada to oversee the remediation efforts.

While we have taken certain actions to address the material weaknesses identified, additional measures may be necessary as we work to improve the overall effectiveness of our internal reporting over financial reporting. Through the actions described above, we believe that we are addressing the deficiencies that affected our internal control over financial reporting as of March 31, 2012. Until the above-described actions are fully implemented and operate for a sufficient period of time, we will not be able to conclude that the material weaknesses have been remediated. We will continue to monitor and assess our remediation activities to address the material weaknesses discussed above through remediation as soon as practicable.

Internal Control over Financial Reporting

There has been no change in our internal control over financial reporting that occurred during our second quarter of fiscal 2012 that has materially affected, or is reasonably likely to materially affect, our internal control over financial reporting.

PART II - OTHER INFORMATION

Item 1. Legal Proceedings

We are involved in various legal proceedings, claims and other disputes arising in the ordinary course of business which, in general, are subject to uncertainties and the outcomes are not predictable. We do not believe that the ultimate conclusion of these disputes could materially affect our financial position or results of operations.

Item 1A. Risk Factors

There are no material changes from the risk factors previously disclosed in our Annual Report on Form 10-K for the fiscal year ended September 30, 2011.

Item 5. Other Information

Compensatory Arrangements of Certain Officers

On May 7, 2012, the company entered into employment agreements with each of the following named executive officers: (i) Don R. Madison, Executive Vice President and Chief Financial and Administrative Officer, and (ii) Milburn E. Honeycutt, Vice President, Controller and Chief Accounting Officer.

Each of the agreements commenced on the date on which such agreement was entered into and expires upon the earlier of (a) the last day of the month in which such executive reaches age 65, in which case, unless otherwise agreed to, such executive's employment shall continue at will and shall be terminable by either the executive or the Company for any reason or (b) the date on which such executive's employment terminates pursuant to the terms of such agreement or the executive's resignation or retirement. Pursuant to the agreements, each executive is entitled to receive a base salary equal to the base salary in effect for such executive as of the date of such agreement and which has been previously disclosed, subject to review annually by and at the sole discretion of the Compensation Committee) of the Board of Directors of the Company. The agreements also provide that each executive will be eligible for an annual discretionary bonus, short term incentive compensation award and/or long-term incentive compensation award based upon achievement of performance objectives established by the Compensation Committee.

Each agreement further provides that in the event the Company terminates the relevant executive's employment without "cause" or if the executive terminates his employment for "good reason" (as these terms are defined in the agreements) during the term of the agreement, the executive will be entitled to receive, among other things, such executive's then-current base salary for the Applicable Period, an amount equal to such executive's then-current target short-term incentive compensation and such executive's equity-based awards will immediately vest. In the event that the Company terminates the executive's employment without cause or the executive terminates his employment for good reason within a specified period following a change in control, the executive will be entitled to receive, among other things, such executive's then-current base salary for the Applicable Period and an amount equal to two-times such executive's then-current target short-term incentive compensation. Additionally, such executive's equity-based awards will immediately vest. For purposes hereof, the term "Applicable Period" shall mean 36 months for Mr. Madison and 24 months for Mr. Honeycutt.

In connection with entry into the employment agreements, each executive has agreed not to (A) compete with the Company for so long as such executive is employed by the Company and for the greater of (i) one year from the date of termination of such executive's employment and, (ii) if applicable, the period during which such executive is entitled to receive severance or (B) solicit or encourage any employee or consultant of the Company to leave employment of the Company or otherwise hire any such employees.

The employment agreements contain a provision that provides that the rights and benefits provided by such agreements are intended to replace the rights and benefits provided by the Company's Executive Severance Protection Plan adopted and effective September 20, 2002. By execution of such employment agreement, each executive, to the extent applicable, waived and released any rights, benefits or payments to which such executive would have been entitled thereunder and acknowledged that such plan was terminated by the Company. There were no termination penalties incurred by the Company in connection with the termination of such plan.

The foregoing description of the employment agreements is qualified in its entirety by reference to Mr. Madison's employment agreement and Mr. Honeycutt's employment agreement, copies of which are filed as Exhibit 10.1 and 10.2, respectively, to this quarterly report and is incorporated by reference herein.

Submission of Matters to a Vote of Security Holders

At the annual meeting of stockholders of the Company held on February 29, 2012, James F. Clark, Scott E. Rozzell, Stephen W. Seale, Jr. and Robert C. Tranchon were re-elected as directors of the Company with terms ending in 2015. The other directors continuing in office after the meeting are Joseph L. Becherer, Eugene L. Butler, Christopher E. Cragg, Bonnie V. Hancock and Thomas W. Powell. As to each nominee for director, the number of votes cast for, against or withheld, as well as the number of abstentions and broker non-votes, were as follows:

		Votes Cast			
<u>Nominee</u>	Votes Cast For	Against	Votes Withheld	Abstentions	Non-Votes
James F. Clark	10,167,688		433,487		
Scott E. Rozzell	10,149,453	—	451,722	—	—
Stephen W. Seale, Jr.	7,392,323	—	3,208,852	—	—
Robert C. Tranchon	7,484,681		3,116,494		—

In addition at this annual meeting, the stockholders approved the Company's "say-on-pay" proposal relating to the compensation paid to the Company's executives and the related compensation discussion and analysis contained in the Company's proxy statement relating to such annual meeting as follows:

Votes Cast For	Votes Cast Against	Votes Abstained
8,094,993	2,483,999	22,183

Item 6. Exhibits

Number		Description of Exhibits
3.1	—	Certificate of Incorporation of Powell Industries, Inc. filed with the Secretary of State of the State of Delaware on February 11, 2004 (filed as Exhibit 3.1 to our Form 8-A/A filed November 1, 2004, and incorporated herein by reference).
3.2	_	By-laws of Powell Industries, Inc. (filed as Exhibit 3.2 to our Form 8-A/A filed November 1, 2004, and incorporated herein by reference).
*10.1	_	Employment Agreement dated May 8, 2012 by and between the Company and Don R. Madison.
*10.2	_	Employment Agreement dated May 8, 2012 by and between the Company and Milburn E. Honeycutt.
*10.3	_	Tenth Amendment to Credit agreement, dated as of March 26, 2012, among Powell Industries, Inc., as Parent, the subsidiaries of Powell Industries, Inc. identified therein, as Borrowers, Bank of America, N.A., as Administrative Agent, Swing Line Lender and L/C issuer, and the Lenders party.
*10.4	—	Amended and Restated Credit Agreement, made as of April 26, 2012, between Powell Canada Inc., as Borrower, Powell Industries, Inc., Nextron Limited, and PPC Technical Services Inc., as Guarantors; and HSBC Bank Canada, as Lender.
*31.1	_	Certification of Chief Executive Officer pursuant to Rule 13a-14(a)/15d-14(a).
*31.2	_	Certification of Chief Financial Officer pursuant to Rule 13a-14(a)/15d-14(a).
*32.1	—	Certification of Chief Executive Officer Pursuant to Section 18 U.S.C. Section 1350, as Adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.
*32.2	—	Certification of Chief Financial Officer Pursuant to Section 18 U.S.C. Section 1350, as Adopted Pursuant to Section 906 of the Sarbanes- Oxley Act of 2002.
101.INS	—	XBRL Instance Document
101.SCH	_	XBRL Taxonomy Extension Schema
101.CAL		XBRL Taxonomy Extension Calculation Linkbase
101.LAB		XBRL Taxonomy Extension Label Linkbase
101.PRE		XBRL Taxonomy Extension Presentation Linkbase

Filed herewith

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

<u>May 9, 2012</u> Date

<u>May 9, 2012</u> Date POWELL INDUSTRIES, INC. (Registrant)

By: /s/ Thomas W. Powell

Thomas W. Powell Chairman of the Board President and Chief Executive Officer (Principal Executive Officer)

By: /s/ Don R. Madison

Don R. Madison Executive Vice President Chief Financial and Administrative Officer (Principal Financial Officer)

EXHIBIT INDEX Number Exhibit Title 3.1 Certificate of Incorporation of Powell Industries, Inc. filed with the Secretary of State of the State of Delaware on February 11, 2004 (filed as Exhibit 3.1 to our Form 8-A/A filed November 1, 2004, and incorporated herein by reference). 3.2 By-laws of Powell Industries, Inc. (filed as Exhibit 3.2 to our Form 8-A/A filed November 1, 2004, and incorporated herein by reference). Employment Agreement dated May 8, 2012 by and between the Company and Don R. Madison. *10.1 *10.2 Employment Agreement dated May 8, 2012 by and between the Company and Milburn E. Honeycutt. *10.3 Tenth Amendment to Credit agreement, dated as of March 26, 2012, among Powell Industries, Inc., as Parent, the subsidiaries of Powell Industries, Inc. identified therein, as Borrowers, Bank of America, N.A., as Administrative Agent, Swing Line Lender and L/C issuer, and the Lenders party. *10.4 Amended and Restated Credit Agreement, made as of April 26, 2012, between Powell Canada Inc., as Borrower, Powell Industries, Inc., Nextron Limited, and PPC Technical Services Inc., as Guarantors; and HSBC Bank Canada, as Lender. *31.1 Certification of Chief Executive Officer pursuant to Rule 13a-14(a)/15d-14(a). *31.2 Certification of Chief Financial Officer pursuant to Rule 13a-14(a)/15d-14(a). *32.1 Certification of Chief Executive Officer Pursuant to Section 18 U.S.C. Section 1350, as Adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002. *32.2 Certification of Chief Financial Officer Pursuant to Section 18 U.S.C. Section 1350, as Adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002. 101.INS XBRL Instance Document 101.SCH XBRL Taxonomy Extension Schema 101.CAL XBRL Taxonomy Extension Calculation Linkbase 101.LAB XBRL Taxonomy Extension Label Linkbase 101.PRE XBRL Taxonomy Extension Presentation Linkbase

Filed herewith



EXECUTIVE EMPLOYMENT AGREEMENT

THIS EMPLOYMENT AGREEMENT and an ancillary agreement to be effective simultaneously herewith entitled "Confidentiality, Non-Competition and Non-Solicitation Agreement" (the "Confidentiality Agreement"), a copy of which is attached hereto as Attachment A, and incorporated herein by reference for all purposes, (this agreement and the Confidentiality Agreement being hereinafter collectively referred to as "this Agreement") is entered into effective as of May 8, 2012 (the "Effective Date"), by and between Powell Industries, Inc. and its affiliates (the "Company") and Don R. Madison ("Executive").

WHEREAS, the Company desires to continue the employment of Executive as Executive Vice President & Chief Financial Officer of the Company from and after the Effective Date until such date as his employment shall end pursuant to the terms and conditions contained herein; and

WHEREAS, Executive desires to continue employment with the Company in such position pursuant to the terms and conditions contained herein;

NOW, THEREFORE, IT IS HEREBY AGREED AS FOLLOWS:

I. EMPLOYMENT TERM.

The term of this Agreement shall commence on the Effective Date and expire at the earlier of:

a. The last day of the month in which the Executive reaches age 65, in which case, unless the parties agree otherwise, Executive's employment shall continue at will and shall be terminable by either party, for any reason, in which event the executive shall be entitled to the severance benefits provided for under Section V.B; or

b. The date Executive's employment terminates subject to the provisions of this Agreement regarding termination, resignation or retirement.

Executive and the Company acknowledge that the employment relationship provided herein may be terminated at any time, upon written notice to the other party for any reason, at the option of either the Company or Executive. However, as provided in this Agreement, Executive may be entitled to certain severance benefits depending upon the circumstances of Executive's termination of employment. The period Executive is employed by the Company under this Agreement is referred to herein as the "Employment Term."

II. CERTAIN DEFINITIONS

A. "Accrued Rights" shall mean:

1. Executive's earned, but unpaid compensation, to include base salary, vehicle allowance, short term incentive and long term incentive compensation through the date of termination;

2. Reimbursement, within sixty (60) days following submission by Executive to the Company of appropriate supporting documentation, for any unreimbursed reasonable business expenses properly incurred by Executive in the performance of Executive's duties in accordance with the Company's expense reimbursement policy prior to the date of Executive's termination, provided claims for such reimbursement (accompanied by appropriate supporting documentation) are submitted to the Company within ninety (90) days following the date such expenses were incurred and within thirty (30) days following Executive's termination; and

3. Such Employee Benefits, if any, as to which Executive may be entitled under the terms of the employee benefit plans of the Company.

B. "<u>Cause</u>" shall mean:

1. Executive's conviction of (or plea of nolo contendere to) a felony;

2. Executive's dishonesty, theft, embezzlement or fraud with respect to the business, property, reputation or affairs of the Company;

3. Executive's willful violation of the Company's Business Code of Conduct and Business Ethics and/or any other of the Company's employment, personnel, safety or other policies as now exist or as may hereafter be amended;

4. Executive's having committed any material violation of any federal or state law regulating securities (without having relied on the advice of the Company's attorney or outside auditor) or having been the subject of any final order, judicial or administrative, obtained or issued by the Securities and Exchange Commission, or any regulatory authority having jurisdiction over the Company's securities for any securities violation involving fraud, including, without limitation, any such order consented to by Executive in which findings of facts or any legal conclusions establishing liability are neither admitted nor denied;

5. Executive's willful and continued failure to devote substantially all of his business time to the Company's business affairs (excluding failures due to illness, incapacity, vacations, incidental civic activities and incidental personal time); or

6. Executive's unauthorized disclosure of confidential information of the Company that is materially injurious to the Company.

Notwithstanding the above, however, and except with regard to the events described in subparagraph (1) above, Cause shall not exist with respect to any matter unless the Company gives the Executive written notice of such matter within ninety (90) days of the date the Company knew of its occurrence. Such notice shall specify with reasonable particularity the acts, events or conditions which are claimed to constitute Cause. If the Company fails to give such notice timely, the Company shall be deemed to have waived its right to terminate Executive for Cause with respect to such matter.

Upon receipt of the notice described above, Executive shall have thirty (30) days to (i) cure or correct the acts, event or conditions specified in the notice, (ii) commence Executive's best efforts to cure or correct the event constituting such and continue such efforts until the act, event or condition is cured; or (iii) if applicable, provide the Company with written evidence or documentation that the acts or events claimed to constitute Cause did not occur, or were not performed or omitted by Executive, or otherwise do not constitute Cause as described in this Agreement.

For purposes of this definition, no act, or failure to act, on Executive's part shall be deemed "willful" unless done, or omitted to be done, by Executive not in good faith and without reasonable belief that Executive's action or omission was in the best interest of the Company.

C. "Change of Control" shall mean any of the following:

1. any "person" (as such term is used in Section 13(d) and 14(d) of the Securities Exchange Act of 1934, as amended (the "Exchange Act")), (other than a trustee or other fiduciary holding securities under an employee benefit plan of the Company or any affiliate, or any corporation owned, directly or indirectly, by the stockholders of the Company in substantially the same proportions as their ownership of stock of the Company), acquires "beneficial ownership" (within the meaning of Rule 13d-3 under the

Exchange Act) of securities of the Company representing 35% or more of the combined voting power of the Company's then outstanding securities; provided, however, that if the Company engages in a merger or consolidation in which the Company or surviving entity in such merger or consolidation becomes a subsidiary of another entity, then references to the Company's then outstanding securities shall be deemed to refer to the outstanding securities of such parent entity;

2. a change in the composition of the Board, as a result of which fewer than a majority of the directors are Incumbent Directors. "Incumbent Directors" shall mean directors who either (i) are directors of the Company as of the Effective Date, or (ii) are elected, or nominated for election, to the Board with the affirmative votes of at least two-thirds of the Incumbent Directors at the time of such election or nomination, but Incumbent Director shall not include an individual whose election or nomination occurs as a result of either (1) an actual or threatened election contest (as such terms are used in Rule 14a-11 of Regulation 14A promulgated under the Exchange Act) or (2) an actual or threatened solicitation of proxies or consents by or on behalf of a person other than the Board of Directors of the Company;

3. the consummation of a merger or consolidation of the Company with any other corporation, other than a merger or consolidation which would result in the voting securities of the Company outstanding immediately prior thereto continuing to represent (either by remaining outstanding or by being converted into voting securities of the surviving entity (or if the surviving entity is or shall become a subsidiary of another entity, then such parent entity) more than 50% of the combined voting power of the voting securities of the Company (or such surviving entity or parent entity, as the case may be) outstanding immediately after such merger or consolidation;

4. the stockholders of the Company approve a plan of complete liquidation of the Company; or

5. the sale or disposition (other than a pledge or similar encumbrance) by the Company of all or substantially all of the assets of the Company other than to a subsidiary or subsidiaries of the Company.

D. "<u>Date of Termination</u>" shall mean the date the Notice of Termination is given unless such Notice of Termination is by Executive in which event the Date of Termination shall not be less than 30 days following the date the Notice of Termination is given. Further, a Notice of Termination given by Executive due to a Good Reason event that is corrected by the Company before the Date of Termination shall be void.

E. "<u>Disability</u>" shall mean that Executive: (i) is unable to perform the essential functions of Executive's job title and duties by reason of any medically determinable physical or mental impairment which can be expected to result in death or can be expected to last for a continuous period of not less than twelve (12) months, provided that Executive or his representative has provided the Company with certification of such disability from a licensed physician or other medical services provider acceptable to the Company in its sole discretion; (ii) is, by reason of any medically determinable physical or mental impairment which can be expected to result in death or can be expected to last for a continuous period of not less than twelve (12) months, receiving income replacement benefits for a period of not less than three months under an accident and health plan or disability insurance policy covering employees of the Company; or (iii) is determined by the Social Security Administration to be disabled.

F. "Good Reason" shall mean:

1. a material reduction in Executive's authority, duties or responsibilities or the assignment to Executive **of** duties or responsibilities inconsistent in any material respect from those of Executive in effect immediately prior to the change;

2. a material reduction of Executive's compensation and benefits, including, without limitation, annual base salary, targeted short-term incentive compensation, targeted long-term incentive compensation, and equity incentive opportunities, from those in effect immediately prior to the change;

3. the Company fails to obtain a written agreement from any successor or assigns of the Company to assume and perform this Agreement as provided in Section VI.I hereof;

4. the Company requires Executive, without Executive's consent, to be based at any office located more than 50 miles from the Company's offices to which Executive was based immediately prior to the Change of Control, except for travel reasonably required in the performance of Executive's duties; or

5. the Company's breach of a material term of this Agreement.

Notwithstanding the above, however, Good Reason shall not exist with respect to any matter unless the Executive gives the Company written notice of such matter within ninety (90) days of the date the Executive knew or reasonably should have known of its occurrence. Such notice shall specify with reasonable particularity, the acts, events or conditions which are claimed to constitute Good Reason. If the Executive fails to give such notice timely, the Executive shall be deemed to have waived Executive's right to resign for Good Reason with respect to such matter.

Upon receipt of the notice described above, the Company shall have sixty (60) days to (i) cure or correct the acts, event or conditions specified in the notice, (ii) commence the Company's best efforts to cure or corrects the event constituting such and continue such efforts until the act, event or condition is cured; or (iii) if applicable, provide the Executive with written evidence or documentation that the acts or events claimed to constitute Good Reason did not occur, or otherwise do not constitute Good Reason as described in this Agreement.

For purposes of this Agreement, "Good Reason" shall be construed to refer to Executive's positions, duties, and responsibilities in the position or positions in which Executive was serving before any event as described in subparagraphs (1) through (5) above, which shall not include titles or positions with subsidiaries and affiliates of the Company that are held primarily for administrative convenience.

"Good Reason" shall also include any of the foregoing acts or omissions by a successor in interest to the Company as referenced in Sections II.C(3), (4) or (5) above.

G. "<u>Notice of Termination</u>" shall mean a written notice delivered to the other party indicating the specific termination provision in this Agreement relied upon for termination of Executive's employment which shall set forth in reasonable detail the facts and circumstances claimed to provide a basis for termination of Executive's employment under the provision so indicated. For the purpose, termination of Executive's employment shall be interpreted consistent with the meaning of the term "Separation from Service" in Section 409A(a)(2)(A)(i) of the Internal Revenue Code of 1986, as amended (the "Code") and applicable regulation authority.

H. "<u>Poor Performance</u>" shall mean Executive's willful and continued failure to perform substantially the duties of Executive's position after a written demand for substantial performance is delivered to him which specifically identifies the nature of such unacceptable performance, and which is not cured by Executive within a reasonable period, not to exceed sixty (60) days. For purposes of the definition in of "Poor Performance" as used herein, no act, or failure to act, on Executive's part shall be deemed "willful" unless done, or omitted to be done, by Executive not in good faith and without reasonable belief that Executive's action or omission was in the best interest of the Company.

I. "Protected Period" shall mean the 36-month period beginning on the effective date of a Change of Control.

J. "<u>Retirement</u>" shall mean Executive has reached 62 years of age ("normal retirement") or age 60 with at least five (5) years of active service ("early retirement"); provided, however that Executive cannot be required to retire and must consent in writing to any Retirement.

K. "<u>Severance Period</u>" shall mean the time period during which the Executive receives salary continuation benefits following a termination of employment by the Company for Poor Performance as described in Section V.C Without Cause or Resignation by Executive for Good Reason either prior to a Change in Control as described Section V.D or after a Change in Control as described in Section V.E.

L. "<u>Targeted STIC</u>" shall mean the targeted value of Executive's annual Short Term Incentive Compensation opportunity for the year in which the Date of Termination occurs, or the target value in place prior to a material reduction in compensation, or the fiscal year immediately preceding a Change of Control whichever is a greater amount.

M. "<u>Targeted LTIC</u>" shall mean the targeted value of Executive's annual Long Term Incentive Compensation opportunity for the year in which the Date of Termination occurs, or the target value in place prior to a material reduction in compensation, or the fiscal year immediately preceding a Change of Control, whichever is a greater amount.

N. "<u>Termination Base Salary</u>" shall be the greater of, the Executive's base salary at the rate in effect at the time the Notice of Termination, or the Executive's base salary in place prior to a material reduction in compensation, or the Executive's base salary in effect immediately prior to a Change of Control.

III. POSITION.

A. During the Employment Term, Executive shall serve as the Company's Executive Vice President & Chief Financial Officer. In such position, Executive shall report to President & Chief Executive Officer or as directed by the Board of Directors of the Company, and shall have the authority, responsibilities, and duties reasonably accorded to, expected of and consistent with Executive's position, as may be assigned to Executive. The Executive's principal place of employment shall be the principal offices of the Company currently located in Houston; <u>provided</u>, <u>however</u>, that Executive understands and agrees that Executive will be required to travel from time to time for business reasons.

B. During the Employment Term, Executive shall devote his full business time, attention and efforts to the performance of Executive's duties hereunder and will not engage in any other activity (for compensation or otherwise without written notice to, and the written consent of the Board of Directors of the Company (the "Board")) which, in the good faith opinion of the Board, could, either individually or in the aggregate, reasonably be expected to conflict or interfere with or otherwise adversely affect the rendition of such performance either directly or indirectly. The foregoing limitations shall not be construed as prohibiting Executive from making personal investments in such form or manner as will neither require Executive's services in the operation or affairs of the companies or businesses in which such investments are made nor violate the terms of this Agreement hereof or otherwise conflict or interfere with Executive's responsibilities to the Company; provided, however, that Executive agrees he will not join any boards (other than community and civic boards which do not interfere with his duties to the Company) without the prior written approval of the Board.

IV. COMPENSATION.

A. <u>Base Salary</u>. The Company shall pay Executive a base salary at the annual rate of \$308,760 payable in accordance with the Company's payroll practices for similarly situated executives (the "Base Salary"). Executive's Base Salary shall be subject to review annually by and at the sole discretion of the Compensation Committee of the Board (the "Compensation Committee").

B. <u>Short Term Incentive Compensation Award</u>. For each fiscal year ("Fiscal Year") of the Company during the Employment Term, Executive shall be given the opportunity to earn annual Short Term Incentive Compensation Award (the "STIC Award"). Executive's annual Short Term Incentive Compensation opportunity for each Fiscal Year ending during the Employment Term shall be set by the Compensation Committee, in its sole discretion. The actual STIC Award payable to Executive with respect to a Fiscal Year shall be dependent upon the achievement of performance objectives established by the Compensation Committee for such Fiscal Year and may be greater or less than the Short Term Incentive Compensation opportunity depending on performance objective results. The Compensation Committee shall also have the sole right to determine whether Executive may be entitled to a discretionary bonus at any time and to determine the criteria to be considered in making such decision. Except as otherwise provided in this Agreement, the payment of STIC Award shall be at the same time as Short Term Incentive Compensation Awards are paid to other similar executives of the Company.

C. Long Term Incentive Compensation Award. During the Employment Term, Executive shall be shall be given the opportunity to earn annual Long Term Incentive Compensation Award (the "Target LTIC Award") under the Company's Equity Incentive Plan (the "Equity Plan"), as modified, amended or replaced from time to time. Executive's annual Targeted Long Term Incentive Compensation Award for each Fiscal Year during the Employment Term shall be set by the Compensation Committee, in its sole discretion. The actual LTIC Award payable to Executive with respect to a Fiscal Year shall be dependent upon the achievement of performance objectives established by the Compensation Committee for such Fiscal Year and may be greater or less than the Target Long Term Incentive Compensation opportunity depending on performance objective results. Except as otherwise provided in this Agreement, the payment of LTIC Award shall be at the same time as Long Term Incentive Compensation Awards are paid to other similar executives of the Company.

D. <u>Employee Benefits</u>. During the Employment Term, Executive shall be eligible to participate in the Company's employee benefit plans as in effect from time to time (collectively, "Employee Benefits") on the same basis as such employee benefit plans are generally made available to other comparable executives of the Company.

E. <u>Vacation</u>. Executive shall be entitled to four (4) weeks of annual vacation leave for each Fiscal Year during which Executive is employed (prorated for Executive's initial year, if not a full year). Such leave shall be administered in accordance with the Company's vacation policy.

F. <u>Automobile Allowance</u>. During the Employment Term, Executive shall be entitled to an automobile allowance of \$2,000 per month paid in accordance with the Company's normal payroll practices.

G. <u>Business Expenses</u>. During the Employment Term, reasonable business expenses incurred by Executive in the performance of Executive's duties hereunder shall be reimbursed by the Company in accordance with the Company's expense reimbursement policy.

V. TERMINATION OF EMPLOYMENT.

Executive shall not have a "termination of employment" for purposes of this Agreement unless such termination constitutes a "separation from service" for purposes of Section 409A of the Internal Revenue Code of 1986, as amended, and the applicable Treasury Regulations thereunder (the "Code"). Notwithstanding any other provision of this Agreement, the provisions of this Section V shall exclusively govern Executive's rights upon termination of employment with the Company and its affiliates.

A. By the Company for Cause or Resignation by Executive Without Good Reason.

1. The Employment Term and Executive's employment hereunder may be terminated by the Company for Cause or by Executive's resignation without Good Reason.

2. If Executive's employment is terminated by the Company for Cause, or if Executive resigns without Good Reason, then, subject to the further terms of this Agreement, Executive shall be entitled to receive:

a. The Accrued Rights (refer to Section II.A)

B. Retirement, Disability or Death.

1. The Employment Term and Executive's employment hereunder shall terminate upon Executive's Retirement, Disability or Death; provided, however, that if Executive retires under circumstances that would constitute "Good Reason", Executive shall be deemed to have terminated for "Good Reason" and be entitled to the applicable rights and benefits provided in this agreement.

2. Upon termination of Executive's employment hereunder for either Retirement, Disability or Death, then Executive or Executive's estate (as the case may be) shall be entitled to receive the following:

a. The Accrued Rights (refer to Section II.A); and

b. A prorated portion of the Targeted STIC for the current Fiscal Year, prorated based on the percentage of the current Fiscal Year that shall have elapsed through the date of termination; and

c. With respect to any outstanding equity-based awards, whether "time-based" or "performance-based" vesting (including, but not limited to, any unvested options, restricted stock, restricted stock units, and performance share units) such outstanding awards shall immediately vest; and

d. In the event of termination for Disability or Death, an amount, paid on the first business day of each month, equal to 100% of the applicable monthly COBRA premium under the Company's group health plan, continued for the lesser of (i) twelve (12) months or (ii) until such COBRA coverage for Executive terminates.

C. By the Company for Poor Performance.

1. The Employment Term and Executive's employment hereunder may be terminated by the Company for Poor Performance.

2. If Executive's Employment is terminated by the Company for Poor Performance then Executive shall be entitled to receive from the Company the following:

a. The Accrued Rights (refer to Section II.A);

b. Continued payment of Executive's Termination Base Salary for twelve (12) months (the "Severance Period") following the date of such termination, payable in accordance with the Company's normal payroll practices as in effect on the date of termination;

c. With respect to any outstanding equity-based awards, whether "time-based" or "performance-based" vesting (including, but not limited to, any unvested options, restricted stock, restricted stock units, and performance share units) such outstanding awards shall be forfeited; and

d. An amount, paid on the first business day of each month equal to one hundred percent (100%) of the applicable COBRA premium under the Company's group health plan, continued for the lesser of (1) twelve (12) months from the date of termination of Executive's employment; or (2) the date on which Executive qualifies for health insurance as a result of employment by or association with a subsequent employer.

D. By the Company Without Cause and not for Poor Performance or Resignation by Executive for Good Reason Prior to a Change in Control.

1. The Employment Term and Executive's employment hereunder may be terminated by the Company without Cause or by Executive's resignation for Good Reason.

2. If Executive's employment is terminated by the Company without Cause (and other than by reason of Executive's death or Disability) or if Executive resigns for Good Reason, then Executive shall be entitled to receive from the Company the following:

a. The Accrued Rights (refer to Section II.A);

b. Continued payment of Executive's Termination Base Salary for twenty-four (24) months (the "Severance Period") following the date of such termination, payable in accordance with the Company's normal payroll practices as in effect on the date of termination;

c. An amount equal to one (1) times the Target Short Term Incentive Compensation of Executive for the Fiscal Year in which Executive's employment terminates, which amount shall be payable in one (1) installment due six (6) months after the date of Executive's termination of employment;

d. With respect to any outstanding equity-based awards, whether "time-based" or "performance-based" vesting (including, but not limited to, any unvested options, restricted stock, restricted stock units, and performance share units) such outstanding awards shall immediately vest.

e. An amount, paid on the first business day of each month, equal to 100% of the applicable monthly COBRA premium under the Company's group health plan, continued for the lesser of (i) eighteen (18) months from the date of termination of Executive's employment or (ii) the date on which Executive qualifies for health insurance as a result of employment by or association with a subsequent employer.;

f. Outplacement services for twelve (12) months from the termination date or until Executive obtains substantially comparable employment (as determined by the Company), whichever is shorter. Such outplacement services shall be commensurate with Executive's position and reasonable in amount, but not to exceed \$25,000; and

g. Notwithstanding anything in this Agreement to the contrary, if Executive is a "disqualified individual" (as defined in Section 280G(c) of the Code), and the payments and benefits provided for in this Section V.D of this Agreement, together with any other payments and benefits which Executive has the right to receive from the Company or any other person, would constitute a "parachute payment" (as defined in Section 280G(b)(2) of the Code), then the payments and benefits provided for in this Agreement shall be either (a) reduced (but not below zero) so that the present value of such total amounts and benefits received by Executive from the Company and/or such person(s) will be \$1.00 less than three (3) times Executive's "base amount" (as defined in Section 280G(b)(3) of the Code) and so that no portion of such amounts and benefits received by Executive shall be subject to the excise tax imposed by Section 4999 of the Code or (b) paid in full, whichever produces the better "net after-tax position" to Executive (taking into account any applicable excise tax under Section 4999 of the Code and any other applicable taxes). The reduction of payments and benefits hereunder, if applicable, shall be made by reducing, first, payments or benefits to be paid in cash hereunder in the order in which such payment or benefit would be paid or provided (beginning with such payment or benefit that would be made last in time and continuing, to the extent necessary, through to such payment or benefit that would be made first in time) and, then, reducing any benefit to be provided in-kind hereunder in a similar order. The determination as to whether any such reduction in the amount of the payments and benefits provided hereunder is necessary shall be made by the Company in good faith. If a reduced payment or benefit is made or provided and through error or otherwise that payment or benefit, when aggregated with other payments and benefits from the Company (or its affiliates) used in determining if a "parachute payment" exists, exceeds \$1.00 less than three (3) times Executive's base amount, then Executive shall immediately repay such excess to the Company upon notification that an overpayment has been made. Nothing in this paragraph shall require the Company to be responsible for, or have any liability or obligation with respect to, Executive's excise tax liabilities under Section 4999 of the Code.

E. By the Company Without Cause and Not for Poor Performance or Resignation by Executive for Good Reason During the Protected Period Following a Change in Control.

1. Upon the effective date of a Change in Control during the Employment Term, all of Executive's unvested incentive, performance and equity-based awards (including, but not limited to, any unvested options, restricted stock, performance, and phantom share units under the Company's equity incentive plan or any other equity plan subsequently adopted by the Company) granted to Executive after the Effective Date shall vest in full.

2. If Executive's employment is terminated by the Company without Cause (and other than by reason of Poor Performance or Executive's death or Disability) or if Executive resigns for Good Reason during the Protected Period immediately following a Change in Control, then Executive shall be entitled to receive from the Company (in lieu of any other severance payments or benefits under this Agreement), the following:

a. The Accrued Rights (refer to Section II.A);

b. Continued payment of Executive's Termination Base Salary for thirty-six (36) months ("Severance Period") following the date of such termination, payable in accordance with the Company's normal payroll practices as in effect on the date of termination;

c. An amount equal to two (2) times the Targeted Short Term Incentive Compensation of Executive for the Fiscal Year in which Executive's employment terminates; which amount shall be payable in one (1) installment due six (6) months after the date of Executive's termination of employment;

d. With respect to any outstanding equity-based awards, whether "time-based" or "performance-based" vesting (including, but not limited to, any unvested options, restricted stock, restricted stock units, and performance share units) such outstanding awards shall immediately vest.

e. An amount, paid on the first business day of each month, equal to 100% of the applicable monthly COBRA premium under the Company's group health plan, continued for the lesser of (i) eighteen (18) months from the date of termination of Executive's employment or (ii) the date on which Executive qualifies for health insurance as a result of employment by or association with a subsequent employer;

f. Outplacement services for twelve (12) months from Executive's termination date or until Executive obtains substantially comparable employment (as determined by the Company), whichever is shorter. Such outplacement services shall be commensurate with Executive's position and reasonable in amount, but not to exceed \$25,000; and

g. Benefits paid to Executive pursuant to this Section V.E of this Agreement shall be grossed-up by the Company to cover (1) any federal excise tax due by that Executive on account of these benefit payments and (2) any federal income and employment taxes due on federal excise tax.

h. Notwithstanding anything in this Agreement to the contrary, if Executive is a "disqualified individual" (as defined in Section 280G(c) of the Code), and the payments and benefits provided for in this Section IV.C of this Agreement, together with any other payments and benefits which Executive has the right to receive from the Company or any other person, would constitute a "parachute payment" (as defined in Section 280G(b)(2) of the Code), then the payments and benefits provided for in this Agreement shall be either (a) reduced (but not below zero) so that the present value of such total amounts and benefits received by Executive from the Company and/or such person(s) will be \$1.00 less than three (3) times Executive's "base amount" (as defined in Section 280G(b)(3) of the Code) and so that no portion of such amounts and benefits received by Executive shall be subject to the excise tax imposed by Section 4999 of the Code or (b) paid in full, whichever produces the better "net after-tax position" to Executive (taking into account any applicable excise tax under Section 4999 of the Code and any other applicable

taxes). The reduction of payments and benefits hereunder, if applicable, shall be made by reducing, first, payments or benefits to be paid in cash hereunder in the order in which such payment or benefit would be paid or provided (beginning with such payment or benefit that would be made last in time and continuing, to the extent necessary, through to such payment or benefit that would be made first in time) and, then, reducing any benefit to be provided in-kind hereunder in a similar order. The determination as to whether any such reduction in the amount of the payments and benefits provided hereunder is necessary shall be made by the Company in good faith. If a reduced payment or benefit is made or provided and through error or otherwise that payment or benefit, when aggregated with other payments and benefits from the Company (or its affiliates) used in determining if a "parachute payment" exists, exceeds \$1.00 less than three (3) times Executive's base amount, then Executive shall immediately repay such excess to the Company upon notification that an overpayment has been made. Nothing in this paragraph shall require the Company to be responsible for, or have any liability or obligation with respect to, Executive's excise tax liabilities under Section 4999 of the Code.

F. <u>Notice of Termination</u>. Any purported termination of employment by the Company or by Executive (other than due to Executive's death) shall be communicated by written Notice of Termination to the other party hereto in accordance with the notice provisions hereof. With respect to any termination of employment by Executive, such notice of termination shall be communicated to the Company at least thirty (30) days prior to such termination.

G. <u>Officer/Board Resignation</u>. Upon termination of Executive's employment for any reason, Executive shall be deemed hereby to have resigned, effective as of the date of such termination and to the extent applicable, from the Board (and any committees thereof) and as an officer of the Company and the board of directors (and any committees thereof) and as an officer of any and all of the Company's affiliates. As a condition to receipt of the severance benefits described herein, Executive agrees to provide written confirmation of such resignations to the Company.

H. <u>Waiver and Release</u>. Notwithstanding any other provisions of this Agreement to the contrary, unless expressly waived, in writing, by the Compensation Committee of the Board, in its sole discretion, the Company shall not make or provide, and Executive shall not be entitled to receive, any severance payments or benefits provided under this Agreement, other than the Accrued Rights, unless (i) within fifty (50) days from the date on which Executive's employment is terminated, Executive (or his estate) executes and delivers to the Company a general release (which shall be provided by the Company not later than five (5) days from the date on which Executive's employment is terminated and be substantially in the form attached hereto as Attachment B, whereby Executive (or his estate or legally appointed personal representative) releases the Company (and affiliates of the Company and other designated persons) from all employment based or related claims of Executive and all obligations of the Company to Executive other than with respect to (x) the Company's obligations to make and provide the severance payments and benefits as provided in this Agreement and (y) any vested benefits to which Executive is entitled under the terms of any Company benefit or equity plan, and (ii) Executive does not revoke such release within any applicable revocation period following Executive's delivery of the executed release to the Company. If the requirements of this Section are satisfied, then the severance payments and benefits which Executive is otherwise entitled to receive under this Agreement shall begin or be made, as applicable, without interest, on the later of (i) the sixtieth (60th) day following the date on which Executive's employment was terminated or (ii) on the tenth (10th) business day after expiration of Executive's right to revoke the release described in this section, provided that Executive does not revoke such release. If the requirements of this Section are not satisfied by Executive (or his estate) pursuant to t

I. Compliance with IRC Section 409A.

1. Notwithstanding anything in this Agreement to the contrary, if, at the time of Executive's termination of employment with the Company and its affiliates, Executive is a "specified employee," as defined in Section 409A of the Code, and the deferral of the commencement of any payments or benefits otherwise payable hereunder as a result of such termination of employment is necessary in order to avoid

the additional tax under Section 409A of the Code, then the Company will defer the payment or the commencement of any such payments or benefits hereunder (without any reduction in such payments or benefits ultimately paid or provided to Executive) until the date that is six months following Executive's termination of employment with the Company (or the earliest date as is permitted under Section 409A of the Code). Any payment amounts deferred pursuant to this Section will be accumulated and paid to Executive (without interest) in a lump sum and the balance of any remaining payments due Executive will be paid monthly or at such times as otherwise provided herein.

2. Any reimbursement of any costs and expenses by the Company to Executive under this Agreement shall be made by the Company in no event later than the close of Executive's taxable year following the taxable year in which the cost or expense is incurred by Executive. The expenses incurred by Executive in any calendar year that are eligible for reimbursement under this Agreement shall not affect the expenses incurred by Executive in any other calendar year that are eligible for reimbursement hereunder and Executive's right to receive any reimbursement hereunder shall not be subject to liquidation or exchange for any other benefit.

3. Each payment that Executive may receive under this Agreement shall be treated as a "separate payment" for purposes of Section 409A of the Code.

4. Except as provided in V.I.1, and notwithstanding anything in this Agreement to the contrary, the payment of an Annual Bonus, Performance Award, cash incentive award or equity-based award due thereunder shall be paid in all events within 2 ¹/₂ months after the end of the year in which such award (or prorated part) first becomes "vested," within the meaning of Section 409A of the Code.

5. To the extent that Section 409A of the Code applies to any terms or conditions of this Agreement, such terms and conditions shall be interpreted in a manner that is consistent with Section 409A of the Code.

VI. MISCELLANEOUS.

A. <u>Agreement Ancillary to Other Agreements</u>. This Agreement is ancillary to and part of other agreements between the Company and Executive including, the Confidentiality Agreement and the Company's agreements to: (i) disclose, and to continue to disclose its Confidential Information and Trade Secrets to Executive; (ii) provide initial and continued training, education and development to Executive; (iii) provide Executive with Confidential Information and Trade Secrets about, and the opportunity to develop relationships with, Company's employees, Customers and Suppliers, and employees and agents of its Customers and Suppliers.

B. <u>Governing Law/Venue</u>. This Agreement shall be governed by and construed in accordance with the laws of the State of Texas, without regard to conflict of laws principles thereof. Each party to this Agreement hereby irrevocably submits to the exclusive jurisdiction of the state and federal courts in Houston, Texas, for the purposes of any proceeding arising out of or based upon this Agreement.

C. <u>Arbitration</u>. Any dispute, claim or controversy arising out of or relating to this Agreement or the breach, termination, enforcement, interpretation or validity thereof shall be determined by arbitration in Houston, Harris County, Texas before one arbitrator. The arbitration shall be administered by the American Arbitration Association pursuant to its rules for the resolution of employment disputes, and the following provisions:

1. Unless otherwise ordered by the arbitrator, limited discovery consisting of one (1) deposition of each party and each expert; not more than fifteen (15) requests for production of documents; and not more than ten (10) interrogatories.

2. Subject to applicable law, the arbitrator may award attorneys' fees and the costs of arbitration to the prevailing party.

3. Anything herein to the contrary notwithstanding, either party shall have the right to seek and obtain injunctive relief to prevent a threatened breach of this Agreement, including the Confidentiality Agreement.

D. Other Agreements.

1. The Confidentiality Agreement attached hereto as Attachment A is an integral part of this Agreement, and this Agreement shall not become effective unless and until Executive has executed both this Agreement and the Confidentiality Agreement. A default under or breach of the Confidentiality Agreement shall constitute a breach of this Agreement. In addition to any and all other remedies available to Company, in the event of a breach of or default under this Agreement, or in the event that the Company obtains any form of equitable relief, order or injunction, whether temporary or permanent, for the enforcement of any of the provisions of this Agreement or the Confidentiality Agreement, the Company shall be entitled to recover, and the Executive (or his estate) shall be obligated to repay and return to the Company, upon written demand therefore, an amount equal to all severance or other benefits paid to, or on behalf of, the Executive (or his estate) pursuant to the provisions of this Agreement (other than the Accrued Rights) on or after the date of termination of Executive's employment.

2. In the event of a conflict between the rights and benefits granted by this agreement, and those granted under any other incentive, stock option, stock grant or similar plan or agreement (with the exception of the "Executive Severance Protection Plan" referred to below), Executive shall be entitled to the rights and benefits described in this agreement.

E. <u>Prior Agreements</u>. Prior to the execution of this Agreement, the Company has issued certain equity-based and/or non-equity based incentives or awards to Executive. This Agreement shall supersede or modify the terms of any such agreements only to the extent specifically stated herein. Otherwise, the terms of this Agreement and the Confidentiality Agreement referred to herein, contain the entire understanding and agreement of the parties with respect to the employment of Executive by the Company and the termination of such employment.

F. <u>No Waiver</u>. The failure of a party to insist upon strict adherence to any term of this Agreement on any occasion shall not be considered a waiver of such party's rights or deprive such party of the right thereafter to insist upon strict adherence to that term or any other term of this Agreement.

G. <u>Severability</u>. In the event that any one or more of the provisions of this Agreement shall be or become invalid, illegal or unenforceable in any respect, the validity, legality and enforceability of the remaining provisions of this Agreement shall not be affected thereby.

H. <u>Assignment</u>. Neither this Agreement nor any of Executive's rights and duties hereunder, shall be assignable or delegable by Executive. Any purported assignment or delegation by Executive in violation of the foregoing shall be null and void ab initio and of no force and effect. This Agreement may be assigned by the Company to a person or entity which is an affiliate or a successor in interest to substantially all of the business operations of the Company. Upon such assignment, the rights and obligations of the Company hereunder shall become the rights and obligations of such affiliate or successor person or entity.

I. <u>Successor Agreement</u>. At, or simultaneously with, a Change of Control (as described in this Agreement), the Company will require any successor to all or substantially all of the business and/or assets of the Company (whether direct or indirect, and whether by purchase, merger, consolidation or otherwise) to expressly assume and agree, in writing, to perform this Agreement in the same manner and to the same extent that the Company would be required to perform it if no succession had taken place. Failure of the successor to so assume this Agreement shall constitute "Good Reason as defined in Section I.F of this Agreement.

J. <u>Notices</u>. For the purpose of this Agreement, notices and all other communications provided for in the Agreement shall be in writing and shall be deemed to have been duly given on the earlier of (i) the date that such notice is delivered by hand or overnight courier or (ii) three (3) days after it has been mailed by United States registered mail, return receipt requested, postage prepaid, addressed to the respective addresses set forth below in this Agreement, or to such other address as either party may have furnished to the other in writing in accordance herewith, except that notice of change of address shall be effective only upon receipt.

IF TO THE COMPANY:

Powell Industries, Inc. Attention: Chief Executive Officer 8550 Mosley Road Houston, Texas 77075

IF TO EXECUTIVE:

Don R. Madison 3910 Canyon Bluff Ct Houston, TX 77059

K. <u>Prior Employment</u>. The Company has employed Executive for Executive's general skills, management abilities and experience in the Company's Business (as defined in the Confidentiality Agreement referred to herein). Executive acknowledges that Executive has been specifically instructed not to bring, disclose or use in any fashion any confidential information, trade secrets, proprietary information, data or technology, nor any confidential pricing information, belonging to any prior employer. In no event is Executive authorized to use or disclose any such information to the Company or any of its employees.

L. <u>Executive's Representations</u>. Executive hereby represents to the Company that (i) all confidential information, trade secrets or proprietary information, data or technology, belonging to any prior employer, including those that might have been contained on Executive's personal computer, cell phone or other electronic communications or storage device have been returned and/or deleted in accordance with any policy of or agreement with Executive's prior employer and (ii) the execution and delivery of this Agreement by Executive and the Company and the performance by Executive of Executive's duties hereunder shall not constitute a breach of, or otherwise contravene, the terms of any employment agreement or other agreement or policy to which Executive is a party or otherwise bound.

M. <u>Reimbursement of Legal Expenses</u>. The Company shall reimburse Executive for reasonable and customary fees charged by Executive's attorney to provide legal counsel review and defense concerning this Agreement, not exceed \$10,000.

N. <u>Cooperation</u>. Executive shall provide Executive's reasonable cooperation in connection with any action or proceeding (or any appeal from any action or proceeding) which relates to events occurring during Executive's employment hereunder. Executive shall be entitled to reimbursement for reasonable and customary expenses incurred for purposes of cooperating in any action or proceeding pursuant to this Section. This provision shall survive any termination of this Agreement.

O. <u>Indemnification</u>. Executive shall be indemnified by the Company against liability as an officer and director of the Company and any subsidiary or affiliate of the Company to the maximum extent permitted by the Company's bylaws by applicable law or by any indemnity agreement heretofore or hereafter executed between the Company and Executive. Executive's rights under this Section shall continue so long as Executive maybe subject to such liability, whether or not this Agreement may have terminated prior thereto. The Company will insure Executive, for the duration of his employment with the Company and thereafter with respect to his acts and omissions occurring during such employment, under a contract of director and officer liability insurance to the same extent as such insurance insures members of the Board.

P. <u>Withholding of Taxes</u>. The Company may withhold from any amounts or benefits payable under this Agreement all taxes it may be required to withhold pursuant to any applicable law or regulation.

Q. <u>Counterparts</u>. This Agreement may be signed in counterparts, each of which shall be an original, with the same effect as if the signatures thereto and hereto were upon the same instrument.

R. <u>Survival.</u> The provisions of this Agreement, together with the provisions of the Confidentiality, Non-Competition and Non-Solicitation Agreement, attached and part of this Agreement as Schedule A, shall each survive the termination of Executive's employment, regardless of how such termination is caused.

S. <u>Executive Severance Protection Plan</u>. The rights and benefits provided in this agreement are intended to replace the rights and benefits heretofore provided by the Powell Industries, Inc. Executive Severance Protection Plan adopted and effective September 20, 2002 (the Prior Plan). By execution of this agreement, Executive waives and releases any rights, benefits or payments to which Executive would have been entitled to under the Prior Plan and acknowledges that the Prior Plan has been or will be terminated by the Company.

IN WITNESS WHEREOF, THE PARTIES HERETO HAVE DULY EXECUTED THIS AGREEMENT EFFECTIVE FOR ALL PURPOSES AS OF THE EFFECTIVE DATE.

BY EXECUTIVE:

/s/ Don R. Madison Don R. Madison Date: May 8, 2012

BY POWELL INDUSTRIES, INC.:

/s/ Thomas W. Powell

Thomas W. Powell President & Chief Executive Officer Chairman of the Board

Date: May 8, 2012



CONFIDENTIALITY, NON-COMPETITION AND NON-SOLICITATION AGREEMENT

This Confidentiality, Non-Competition and Non-Solicitation Agreement ("this Agreement") is entered into between Powell Industries, Inc., on behalf of itself, and any and all of its subsidiaries, affiliates or successors (all of whom are hereinafter collectively referred to as "Company") and Don R. Madison ("Executive") in connection with and ancillary to an Executive Employment Agreement (the "Employment Agreement") being entered into between the Company and Executive of even date herewith.

I. Nondisclosure of Confidential Information and Trade Secrets

A. <u>Company's Agreements</u>. During the course of Executive's employment by the Company, the Company agrees: (i) to provide Executive with specialized training and continuing training and development regarding its products, services, methods, systems and operations; (ii) to provide Executive with access to its Confidential Information and Trade Secrets (as defined herein); and (iii) to provide Executive with Confidential Information and Trade Secrets about, and the opportunity to develop close relationships with the Company's management personnel, employees, Customers (as defined herein), Suppliers (as defined herein) and the employees, agents and representatives of Customers and Suppliers.

B. <u>Company's Business</u>. Company is engaged in the highly competitive business of the design, manufacture and packaging of equipment and systems for the distribution, control, generation and management of electrical and other power sources (all of which is hereinafter collectively referred to as the "Company's Business"). Executive acknowledges that because of the highly competitive nature of the Company's Business, the use and protection of the Company's Confidential Information and Trade Secrets as defined in this Agreement is critical to the Company's continued successful operation and business and is an essential element of this Agreement.

C. <u>Definition of Confidential Information and Trade Secrets</u>. Confidential Information and Trade Secrets, as used in this Agreement, includes, but is not limited to, written, electronic, oral and visual information relating to:

1. Lists of, and all information about, each person or entity to which Company has sold, or made a proposal to sell any products, goods, services or equipment which comprise any part of the Company's Business (all of which are hereinafter collectively referred to as "Customers");

2. Lists of, and all information about, each person or entity from which the Company has acquired equipment, inventory, components, products or services used by the Company to design, manufacture, fabricate, sell or deliver any of the products or services which comprise the Company's Business (all of which are hereinafter collectively referred to as Supplier;

3. All Customer contact information, which includes information about the identity and location of individuals with decision-making authority and the particular preferences, needs or requirements of the Customer, or such individual, with respect to any of the products, goods, services or equipment which comprise any part of the Company's Business, and all information about the particular needs or requirements of Customers based on geographical, economic or other factors;

4. Financial information of any kind about Customers, including sales and purchase histories, trend information about the growth or shrinking of a particular Customer's needs, purchases or requirements; profit margins or markups, as well as all information about the costs and expenses which the Company incurs to provide products or services to its Customers;

5. The Company's procedures, forms, methods, and systems for marketing to Customers and potential customers including all of its Customer development techniques and procedures, including training and other internal manuals, forms and documents;

6. All Supplier contact information, which includes information about the identify and location of individuals with decision-making authority and the particular capabilities, capacities, expertise, prices and/or schedules of such Suppliers;

7. All of the Company's non-public business, expansion, marketing, development, financial or budgeting plans, strategies, forecasts or proposals;

8. All of the Company's pricing and hedging formulas, methodologies, practices and systems, including those based upon particular Customers, quantities, or geographic, seasonal, economic or other factors, including all information about the price, terms, quantities or conditions of any products or services sold or furnished by the Company to its Customers;

9. Technical information about the Company, including designs, drawings, engineering and information regarding the configuration, assembly or contents of any of the Company's products or any of its hardware, equipment, tools, machinery or other manufacturing, fabrication or assembly devices or processes, or those of any of its Customers, consultants, vendors, suppliers, or any person or entity which provides manufacturing or fabrication services to the Company;

10. Any non-public financial information of any kind about the Company or its operations;

11. Information disclosed to the Company by third parties, concerning the Company's products, goods or services, bids or bidding processes, product or manufacturing specifications (except to the extent such information is publicly disclosed), contracts, procedures, or business practices;

12. Employee lists, phone numbers and addresses, pay rates, benefits and compensation packages, training programs and manuals, and other confidential information regarding the Company's personnel.

D. <u>Confidential Information and Trade Secrets</u>. Company and Executive agree that Confidential Information and Trade Secrets includes current, updated and future data, information, reports, evaluations and analyses of Company, its financial performance and results, or its Executives, including their compensation, performance or evaluation, as well as correspondence, proposals, contracts and other communications with, or financial, sales or other information about the Company's Customers and Suppliers, and includes (i) those which are provided to Executive after the date hereof, (ii) those which Executive creates, in whole or in part; (iii) those to which or for which Executive provides input or information; and (iv) those which Executive uses for the purpose of performing Executive's duties for the Company or making decisions relating to the Company's Business, its Customers, Suppliers or employees. Anything to the contrary not withstanding, however, Confidential Information and Trade Secrets shall not include (i) general industry knowledge acquired by Executive as a result of Executive's prior employment, (ii) non-confidential information acquired by Executive from any prior employment, (iii) contact information about Customers, Suppliers and others with whom Executive dealt prior to Executive's employment with the Company; and (iv) any other information generally available to the public.

E. <u>Protection of Confidential Information and Trade Secrets</u>. During the term of Executive's employment and at all times thereafter, Executive will keep all Confidential Information and Trade Secrets in strict confidence and will not use or disclose any Confidential Information and Trade Secrets for any purpose other than the performance of Executive's duties for Company. Executive will not use any Confidential Information and Trade Secrets for the gain or benefit of any person or entity other than the Company or for Executive's own personal gain or benefit. Executive will not cause the transmission, removal or transport of Confidential Information and Trade Secrets from the Company's premises except in accordance with the Company's approved procedures and then only to the extent necessary to perform Executive's duties, while employed by the Company. Executive will not provide any information about the Company's Executives to any competitor or recruiter.

II. Intellectual Property and Work Product.

A. If Executive creates, invents, designs, develops, contributes to or improves any Works, either alone or with third parties, at any time during Executive's employment by the Company and within the scope of such employment and/or with the use of any the Company resources ("Company Works"), Executive shall promptly and fully disclose same to the Company and hereby irrevocably assigns, transfers and conveys, to the maximum extent permitted by applicable law, all rights and intellectual property rights therein (including rights under patent, industrial property, copyright, trademark, trade secret, unfair competition and related laws) to the Company to the extent ownership of any such rights does not vest originally in the Company.

B. Executive agrees to keep and maintain adequate and current written records (in the form of notes, sketches, drawings, and any other form or media requested by the Company) of all Company Works. The records will be available to and remain the sole property and intellectual property of the Company at all times.

C. During the Employment Term, Executive shall take all requested actions and execute all requested documents (including any licenses or assignments required by a government contract) at the Company's expense (but without further remuneration) to assist the Company in validating, maintaining, protecting, enforcing, perfecting, recording, patenting or registering any of the Company's rights in the Prior Works and Company Works. If the Company is unable for any other reason to secure Executive's signature on any document for this purpose, then Executive hereby irrevocably designates and appoints the Company and its duly authorized officers and agents as Executive's agent and attorney in fact, to act for and in Executive's behalf and stead to execute any documents and to do all other lawfully permitted acts in connection with the foregoing.

D. Executive shall not improperly use for the benefit of, bring to any premises of, divulge, disclose, communicate, reveal, transfer or provide access to, or share with the Company any confidential, proprietary or non-public information or intellectual property relating to a former employer or other third party without the prior written permission of such third party. Executive hereby indemnifies, holds harmless and agrees to defend the Company and its officers, directors, partners, employees, agents and representatives from any breach of the foregoing covenant. Executive shall comply with all relevant policies and guidelines of the Company, including regarding the protection of confidential information and intellectual property and potential conflicts of interest. Executive acknowledges that the Company may amend any such policies and guidelines from time to time, and that Executive remains at all times bound by their most current version.

III. Non-Competition and Non-Solicitation of Customers

A. Non-Competition.

1. So long as Executive is employed by the Company or one of its affiliates, and for the greater of (i) one year from the date of the termination of Executive's employment or (ii) the "Severance Period" as defined in Section II-I of the Executive's Employment Agreement (collectively the "Restricted Period"), Executive will not, whether on Executive's own behalf or on behalf of or in conjunction with any person, firm, partnership, joint venture, association, corporation or other business organization, entity or enterprise whatsoever ("Person"), directly or indirectly:

- (a) call upon, communicate with, solicit or assist in soliciting any Customer or Supplier, or any agent or employer of either, using any Confidential Information and Trade Secrets in any way;
- (b) participate in, work on or otherwise be involved in or with any project, contract, proposal, work, sale, bid or other undertaking (collectively "Project"), if Executive worked on, participated in, was involved, or communicated with other employees of the Company, Customers, Suppliers or other third parties, with regard to any such Project during the six (6) months prior to the date of the termination of Executive's employment.

B. Non-Solicitation.

1. During the Restricted Period, Executive will not, whether on Executive's own behalf or on behalf of or in conjunction with any Person, directly or indirectly:

- (a) solicit or encourage any employee of the Company or its affiliates to leave the employment of the Company or its affiliates; or
- (b) hire any employee who was employed by the Company or its affiliates as of the date of Executive's termination of employment with the Company or who left the employment of the Company or its affiliates coincident with, or within one year prior to or after, the termination of Executive's employment with the Company.

2. During the Restricted Period, Executive will not, directly or indirectly, solicit or encourage to cease to work with the Company or its affiliates any consultant then under contract with the Company or its affiliates.

3. It is expressly understood and agreed that although Executive and the Company consider the restrictions contained in this Section 3 to be reasonable, if a final judicial determination is made by a court of competent jurisdiction that the time or territory or any other restriction contained in this Agreement is an unenforceable restriction against Executive, the provisions of this Agreement shall not be rendered void but shall be deemed amended to apply as to such maximum time and territory and to such maximum extent as such court may judicially determine or indicate to be enforceable. Alternatively, if any court of competent jurisdiction finds that any restriction contained in this Agreement is unenforceable, and such restriction cannot be amended so as to make it enforceable, such finding shall not affect the enforceability of any of the other restrictions contained herein.

IV. Company Property

Executive also agrees that all (i) correspondence, proposals, notes, reports, memoranda, records and files; (ii) plans, specifications, drawings, blueprints, and designs; (iii), training, service or other manuals; (iv) Customer or personnel lists or files, including mailing or contact lists; (v) computer software, programs, disks or files; (vi) tools, materials or equipment; (vii) photographs, photostats, negatives, undeveloped film; (viii) tape or electronic recordings (ix) information contained on any electronic storage or communications device used by Executive during Executive's employment with the Company, including those furnished by the Company and those owned by Executive, and (x) any other documents or programs, whether compiled by Executive or other Executives of the Company, or its contractors, vendors or consultants, and those which were made available to Executive while employed at the Company, which contain any Confidential Information and Trade Secrets or concern or describe any part of the Company's Business, Executive's employment or the Company's or Executive's dealings, transactions or communications with any Customers (all of which is hereinafter collectively referred to as Company Information), are and shall remain the sole and exclusive property of the Company. Executive agrees that this includes any Company Information contained on or within any personal computer, blackberry, cell phone, iPad, or any other telephonic or electronic communication or data storage device, including those owned by Executive which were used during Executive's employment with the Company (all of which are hereinafter collectively called Electronic Devices). At any time upon the Company any files, records, notes or other documents which were used during Executive's employment with the Company or which contain any Company Information. Executive will not keep in Executive's possession nor disclose nor deliver to anyone else any Company Information whether in electronic, paper or any other format.

V. Rights and Remedies Upon Breach

A. If Executive breaches any of the provisions of this Agreement, the Company will have all of the following rights and remedies, each of which shall be independent of the other and severally enforceable, and all of which shall be in addition to, and not in lieu of, any other rights and remedies available to the Company under law or in equity: (i) to have this Agreement specifically enforced by any court of competent jurisdiction; (ii) to seek and

obtain injunctive or other equitable relief of any kind, Executive hereby acknowledging and agreeing that any such breach or threatened breach will cause irreparable injury to the Company and that monetary damages will not provide an adequate remedy to the Company; (iii) to require Executive to account for and pay over to the Company all compensation, profits, monies, or other benefits derived or received by Executive as a result of any act or transaction constituting a breach of this Agreement.

B. Executive agrees and stipulates that in any action or claim brought by Executive or in any action or claim brought against Executive involving the provisions of this Agreement, Executive hereby expressly waives any claim or defense that the non-competition, non-solicitation and non-disclosure covenants contained in this Agreement are unenforceable, void or voidable, for any reason, including, but not limited to, fraud, misrepresentation, illegality, failure of consideration, illusory contract, mistake, or any other legal defense as to the validity or enforceability of this Agreement.

C. In addition to any and all other remedies available to Company, in the event of a breach of or default under this Agreement, or in the event that the Company obtains a judgment, which becomes final after the expiration of time for all appeals, that Executive has violated any of the provisions of Section II or Section III of this Attachment A, the Company shall be entitled to recover, and the Executive (or his estate) shall be obligated to repay and return to the Company, upon written demand therefore, an amount equal to all severance or other benefits paid to, or on behalf of, the Executive (or his estate) pursuant to the provisions of the Employment Agreement (other than the Accrued Rights) on or after the date of termination of Executive's employment.

VI. General Provisions

A. <u>Employment Agreement</u>. The Employment Agreement is an integral part of this Agreement, and this Agreement is an integral part of the Employment Agreement. A breach of or default under this Agreement shall constitute a material breach of the Employment Agreement; provided, however, that none of the notice requirements of the Employment Agreement shall be applicable to any actual breach of this Agreement.

B. <u>Other Agreements</u>. To the extent that Executive has heretofore entered into an agreement with the Company containing confidentiality, non disclosure, non competition and/or non-solicitation provisions, this Agreement shall constitute an amendment, modification and continuation of all such agreements and obligations, which shall be deemed to be modified as provided herein. No modification of or amendment to this Agreement, nor any waiver of rights under this Agreement, shall be effective unless it is in writing and signed by both Executive and the Company. Any subsequent change or changes in Executive's duties, salary or compensation will not affect the validity or scope of this Agreement.

C. <u>Agreement Ancillary to Other Agreements</u>. This Agreement is ancillary to and part of other agreements between the Company and the Executive, including the Employment Agreement and the Company's agreements to: (i) disclose, and to continue to disclose its Confidential Information and Trade Secrets to Executive; (ii) provide initial and continued training, education and development to Executive; (iii) provide Executive with Confidential Information and Trade Secrets about, and the opportunity to develop close relationships with the Company's management personnel, employees, Customers, Suppliers and the employee's agents and representative of Customers and Suppliers.

D. <u>Severability</u>. If one or more of the provisions in this Agreement are held to be void or unenforceable in whole or in part, the remaining provisions will continue in full force and effect. Executive further agrees that in the event the length of time, the geographic area or definition of business activity as set forth herein, is deemed unreasonable, or otherwise unenforceable, in any court proceedings, the Executive and the Court may reform, modify or reduce such restrictions such that they are reasonable and enforceable.

E. <u>"At Will" and Termination</u>. This Agreement does not alter in any way the at-will nature of employment between Executive and the Company, which may be terminated by the Company or by Executive in accordance with the terms of Executive Employment Agreement of even date herewith.

F. <u>Choice of Law</u>. This Agreement shall be construed and interpreted in accordance with the laws of the State of Texas. All obligations payable or performable hereunder shall be payable and performable at the Company's offices in Houston, Harris County, Texas.

G. <u>Enforceability</u>. This Agreement shall be enforceable by the Company, and any of its successors, assigns, affiliates, subsidiaries, parent or related corporations or entities, including any person or entity to which the Company sells, transfers or assigns all or any part of its assets, or any entity to, in or with which the Company may hereafter enter into a merger transaction of any kind. Executive shall have no right to transfer or assign Executive's rights or obligations hereunder.

H. <u>Survival</u>. The provisions of this Agreement shall survive the termination of Executive's employment by the Company, regardless of how such employment is terminated.

BY SIGNING BELOW, EXECUTIVE REPRESENTS THAT EMPLOYEE HAS READ THIS AGREEMENT CAREFULLY AND UNDERSTANDS AND AGREES TO ITS TERMS, INCLUDING THOSE LIMITING EXECUTIVE'S RIGHTS TO SOLICIT CUSTOMERS OR EXECUTIVES OF THE COMPANY.

BY EXECUTIVE:

May 8, 2012 Date

BY POWELL INDUSTRIES, INC.:

May 8, 2012 Date

/s/ Don R. Madison

Don R. Madison Executive Vice President & Chief Financial Officer /s/ Thomas W. Powell

Thomas W. Powell President & Chief Executive Officer Chairman of the Board



EXECUTIVE EMPLOYMENT AGREEMENT

THIS EMPLOYMENT AGREEMENT and an ancillary agreement to be effective simultaneously herewith entitled "Confidentiality, Non-Competition and Non-Solicitation Agreement" (the "Confidentiality Agreement"), a copy of which is attached hereto as Attachment A, and incorporated herein by reference for all purposes, (this agreement and the Confidentiality Agreement being hereinafter collectively referred to as "this Agreement") is entered into effective as of May 8, 2012 (the "Effective Date"), by and between Powell Industries, Inc. and its affiliates (the "Company") and Milburn E. Honeycutt ("Executive").

WHEREAS, the Company desires to continue the employment of Executive as Vice President & Chief Accounting Officer of the Company from and after the Effective Date until such date as his employment shall end pursuant to the terms and conditions contained herein; and

WHEREAS, Executive desires to continue employment with the Company in such position pursuant to the terms and conditions contained herein;

NOW, THEREFORE, IT IS HEREBY AGREED AS FOLLOWS:

I. EMPLOYMENT TERM.

The term of this Agreement shall commence on the Effective Date and expire at the earlier of:

a. The last day of the month in which the Executive reaches age 65, in which case, unless the parties agree otherwise, Executive's employment shall continue at will and shall be terminable by either party, for any reason, in which event the executive shall be entitled to the severance benefits provided for under Section V.B; or

b. The date Executive's employment terminates subject to the provisions of this Agreement regarding termination, resignation or retirement.

Executive and the Company acknowledge that the employment relationship provided herein may be terminated at any time, upon written notice to the other party for any reason, at the option of either the Company or Executive. However, as provided in this Agreement, Executive may be entitled to certain severance benefits depending upon the circumstances of Executive's termination of employment. The period Executive is employed by the Company under this Agreement is referred to herein as the "Employment Term."

II. CERTAIN DEFINITIONS

A. "Accrued Rights" shall mean:

1. Executive's earned, but unpaid compensation, to include base salary, vehicle allowance, short term incentive and long term incentive compensation through the date of termination;

2. Reimbursement, within sixty (60) days following submission by Executive to the Company of appropriate supporting documentation, for any unreimbursed reasonable business expenses properly incurred by Executive in the performance of Executive's duties in accordance with the Company's expense reimbursement policy prior to the date of Executive's termination, provided claims for such reimbursement (accompanied by appropriate supporting documentation) are submitted to the Company within ninety (90) days following the date such expenses were incurred and within thirty (30) days following Executive's termination; and

3. Such Employee Benefits, if any, as to which Executive may be entitled under the terms of the employee benefit plans of the Company.

B. "<u>Cause</u>" shall mean:

1. Executive's conviction of (or plea of nolo contendere to) a felony;

2. Executive's dishonesty, theft, embezzlement or fraud with respect to the business, property, reputation or affairs of the Company;

3. Executive's willful violation of the Company's Business Code of Conduct and Business Ethics and/or any other of the Company's employment, personnel, safety or other policies as now exist or as may hereafter be amended;

4. Executive's having committed any material violation of any federal or state law regulating securities (without having relied on the advice of the Company's attorney or outside auditor) or having been the subject of any final order, judicial or administrative, obtained or issued by the Securities and Exchange Commission, or any regulatory authority having jurisdiction over the Company's securities for any securities violation involving fraud, including, without limitation, any such order consented to by Executive in which findings of facts or any legal conclusions establishing liability are neither admitted nor denied;

5. Executive's willful and continued failure to devote substantially all of his business time to the Company's business affairs (excluding failures due to illness, incapacity, vacations, incidental civic activities and incidental personal time); or

6. Executive's unauthorized disclosure of confidential information of the Company that is materially injurious to the Company.

Notwithstanding the above, however, and except with regard to the events described in subparagraph (1) above, Cause shall not exist with respect to any matter unless the Company gives the Executive written notice of such matter within ninety (90) days of the date the Company knew of its occurrence. Such notice shall specify with reasonable particularity the acts, events or conditions which are claimed to constitute Cause. If the Company fails to give such notice timely, the Company shall be deemed to have waived its right to terminate Executive for Cause with respect to such matter.

Upon receipt of the notice described above, Executive shall have thirty (30) days to (i) cure or correct the acts, event or conditions specified in the notice, (ii) commence Executive's best efforts to cure or correct the event constituting such and continue such efforts until the act, event or condition is cured; or (iii) if applicable, provide the Company with written evidence or documentation that the acts or events claimed to constitute Cause did not occur, or were not performed or omitted by Executive, or otherwise do not constitute Cause as described in this Agreement.

For purposes of this definition, no act, or failure to act, on Executive's part shall be deemed "willful" unless done, or omitted to be done, by Executive not in good faith and without reasonable belief that Executive's action or omission was in the best interest of the Company.

C. "Change of Control" shall mean any of the following:

1. any "person" (as such term is used in Section 13(d) and 14(d) of the Securities Exchange Act of 1934, as amended (the "Exchange Act")), (other than a trustee or other fiduciary holding securities under an employee benefit plan of the Company or any affiliate, or any corporation owned, directly or

indirectly, by the stockholders of the Company in substantially the same proportions as their ownership of stock of the Company), acquires "beneficial ownership" (within the meaning of Rule 13d-3 under the Exchange Act) of securities of the Company representing 35% or more of the combined voting power of the Company's then outstanding securities; provided, however, that if the Company engages in a merger or consolidation in which the Company or surviving entity in such merger or consolidation becomes a subsidiary of another entity, then references to the Company's then outstanding securities shall be deemed to refer to the outstanding securities of such parent entity;

2. a change in the composition of the Board, as a result of which fewer than a majority of the directors are Incumbent Directors. "Incumbent Directors" shall mean directors who either (i) are directors of the Company as of the Effective Date, or (ii) are elected, or nominated for election, to the Board with the affirmative votes of at least two-thirds of the Incumbent Directors at the time of such election or nomination, but Incumbent Director shall not include an individual whose election or nomination occurs as a result of either (1) an actual or threatened election contest (as such terms are used in Rule 14a-11 of Regulation 14A promulgated under the Exchange Act) or (2) an actual or threatened solicitation of proxies or consents by or on behalf of a person other than the Board of Directors of the Company;

3. the consummation of a merger or consolidation of the Company with any other corporation, other than a merger or consolidation which would result in the voting securities of the Company outstanding immediately prior thereto continuing to represent (either by remaining outstanding or by being converted into voting securities of the surviving entity (or if the surviving entity is or shall become a subsidiary of another entity, then such parent entity) more than 50% of the combined voting power of the voting securities of the Company (or such surviving entity or parent entity, as the case may be) outstanding immediately after such merger or consolidation;

4. the stockholders of the Company approve a plan of complete liquidation of the Company; or

5. the sale or disposition (other than a pledge or similar encumbrance) by the Company of all or substantially all of the assets of the Company other than to a subsidiary or subsidiaries of the Company.

D. "<u>Date of Termination</u>" shall mean the date the Notice of Termination is given unless such Notice of Termination is by Executive in which event the Date of Termination shall not be less than 30 days following the date the Notice of Termination is given. Further, a Notice of Termination given by Executive due to a Good Reason event that is corrected by the Company before the Date of Termination shall be void.

E. "<u>Disability</u>" shall mean that Executive: (i) is unable to perform the essential functions of Executive's job title and duties by reason of any medically determinable physical or mental impairment which can be expected to result in death or can be expected to last for a continuous period of not less than twelve (12) months, provided that Executive or his representative has provided the Company with certification of such disability from a licensed physician or other medical services provider acceptable to the Company in its sole discretion; (ii) is, by reason of any medically determinable physical or mental impairment which can be expected to last for a continuous period of not less than twelve (12) months, receiving income replacement benefits for a period of not less than three months under an accident and health plan or disability insurance policy covering employees of the Company; or (iii) is determined by the Social Security Administration to be disabled.

F. "Good Reason" shall mean:

1. a material reduction in Executive's authority, duties or responsibilities or the assignment to Executive **of** duties or responsibilities inconsistent in any material respect from those of Executive in effect immediately prior to the change;

2. a material reduction of Executive's compensation and benefits, including, without limitation, annual base salary, targeted short-term incentive compensation, targeted long-term incentive compensation, and equity incentive opportunities, from those in effect immediately prior to the change;

3. the Company fails to obtain a written agreement from any successor or assigns of the Company to assume and perform this Agreement as provided in Section VI.I hereof;

4. the Company requires Executive, without Executive's consent, to be based at any office located more than 50 miles from the Company's offices to which Executive was based immediately prior to the Change of Control, except for travel reasonably required in the performance of Executive's duties; or

5. the Company's breach of a material term of this Agreement.

Notwithstanding the above, however, Good Reason shall not exist with respect to any matter unless the Executive gives the Company written notice of such matter within ninety (90) days of the date the Executive knew or reasonably should have known of its occurrence. Such notice shall specify with reasonable particularity, the acts, events or conditions which are claimed to constitute Good Reason. If the Executive fails to give such notice timely, the Executive shall be deemed to have waived Executive's right to resign for Good Reason with respect to such matter.

Upon receipt of the notice described above, the Company shall have sixty (60) days to (i) cure or correct the acts, event or conditions specified in the notice, (ii) commence the Company's best efforts to cure or corrects the event constituting such and continue such efforts until the act, event or condition is cured; or (iii) if applicable, provide the Executive with written evidence or documentation that the acts or events claimed to constitute Good Reason did not occur, or otherwise do not constitute Good Reason as described in this Agreement.

For purposes of this Agreement, "Good Reason" shall be construed to refer to Executive's positions, duties, and responsibilities in the position or positions in which Executive was serving before any event as described in subparagraphs (1) through (5) above, which shall not include titles or positions with subsidiaries and affiliates of the Company that are held primarily for administrative convenience.

"Good Reason" shall also include any of the foregoing acts or omissions by a successor in interest to the Company as referenced in Sections II.C(3), (4) or (5) above.

G. "<u>Notice of Termination</u>" shall mean a written notice delivered to the other party indicating the specific termination provision in this Agreement relied upon for termination of Executive's employment which shall set forth in reasonable detail the facts and circumstances claimed to provide a basis for termination of Executive's employment under the provision so indicated. For the purpose, termination of Executive's employment shall be interpreted consistent with the meaning of the term "Separation from Service" in Section 409A(a)(2)(A)(i) of the Internal Revenue Code of 1986, as amended (the "Code") and applicable regulation authority.

H. "<u>Poor Performance</u>" shall mean Executive's willful and continued failure to perform substantially the duties of Executive's position after a written demand for substantial performance is delivered to him which specifically identifies the nature of such unacceptable performance, and which is not cured by Executive within a reasonable period, not to exceed sixty (60) days. For purposes of the definition in of "Poor Performance" as used herein, no act, or failure to act, on Executive's part shall be deemed "willful" unless done, or omitted to be done, by Executive not in good faith and without reasonable belief that Executive's action or omission was in the best interest of the Company.

I. "Protected Period" shall mean the 24-month period beginning on the effective date of a Change of Control.

J. "<u>Retirement</u>" shall mean Executive has reached 62 years of age ("normal retirement") or age 60 with at least five (5) years of active service ("early retirement"); provided, however that Executive cannot be required to retire and must consent in writing to any Retirement.

K. "<u>Severance Period</u>" shall mean the time period during which the Executive receives salary continuation benefits following a termination of employment by the Company for Poor Performance as described in Section V.C Without Cause or Resignation by Executive for Good Reason either prior to a Change in Control as described Section V.D or after a Change in Control as described in Section V.E.

L. "<u>Targeted STIC</u>" shall mean the targeted value of Executive's annual Short Term Incentive Compensation opportunity for the year in which the Date of Termination occurs, or the target value in place prior to a material reduction in compensation, or the fiscal year immediately preceding a Change of Control whichever is a greater amount.

M. "<u>Targeted LTIC</u>" shall mean the targeted value of Executive's annual Long Term Incentive Compensation opportunity for the year in which the Date of Termination occurs, or the target value in place prior to a material reduction in compensation, or the fiscal year immediately preceding a Change of Control, whichever is a greater amount.

N. "<u>Termination Base Salary</u>" shall be the greater of, the Executive's base salary at the rate in effect at the time the Notice of Termination, or the Executive's base salary in place prior to a material reduction in compensation, or the Executive's base salary in effect immediately prior to a Change of Control.

III. POSITION.

A. During the Employment Term, Executive shall serve as the Company's Vice President & Chief Accounting Officer. In such position, Executive shall report to Chief Financial Officer, or as directed by the Board of Directors of the Company, and shall have the authority, responsibilities, and duties reasonably accorded to, expected of and consistent with Executive's position, as may be assigned to Executive. The Executive's principal place of employment shall be the principal offices of the Company currently located in Houston; <u>provided</u>, <u>however</u>, that Executive understands and agrees that Executive will be required to travel from time to time for business reasons.

B. During the Employment Term, Executive shall devote his full business time, attention and efforts to the performance of Executive's duties hereunder and will not engage in any other activity (for compensation or otherwise without written notice to, and the written consent of the Board of Directors of the Company (the "Board")) which, in the good faith opinion of the Board, could, either individually or in the aggregate, reasonably be expected to conflict or interfere with or otherwise adversely affect the rendition of such performance either directly or indirectly. The foregoing limitations shall not be construed as prohibiting Executive from making personal investments in such form or manner as will neither require Executive's services in the operation or affairs of the companies or businesses in which such investments are made nor violate the terms of this Agreement hereof or otherwise conflict or interfere with Executive's responsibilities to the Company; provided, however, that Executive agrees he will not join any boards (other than community and civic boards which do not interfere with his duties to the Company) without the prior written approval of the Board.

IV. COMPENSATION.

A. <u>Base Salary</u>. The Company shall pay Executive a base salary at the annual rate of \$200,850 payable in accordance with the Company's payroll practices for similarly situated executives (the "Base Salary"). Executive's Base Salary shall be subject to review annually by and at the sole discretion of the Compensation Committee of the Board (the "Compensation Committee").

B. <u>Short Term Incentive Compensation Award</u>. For each fiscal year ("Fiscal Year") of the Company during the Employment Term, Executive shall be given the opportunity to earn annual Short Term Incentive Compensation Award (the "STIC Award"). Executive's annual Short Term Incentive Compensation opportunity for each Fiscal Year ending during the Employment Term shall be set by the Compensation Committee, in its sole discretion. The actual STIC Award payable to Executive with respect to a Fiscal Year shall be dependent upon the achievement of performance objectives established by the Compensation Committee for such Fiscal Year and may be greater or less than the Short Term Incentive Compensation opportunity depending on performance objective results. The Compensation Committee shall also have the sole right to determine whether Executive may be entitled to a discretionary bonus at any time and to determine the criteria to be considered in making such decision. Except as otherwise provided in this Agreement, the payment of STIC Award shall be at the same time as Short Term Incentive Compensation Awards are paid to other similar executives of the Company.



C. Long Term Incentive Compensation Award. During the Employment Term, Executive shall be shall be given the opportunity to earn annual Long Term Incentive Compensation Award (the "Target LTIC Award") under the Company's Equity Incentive Plan (the "Equity Plan"), as modified, amended or replaced from time to time. Executive's annual Targeted Long Term Incentive Compensation Award for each Fiscal Year during the Employment Term shall be set by the Compensation Committee, in its sole discretion. The actual LTIC Award payable to Executive with respect to a Fiscal Year shall be dependent upon the achievement of performance objectives established by the Compensation Committee for such Fiscal Year and may be greater or less than the Target Long Term Incentive Compensation opportunity depending on performance objective results. Except as otherwise provided in this Agreement, the payment of LTIC Award shall be at the same time as Long Term Incentive Compensation Awards are paid to other similar executives of the Company.

D. <u>Employee Benefits</u>. During the Employment Term, Executive shall be eligible to participate in the Company's employee benefit plans as in effect from time to time (collectively, "Employee Benefits") on the same basis as such employee benefit plans are generally made available to other comparable executives of the Company.

E. <u>Vacation</u>. Executive shall be entitled to four (4) weeks of annual vacation leave for each Fiscal Year during which Executive is employed (prorated for Executive's initial year, if not a full year). Such leave shall be administered in accordance with the Company's vacation policy.

F. <u>Automobile Allowance</u>. During the Employment Term, Executive shall be entitled to an automobile allowance of \$2,000 per month paid in accordance with the Company's normal payroll practices.

G. <u>Business Expenses</u>. During the Employment Term, reasonable business expenses incurred by Executive in the performance of Executive's duties hereunder shall be reimbursed by the Company in accordance with the Company's expense reimbursement policy.

V. TERMINATION OF EMPLOYMENT.

Executive shall not have a "termination of employment" for purposes of this Agreement unless such termination constitutes a "separation from service" for purposes of Section 409A of the Internal Revenue Code of 1986, as amended, and the applicable Treasury Regulations thereunder (the "Code"). Notwithstanding any other provision of this Agreement, the provisions of this Section V shall exclusively govern Executive's rights upon termination of employment with the Company and its affiliates.

A. By the Company for Cause or Resignation by Executive Without Good Reason.

1. The Employment Term and Executive's employment hereunder may be terminated by the Company for Cause or by Executive's resignation without Good Reason.

2. If Executive's employment is terminated by the Company for Cause, or if Executive resigns without Good Reason, then, subject to the further terms of this Agreement, Executive shall be entitled to receive:

a. The Accrued Rights (refer to Section II.A)

B. Retirement, Disability or Death.

1. The Employment Term and Executive's employment hereunder shall terminate upon Executive's Retirement, Disability or Death; provided, however, that if Executive retires under circumstances that would constitute "Good Reason", Executive shall be deemed to have terminated for "Good Reason" and be entitled to the applicable rights and benefits provided in this agreement.

2. Upon termination of Executive's employment hereunder for either Retirement, Disability or Death, then Executive or Executive's estate (as the case may be) shall be entitled to receive the following:

a. The Accrued Rights (refer to Section II.A); and

b. A prorated portion of the Targeted STIC for the current Fiscal Year, prorated based on the percentage of the current Fiscal Year that shall have elapsed through the date of termination; and

c. With respect to any outstanding equity-based awards, whether "time-based" or "performance-based" vesting (including, but not limited to, any unvested options, restricted stock, restricted stock units, and performance share units) such outstanding awards shall immediately vest; and

d. In the event of termination for Disability or Death, an amount, paid on the first business day of each month, equal to 100% of the applicable monthly COBRA premium under the Company's group health plan, continued for the lesser of (i) twelve (12) months or (ii) until such COBRA coverage for Executive terminates.

C. By the Company for Poor Performance.

1. The Employment Term and Executive's employment hereunder may be terminated by the Company for Poor Performance.

2. If Executive's Employment is terminated by the Company for Poor Performance then Executive shall be entitled to receive from the Company the following:

a. The Accrued Rights (refer to Section II.A);

b. Continued payment of Executive's Termination Base Salary for twelve (12) months (the "Severance Period") following the date of such termination, payable in accordance with the Company's normal payroll practices as in effect on the date of termination;

c. With respect to any outstanding equity-based awards, whether "time-based" or "performance-based" vesting (including, but not limited to, any unvested options, restricted stock, restricted stock units, and performance share units) such outstanding awards shall be forfeited; and

d. An amount, paid on the first business day of each month equal to one hundred percent (100%) of the applicable COBRA premium under the Company's group health plan, continued for the lesser of (1) twelve (12) months from the date of termination of Executive's employment; or (2) the date on which Executive qualifies for health insurance as a result of employment by or association with a subsequent employer.

D. By the Company Without Cause and not for Poor Performance or Resignation by Executive for Good Reason Prior to a Change in Control.

1. The Employment Term and Executive's employment hereunder may be terminated by the Company without Cause or by Executive's resignation for Good Reason.

2. If Executive's employment is terminated by the Company without Cause (and other than by reason of Executive's death or Disability) or if Executive resigns for Good Reason, then Executive shall be entitled to receive from the Company the following:

a. The Accrued Rights (refer to Section II.A);

b. Continued payment of Executive's Termination Base Salary for twelve (12) months (the "Severance Period") following the date of such termination, payable in accordance with the Company's normal payroll practices as in effect on the date of termination;

c. An amount equal to one (1) times the Target Short Term Incentive Compensation of Executive for the Fiscal Year in which Executive's employment terminates, which amount shall be payable in one (1) installment due six (6) months after the date of Executive's termination of employment;

d. With respect to any outstanding equity-based awards, whether "time-based" or "performance-based" vesting (including, but not limited to, any unvested options, restricted stock, restricted stock units, and performance share units) such outstanding awards shall immediately vest.

e. An amount, paid on the first business day of each month, equal to 100% of the applicable monthly COBRA premium under the Company's group health plan, continued for the lesser of (i) eighteen (18) months from the date of termination of Executive's employment or (ii) the date on which Executive qualifies for health insurance as a result of employment by or association with a subsequent employer;

f. Outplacement services for twelve (12) months from the termination date or until Executive obtains substantially comparable employment (as determined by the Company), whichever is shorter. Such outplacement services shall be commensurate with Executive's position and reasonable in amount, but not to exceed \$25,000; and

g. Notwithstanding anything in this Agreement to the contrary, if Executive is a "disqualified individual" (as defined in Section 280G(c) of the Code), and the payments and benefits provided for in this Section V.D of this Agreement, together with any other payments and benefits which Executive has the right to receive from the Company or any other person, would constitute a "parachute payment" (as defined in Section 280G(b)(2) of the Code), then the payments and benefits provided for in this Agreement shall be either (a) reduced (but not below zero) so that the present value of such total amounts and benefits received by Executive from the Company and/or such person(s) will be \$1.00 less than three (3) times Executive's "base amount" (as defined in Section 280G(b)(3) of the Code) and so that no portion of such amounts and benefits received by Executive shall be subject to the excise tax imposed by Section 4999 of the Code or (b) paid in full, whichever produces the better "net after-tax position" to Executive (taking into account any applicable excise tax under Section 4999 of the Code and any other applicable taxes). The reduction of payments and benefits hereunder, if applicable, shall be made by reducing, first, payments or benefits to be paid in cash hereunder in the order in which such payment or benefit would be paid or provided (beginning with such payment or benefit that would be made last in time and continuing, to the extent necessary, through to such payment or benefit that would be made first in time) and, then, reducing any benefit to be provided in-kind hereunder in a similar order. The determination as to whether any such reduction in the amount of the payments and benefits provided hereunder is necessary shall be made by the Company in good faith. If a reduced payment or benefit is made or provided and through error or otherwise that payment or benefit, when aggregated with other payments and benefits from the Company (or its affiliates) used in determining if a "parachute payment" exists, exceeds \$1.00 less than three (3) times Executive's base amount, then Executive shall immediately repay such excess to the Company upon notification that an overpayment has been made. Nothing in this paragraph shall require the Company to be responsible for, or have any liability or obligation with respect to, Executive's excise tax liabilities under Section 4999 of the Code.

E. By the Company Without Cause and Not for Poor Performance or Resignation by Executive for Good Reason During the Protected Period Following a Change in Control.

1. Upon the effective date of a Change in Control during the Employment Term, all of Executive's unvested incentive, performance and equity-based awards (including, but not limited to, any unvested options, restricted stock, performance, and phantom share units under the Company's equity incentive plan or any other equity plan subsequently adopted by the Company) granted to Executive after the Effective Date shall vest in full.

2. If Executive's employment is terminated by the Company without Cause (and other than by reason of Poor Performance or Executive's death or Disability) or if Executive resigns for Good Reason during the Protected Period immediately following a Change in Control, then Executive shall be entitled to receive from the Company (in lieu of any other severance payments or benefits under this Agreement), the following:

a. The Accrued Rights (refer to Section II.A);

b. Continued payment of Executive's Termination Base Salary for twenty-four (24) months ("Severance Period") following the date of such termination, payable in accordance with the Company's normal payroll practices as in effect on the date of termination;

c. An amount equal to two (2) times the Targeted Short Term Incentive Compensation of Executive for the Fiscal Year in which Executive's employment terminates; which amount shall be payable in one (1) installment due six (6) months after the date of Executive's termination of employment;

d. With respect to any outstanding equity-based awards, whether "time-based" or "performance-based" vesting (including, but not limited to, any unvested options, restricted stock, restricted stock units, and performance share units) such outstanding awards shall immediately vest.

e. An amount, paid on the first business day of each month, equal to 100% of the applicable monthly COBRA premium under the Company's group health plan, continued for the lesser of (i) eighteen (18) months from the date of termination of Executive's employment or (ii) the date on which Executive qualifies for health insurance as a result of employment by or association with a subsequent employer;

f. Outplacement services for twelve (12) months from Executive's termination date or until Executive obtains substantially comparable employment (as determined by the Company), whichever is shorter. Such outplacement services shall be commensurate with Executive's position and reasonable in amount, but not to exceed \$25,000; and

g. Benefits paid to Executive pursuant to this Section V.E of this Agreement shall be grossed-up by the Company to cover (1) any federal excise tax due by that Executive on account of these benefit payments and (2) any federal income and employment taxes due on federal excise tax.

h. Notwithstanding anything in this Agreement to the contrary, if Executive is a "disqualified individual" (as defined in Section 280G(c) of the Code), and the payments and benefits provided for in this Section IV.C of this Agreement, together with any other payments and benefits which Executive has the right to receive from the Company or any other person, would constitute a "parachute payment" (as defined in Section 280G(b)(2) of the Code), then the payments and benefits provided for in this Agreement shall be either (a) reduced (but not below zero) so that the present value of such total amounts and benefits received by Executive from the Company and/or such person(s) will be \$1.00 less than three (3) times Executive's "base amount" (as defined in Section 280G(b)(3) of the Code) and so that no portion of such amounts and benefits received by Executive shall be subject to the excise tax imposed by Section 4999 of the Code or (b) paid in full, whichever produces the better "net after-tax position" to Executive (taking into account any applicable excise tax under Section 4999 of the Code and any other applicable

taxes). The reduction of payments and benefits hereunder, if applicable, shall be made by reducing, first, payments or benefits to be paid in cash hereunder in the order in which such payment or benefit would be paid or provided (beginning with such payment or benefit that would be made last in time and continuing, to the extent necessary, through to such payment or benefit that would be made first in time) and, then, reducing any benefit to be provided in-kind hereunder in a similar order. The determination as to whether any such reduction in the amount of the payments and benefits provided hereunder is necessary shall be made by the Company in good faith. If a reduced payment or benefit is made or provided and through error or otherwise that payment or benefit, when aggregated with other payments and benefits from the Company (or its affiliates) used in determining if a "parachute payment" exists, exceeds \$1.00 less than three (3) times Executive's base amount, then Executive shall immediately repay such excess to the Company upon notification that an overpayment has been made. Nothing in this paragraph shall require the Company to be responsible for, or have any liability or obligation with respect to, Executive's excise tax liabilities under Section 4999 of the Code.

F. <u>Notice of Termination</u>. Any purported termination of employment by the Company or by Executive (other than due to Executive's death) shall be communicated by written Notice of Termination to the other party hereto in accordance with the notice provisions hereof. With respect to any termination of employment by Executive, such notice of termination shall be communicated to the Company at least thirty (30) days prior to such termination.

G. <u>Officer/Board Resignation</u>. Upon termination of Executive's employment for any reason, Executive shall be deemed hereby to have resigned, effective as of the date of such termination and to the extent applicable, from the Board (and any committees thereof) and as an officer of the Company and the board of directors (and any committees thereof) and as an officer of any and all of the Company's affiliates. As a condition to receipt of the severance benefits described herein, Executive agrees to provide written confirmation of such resignations to the Company.

H. <u>Waiver and Release</u>. Notwithstanding any other provisions of this Agreement to the contrary, unless expressly waived, in writing, by the Compensation Committee of the Board, in its sole discretion, the Company shall not make or provide, and Executive shall not be entitled to receive, any severance payments or benefits provided under this Agreement, other than the Accrued Rights, unless (i) within fifty (50) days from the date on which Executive's employment is terminated, Executive (or his estate) executes and delivers to the Company a general release (which shall be provided by the Company not later than five (5) days from the date on which Executive's employment is terminated and be substantially in the form attached hereto as Attachment B, whereby Executive (or his estate or legally appointed personal representative) releases the Company (and affiliates of the Company and other designated persons) from all employment based or related claims of Executive and all obligations of the Company to Executive other than with respect to (x) the Company's obligations to make and provide the severance payments and benefits as provided in this Agreement and (y) any vested benefits to which Executive is entitled under the terms of any Company benefit or equity plan, and (ii) Executive does not revoke such release within any applicable revocation period following Executive's delivery of the executed release to the Company. If the requirements of this Section are satisfied, then the severance payments and benefits which Executive is otherwise entitled to receive under this Agreement shall begin or be made, as applicable, without interest, on the later of (i) the sixtieth (60th) day following the date on which Executive's employment was terminated or (ii) on the tenth (10th) business day after expiration of Executive's right to revoke the release described in this section, provided that Executive does not revoke such release. If the requirements of this Section are not satisfied by Executive (or his estate) pursuant to t

I. Compliance with IRC Section 409A.

1. Notwithstanding anything in this Agreement to the contrary, if, at the time of Executive's termination of employment with the Company and its affiliates, Executive is a "specified employee," as defined in Section 409A of the Code, and the deferral of the commencement of any payments or benefits otherwise payable hereunder as a result of such termination of employment is necessary in order to avoid

the additional tax under Section 409A of the Code, then the Company will defer the payment or the commencement of any such payments or benefits hereunder (without any reduction in such payments or benefits ultimately paid or provided to Executive) until the date that is six months following Executive's termination of employment with the Company (or the earliest date as is permitted under Section 409A of the Code). Any payment amounts deferred pursuant to this Section will be accumulated and paid to Executive (without interest) in a lump sum and the balance of any remaining payments due Executive will be paid monthly or at such times as otherwise provided herein.

2. Any reimbursement of any costs and expenses by the Company to Executive under this Agreement shall be made by the Company in no event later than the close of Executive's taxable year following the taxable year in which the cost or expense is incurred by Executive. The expenses incurred by Executive in any calendar year that are eligible for reimbursement under this Agreement shall not affect the expenses incurred by Executive in any other calendar year that are eligible for reimbursement hereunder and Executive's right to receive any reimbursement hereunder shall not be subject to liquidation or exchange for any other benefit.

3. Each payment that Executive may receive under this Agreement shall be treated as a "separate payment" for purposes of Section 409A of the Code.

4. Except as provided in V.I.1, and notwithstanding anything in this Agreement to the contrary, the payment of an Annual Bonus, Performance Award, cash incentive award or equity-based award due thereunder shall be paid in all events within 2 ¹/₂ months after the end of the year in which such award (or prorated part) first becomes "vested," within the meaning of Section 409A of the Code.

5. To the extent that Section 409A of the Code applies to any terms or conditions of this Agreement, such terms and conditions shall be interpreted in a manner that is consistent with Section 409A of the Code.

VI. MISCELLANEOUS.

A. <u>Agreement Ancillary to Other Agreements</u>. This Agreement is ancillary to and part of other agreements between the Company and Executive including, the Confidentiality Agreement and the Company's agreements to: (i) disclose, and to continue to disclose its Confidential Information and Trade Secrets to Executive; (ii) provide initial and continued training, education and development to Executive; (iii) provide Executive with Confidential Information and Trade Secrets about, and the opportunity to develop relationships with, Company's employees, Customers and Suppliers, and employees and agents of its Customers and Suppliers.

B. <u>Governing Law/Venue</u>. This Agreement shall be governed by and construed in accordance with the laws of the State of Texas, without regard to conflict of laws principles thereof. Each party to this Agreement hereby irrevocably submits to the exclusive jurisdiction of the state and federal courts in Houston, Texas, for the purposes of any proceeding arising out of or based upon this Agreement.

C. <u>Arbitration</u>. Any dispute, claim or controversy arising out of or relating to this Agreement or the breach, termination, enforcement, interpretation or validity thereof shall be determined by arbitration in Houston, Harris County, Texas before one arbitrator. The arbitration shall be administered by the American Arbitration Association pursuant to its rules for the resolution of employment disputes, and the following provisions:

1. Unless otherwise ordered by the arbitrator, limited discovery consisting of one (1) deposition of each party and each expert; not more than fifteen (15) requests for production of documents; and not more than ten (10) interrogatories.

2. Subject to applicable law, the arbitrator may award attorneys' fees and the costs of arbitration to the prevailing party.

3. Anything herein to the contrary notwithstanding, either party shall have the right to seek and obtain injunctive relief to prevent a threatened breach of this Agreement, including the Confidentiality Agreement.

D. Other Agreements.

1. The Confidentiality Agreement attached hereto as Attachment A is an integral part of this Agreement, and this Agreement shall not become effective unless and until Executive has executed both this Agreement and the Confidentiality Agreement. A default under or breach of the Confidentiality Agreement shall constitute a breach of this Agreement. In addition to any and all other remedies available to Company, in the event of a breach of or default under this Agreement, or in the event that the Company obtains any form of equitable relief, order or injunction, whether temporary or permanent, for the enforcement of any of the provisions of this Agreement or the Confidentiality Agreement, the Company shall be entitled to recover, and the Executive (or his estate) shall be obligated to repay and return to the Company, upon written demand therefore, an amount equal to all severance or other benefits paid to, or on behalf of, the Executive (or his estate) pursuant to the provisions of this Agreement (other than the Accrued Rights) on or after the date of termination of Executive's employment.

2. In the event of a conflict between the rights and benefits granted by this agreement, and those granted under any other incentive, stock option, stock grant or similar plan or agreement (with the exception of the "Executive Severance Protection Plan" referred to below), Executive shall be entitled to the rights and benefits described in this agreement.

E. <u>Prior Agreements</u>. Prior to the execution of this Agreement, the Company has issued certain equity-based and/or non-equity based incentives or awards to Executive. This Agreement shall supersede or modify the terms of any such agreements only to the extent specifically stated herein. Otherwise, the terms of this Agreement and the Confidentiality Agreement referred to herein, contain the entire understanding and agreement of the parties with respect to the employment of Executive by the Company and the termination of such employment.

F. <u>No Waiver</u>. The failure of a party to insist upon strict adherence to any term of this Agreement on any occasion shall not be considered a waiver of such party's rights or deprive such party of the right thereafter to insist upon strict adherence to that term or any other term of this Agreement.

G. <u>Severability</u>. In the event that any one or more of the provisions of this Agreement shall be or become invalid, illegal or unenforceable in any respect, the validity, legality and enforceability of the remaining provisions of this Agreement shall not be affected thereby.

H. <u>Assignment</u>. Neither this Agreement nor any of Executive's rights and duties hereunder, shall be assignable or delegable by Executive. Any purported assignment or delegation by Executive in violation of the foregoing shall be null and void ab initio and of no force and effect. This Agreement may be assigned by the Company to a person or entity which is an affiliate or a successor in interest to substantially all of the business operations of the Company. Upon such assignment, the rights and obligations of the Company hereunder shall become the rights and obligations of such affiliate or successor person or entity.

I. <u>Successor Agreement</u>. At, or simultaneously with, a Change of Control (as described in this Agreement), the Company will require any successor to all or substantially all of the business and/or assets of the Company (whether direct or indirect, and whether by purchase, merger, consolidation or otherwise) to expressly assume and agree, in writing, to perform this Agreement in the same manner and to the same extent that the Company would be required to perform it if no succession had taken place. Failure of the successor to so assume this Agreement shall constitute "Good Reason as defined in Section I.F of this Agreement.

J. <u>Notices</u>. For the purpose of this Agreement, notices and all other communications provided for in the Agreement shall be in writing and shall be deemed to have been duly given on the earlier of (i) the date that such notice is delivered by hand or overnight courier or (ii) three (3) days after it has been mailed by United States registered mail, return receipt requested, postage prepaid, addressed to the respective addresses set forth below in this Agreement, or to such other address as either party may have furnished to the other in writing in accordance herewith, except that notice of change of address shall be effective only upon receipt.

IF TO THE COMPANY:

Powell Industries, Inc. Attention: Chief Executive Officer 8550 Mosley Road Houston, Texas 77075

IF TO EXECUTIVE:

Milburn E. Honeycutt 20 Stonecliff Circle Sugar Land, TX 77479

K. <u>Prior Employment</u>. The Company has employed Executive for Executive's general skills, management abilities and experience in the Company's Business (as defined in the Confidentiality Agreement referred to herein). Executive acknowledges that Executive has been specifically instructed not to bring, disclose or use in any fashion any confidential information, trade secrets, proprietary information, data or technology, nor any confidential pricing information, belonging to any prior employer. In no event is Executive authorized to use or disclose any such information to the Company or any of its employees.

L. <u>Executive's Representations</u>. Executive hereby represents to the Company that (i) all confidential information, trade secrets or proprietary information, data or technology, belonging to any prior employer, including those that might have been contained on Executive's personal computer, cell phone or other electronic communications or storage device have been returned and/or deleted in accordance with any policy of or agreement with Executive's prior employer and (ii) the execution and delivery of this Agreement by Executive and the Company and the performance by Executive of Executive's duties hereunder shall not constitute a breach of, or otherwise contravene, the terms of any employment agreement or other agreement or policy to which Executive is a party or otherwise bound.

M. <u>Reimbursement of Legal Expenses</u>. The Company shall reimburse Executive for reasonable and customary fees charged by Executive's attorney to provide legal counsel review and defense concerning this Agreement, not exceed \$10,000.

N. <u>Cooperation</u>. Executive shall provide Executive's reasonable cooperation in connection with any action or proceeding (or any appeal from any action or proceeding) which relates to events occurring during Executive's employment hereunder. Executive shall be entitled to reimbursement for reasonable and customary expenses incurred for purposes of cooperating in any action or proceeding pursuant to this Section. This provision shall survive any termination of this Agreement.

O. <u>Indemnification</u>. Executive shall be indemnified by the Company against liability as an officer and director of the Company and any subsidiary or affiliate of the Company to the maximum extent permitted by the Company's bylaws by applicable law or by any indemnity agreement heretofore or hereafter executed between the Company and Executive. Executive's rights under this Section shall continue so long as Executive maybe subject to such liability, whether or not this Agreement may have terminated prior thereto. The Company will insure Executive, for the duration of his employment with the Company and thereafter with respect to his acts and omissions occurring during such employment, under a contract of director and officer liability insurance to the same extent as such insurance insures members of the Board.

P. <u>Withholding of Taxes</u>. The Company may withhold from any amounts or benefits payable under this Agreement all taxes it may be required to withhold pursuant to any applicable law or regulation.

Q. <u>Counterparts</u>. This Agreement may be signed in counterparts, each of which shall be an original, with the same effect as if the signatures thereto and hereto were upon the same instrument.

R. <u>Survival</u>. The provisions of this Agreement, together with the provisions of the Confidentiality, Non-Competition and Non-Solicitation Agreement, attached and part of this Agreement as Schedule A, shall each survive the termination of Executive's employment, regardless of how such termination is caused.

S. <u>Executive Severance Protection Plan</u>. The rights and benefits provided in this agreement are intended to replace the rights and benefits heretofore provided by the Powell Industries, Inc. Executive Severance Protection Plan adopted and effective September 20, 2002 (the Prior Plan). By execution of this agreement, Executive waives and releases any rights, benefits or payments to which Executive would have been entitled to under the Prior Plan and acknowledges that the Prior Plan has been or will be terminated by the Company.

IN WITNESS WHEREOF, THE PARTIES HERETO HAVE DULY EXECUTED THIS AGREEMENT EFFECTIVE FOR ALL PURPOSES AS OF THE EFFECTIVE DATE.

BY EXECUTIVE:

/s/ Milburn E. Honeycutt Milburn E. Honeycutt

Date: May 8, 2012

BY POWELL INDUSTRIES, INC.:

/s/ Thomas W. Powell

Thomas W. Powell President & Chief Executive Officer Chairman of the Board

Date: May 8, 2012



This Confidentiality, Non-Competition and Non-Solicitation Agreement ("this Agreement") is entered into between Powell Industries, Inc., on behalf of itself, and any and all of its subsidiaries, affiliates or successors (all of whom are hereinafter collectively referred to as "Company") and Milburn E. Honeycutt ("Executive") in connection with and ancillary to an Executive Employment Agreement (the "Employment Agreement") being entered into between the Company and Executive of even date herewith.

I. Nondisclosure of Confidential Information and Trade Secrets

A. <u>Company's Agreements</u>. During the course of Executive's employment by the Company, the Company agrees: (i) to provide Executive with specialized training and continuing training and development regarding its products, services, methods, systems and operations; (ii) to provide Executive with access to its Confidential Information and Trade Secrets (as defined herein); and (iii) to provide Executive with Confidential Information and Trade Secrets about, and the opportunity to develop close relationships with the Company's management personnel, employees, Customers (as defined herein), Suppliers (as defined herein) and the employees, agents and representatives of Customers and Suppliers.

B. <u>Company's Business</u>. Company is engaged in the highly competitive business of the design, manufacture and packaging of equipment and systems for the distribution, control, generation and management of electrical and other power sources (all of which is hereinafter collectively referred to as the "Company's Business"). Executive acknowledges that because of the highly competitive nature of the Company's Business, the use and protection of the Company's Confidential Information and Trade Secrets as defined in this Agreement is critical to the Company's continued successful operation and business and is an essential element of this Agreement.

C. <u>Definition of Confidential Information and Trade Secrets</u>. Confidential Information and Trade Secrets, as used in this Agreement, includes, but is not limited to, written, electronic, oral and visual information relating to:

1. Lists of, and all information about, each person or entity to which Company has sold, or made a proposal to sell any products, goods, services or equipment which comprise any part of the Company's Business (all of which are hereinafter collectively referred to as "Customers");

2. Lists of, and all information about, each person or entity from which the Company has acquired equipment, inventory, components, products or services used by the Company to design, manufacture, fabricate, sell or deliver any of the products or services which comprise the Company's Business (all of which are hereinafter collectively referred to as Supplier;

3. All Customer contact information, which includes information about the identity and location of individuals with decision-making authority and the particular preferences, needs or requirements of the Customer, or such individual, with respect to any of the products, goods, services or equipment which comprise any part of the Company's Business, and all information about the particular needs or requirements of Customers based on geographical, economic or other factors;

4. Financial information of any kind about Customers, including sales and purchase histories, trend information about the growth or shrinking of a particular Customer's needs, purchases or requirements; profit margins or markups, as well as all information about the costs and expenses which the Company incurs to provide products or services to its Customers;

5. The Company's procedures, forms, methods, and systems for marketing to Customers and potential customers including all of its Customer development techniques and procedures, including training and other internal manuals, forms and documents;

6. All Supplier contact information, which includes information about the identify and location of individuals with decision-making authority and the particular capabilities, capacities, expertise, prices and/or schedules of such Suppliers;

7. All of the Company's non-public business, expansion, marketing, development, financial or budgeting plans, strategies, forecasts or proposals;

8. All of the Company's pricing and hedging formulas, methodologies, practices and systems, including those based upon particular Customers, quantities, or geographic, seasonal, economic or other factors, including all information about the price, terms, quantities or conditions of any products or services sold or furnished by the Company to its Customers;

9. Technical information about the Company, including designs, drawings, engineering and information regarding the configuration, assembly or contents of any of the Company's products or any of its hardware, equipment, tools, machinery or other manufacturing, fabrication or assembly devices or processes, or those of any of its Customers, consultants, vendors, suppliers, or any person or entity which provides manufacturing or fabrication services to the Company;

10. Any non-public financial information of any kind about the Company or its operations;

11. Information disclosed to the Company by third parties, concerning the Company's products, goods or services, bids or bidding processes, product or manufacturing specifications (except to the extent such information is publicly disclosed), contracts, procedures, or business practices;

12. Employee lists, phone numbers and addresses, pay rates, benefits and compensation packages, training programs and manuals, and other confidential information regarding the Company's personnel.

D. <u>Confidential Information and Trade Secrets</u>. Company and Executive agree that Confidential Information and Trade Secrets includes current, updated and future data, information, reports, evaluations and analyses of Company, its financial performance and results, or its Executives, including their compensation, performance or evaluation, as well as correspondence, proposals, contracts and other communications with, or financial, sales or other information about the Company's Customers and Suppliers, and includes (i) those which are provided to Executive after the date hereof, (ii) those which Executive creates, in whole or in part; (iii) those to which or for which Executive provides input or information; and (iv) those which Executive uses for the purpose of performing Executive's duties for the Company or making decisions relating to the Company's Business, its Customers, Suppliers or employees. Anything to the contrary not withstanding, however, Confidential Information and Trade Secrets shall not include (i) general industry knowledge acquired by Executive as a result of Executive's prior employment, (ii) non-confidential information acquired by Executive from any prior employment, (iii) contact information about Customers, Suppliers and others with whom Executive dealt prior to Executive's employment with the Company; and (iv) any other information generally available to the public.

E. <u>Protection of Confidential Information and Trade Secrets</u>. During the term of Executive's employment and at all times thereafter, Executive will keep all Confidential Information and Trade Secrets in strict confidence and will not use or disclose any Confidential Information and Trade Secrets for any purpose other than the performance of Executive's duties for Company. Executive will not use any Confidential Information and Trade Secrets for the gain or benefit of any person or entity other than the Company or for Executive's own personal gain or benefit. Executive will not cause the transmission, removal or transport of Confidential Information and Trade Secrets from the Company's premises except in accordance with the Company's approved procedures and then only to the extent necessary to perform Executive's duties, while employed by the Company. Executive will not provide any information about the Company's Executives to any competitor or recruiter.

II. Intellectual Property and Work Product.

A. If Executive creates, invents, designs, develops, contributes to or improves any Works, either alone or with third parties, at any time during Executive's employment by the Company and within the scope of such employment and/or with the use of any the Company resources ("Company Works"), Executive shall promptly and fully disclose same to the Company and hereby irrevocably assigns, transfers and conveys, to the maximum extent permitted by applicable law, all rights and intellectual property rights therein (including rights under patent, industrial property, copyright, trademark, trade secret, unfair competition and related laws) to the Company to the extent ownership of any such rights does not vest originally in the Company.

B. Executive agrees to keep and maintain adequate and current written records (in the form of notes, sketches, drawings, and any other form or media requested by the Company) of all Company Works. The records will be available to and remain the sole property and intellectual property of the Company at all times.

C. During the Employment Term, Executive shall take all requested actions and execute all requested documents (including any licenses or assignments required by a government contract) at the Company's expense (but without further remuneration) to assist the Company in validating, maintaining, protecting, enforcing, perfecting, recording, patenting or registering any of the Company's rights in the Prior Works and Company Works. If the Company is unable for any other reason to secure Executive's signature on any document for this purpose, then Executive hereby irrevocably designates and appoints the Company and its duly authorized officers and agents as Executive's agent and attorney in fact, to act for and in Executive's behalf and stead to execute any documents and to do all other lawfully permitted acts in connection with the foregoing.

D. Executive shall not improperly use for the benefit of, bring to any premises of, divulge, disclose, communicate, reveal, transfer or provide access to, or share with the Company any confidential, proprietary or non-public information or intellectual property relating to a former employer or other third party without the prior written permission of such third party. Executive hereby indemnifies, holds harmless and agrees to defend the Company and its officers, directors, partners, employees, agents and representatives from any breach of the foregoing covenant. Executive shall comply with all relevant policies and guidelines of the Company, including regarding the protection of confidential information and intellectual property and potential conflicts of interest. Executive acknowledges that the Company may amend any such policies and guidelines from time to time, and that Executive remains at all times bound by their most current version.

III. Non-Competition and Non-Solicitation of Customers

A. Non-Competition.

1. So long as Executive is employed by the Company or one of its affiliates, and for the greater of (i) one year from the date of the termination of Executive's employment or (ii) the "Severance Period" as defined in Section II-I of the Executive's Employment Agreement (collectively the "Restricted Period"), Executive will not, whether on Executive's own behalf or on behalf of or in conjunction with any person, firm, partnership, joint venture, association, corporation or other business organization, entity or enterprise whatsoever ("Person"), directly or indirectly:

- (a) call upon, communicate with, solicit or assist in soliciting any Customer or Supplier, or any agent or employer of either, using any Confidential Information and Trade Secrets in any way;
- (b) participate in, work on or otherwise be involved in or with any project, contract, proposal, work, sale, bid or other undertaking (collectively "Project"), if Executive worked on, participated in, was involved, or communicated with other employees of the Company, Customers, Suppliers or other third parties, with regard to any such Project during the six (6) months prior to the date of the termination of Executive's employment.

B. Non-Solicitation.

1. During the Restricted Period, Executive will not, whether on Executive's own behalf or on behalf of or in conjunction with any Person, directly or indirectly:

- (a) solicit or encourage any employee of the Company or its affiliates to leave the employment of the Company or its affiliates; or
- (b) hire any employee who was employed by the Company or its affiliates as of the date of Executive's termination of employment with the Company or who left the employment of the Company or its affiliates coincident with, or within one year prior to or after, the termination of Executive's employment with the Company.

2. During the Restricted Period, Executive will not, directly or indirectly, solicit or encourage to cease to work with the Company or its affiliates any consultant then under contract with the Company or its affiliates.

3. It is expressly understood and agreed that although Executive and the Company consider the restrictions contained in this Section 3 to be reasonable, if a final judicial determination is made by a court of competent jurisdiction that the time or territory or any other restriction contained in this Agreement is an unenforceable restriction against Executive, the provisions of this Agreement shall not be rendered void but shall be deemed amended to apply as to such maximum time and territory and to such maximum extent as such court may judicially determine or indicate to be enforceable. Alternatively, if any court of competent jurisdiction finds that any restriction contained in this Agreement is unenforceable, and such restriction cannot be amended so as to make it enforceable, such finding shall not affect the enforceability of any of the other restrictions contained herein.

IV. Company Property

Executive also agrees that all (i) correspondence, proposals, notes, reports, memoranda, records and files; (ii) plans, specifications, drawings, blueprints, and designs; (iii), training, service or other manuals; (iv) Customer or personnel lists or files, including mailing or contact lists; (v) computer software, programs, disks or files; (vi) tools, materials or equipment; (vii) photographs, photostats, negatives, undeveloped film; (viii) tape or electronic recordings (ix) information contained on any electronic storage or communications device used by Executive during Executive's employment with the Company, including those furnished by the Company and those owned by Executive, and (x) any other documents or programs, whether compiled by Executive or other Executives of the Company, or its contractors, vendors or consultants, and those which were made available to Executive while employed at the Company, which contain any Confidential Information and Trade Secrets or concern or describe any part of the Company's Business, Executive's employment or the Company's or Executive's dealings, transactions or communications with any Customers (all of which is hereinafter collectively referred to as Company Information), are and shall remain the sole and exclusive property of the Company. Executive agrees that this includes any Company Information contained on or within any personal computer, blackberry, cell phone, iPad, or any other telephonic or electronic communication or data storage device, including those owned by Executive which were used during Executive's employment with the Company (all of which are hereinafter collectively called Electronic Devices). At any time upon the Company any files, records, notes or other documents which were used during Executive's employment with the Company or which contain any Company Information. Executive will not keep in Executive's possession nor disclose nor deliver to anyone else any Company Information whether in electronic, paper or any other format.

V. Rights and Remedies Upon Breach

A. If Executive breaches any of the provisions of this Agreement, the Company will have all of the following rights and remedies, each of which shall be independent of the other and severally enforceable, and all of which shall be in addition to, and not in lieu of, any other rights and remedies available to the Company under law or in equity: (i) to have this Agreement specifically enforced by any court of competent jurisdiction; (ii) to seek and

obtain injunctive or other equitable relief of any kind, Executive hereby acknowledging and agreeing that any such breach or threatened breach will cause irreparable injury to the Company and that monetary damages will not provide an adequate remedy to the Company; (iii) to require Executive to account for and pay over to the Company all compensation, profits, monies, or other benefits derived or received by Executive as a result of any act or transaction constituting a breach of this Agreement.

B. Executive agrees and stipulates that in any action or claim brought by Executive or in any action or claim brought against Executive involving the provisions of this Agreement, Executive hereby expressly waives any claim or defense that the non-competition, non-solicitation and non-disclosure covenants contained in this Agreement are unenforceable, void or voidable, for any reason, including, but not limited to, fraud, misrepresentation, illegality, failure of consideration, illusory contract, mistake, or any other legal defense as to the validity or enforceability of this Agreement.

C. In addition to any and all other remedies available to Company, in the event of a breach of or default under this Agreement, or in the event that the Company obtains a judgment, which becomes final after the expiration of time for all appeals, that Executive has violated any of the provisions of Section II or Section III of this Attachment A, the Company shall be entitled to recover, and the Executive (or his estate) shall be obligated to repay and return to the Company, upon written demand therefore, an amount equal to all severance or other benefits paid to, or on behalf of, the Executive (or his estate) pursuant to the provisions of the Employment Agreement (other than the Accrued Rights) on or after the date of termination of Executive's employment.

VI. General Provisions

A. <u>Employment Agreement</u>. The Employment Agreement is an integral part of this Agreement, and this Agreement is an integral part of the Employment Agreement. A breach of or default under this Agreement shall constitute a material breach of the Employment Agreement; provided, however, that none of the notice requirements of the Employment Agreement shall be applicable to any actual breach of this Agreement.

B. <u>Other Agreements</u>. To the extent that Executive has heretofore entered into an agreement with the Company containing confidentiality, non disclosure, non competition and/or non-solicitation provisions, this Agreement shall constitute an amendment, modification and continuation of all such agreements and obligations, which shall be deemed to be modified as provided herein. No modification of or amendment to this Agreement, nor any waiver of rights under this Agreement, shall be effective unless it is in writing and signed by both Executive and the Company. Any subsequent change or changes in Executive's duties, salary or compensation will not affect the validity or scope of this Agreement.

C. <u>Agreement Ancillary to Other Agreements</u>. This Agreement is ancillary to and part of other agreements between the Company and the Executive, including the Employment Agreement and the Company's agreements to: (i) disclose, and to continue to disclose its Confidential Information and Trade Secrets to Executive; (ii) provide initial and continued training, education and development to Executive; (iii) provide Executive with Confidential Information and Trade Secrets about, and the opportunity to develop close relationships with the Company's management personnel, employees, Customers, Suppliers and the employee's agents and representative of Customers and Suppliers.

D. <u>Severability</u>. If one or more of the provisions in this Agreement are held to be void or unenforceable in whole or in part, the remaining provisions will continue in full force and effect. Executive further agrees that in the event the length of time, the geographic area or definition of business activity as set forth herein, is deemed unreasonable, or otherwise unenforceable, in any court proceedings, the Executive and the Court may reform, modify or reduce such restrictions such that they are reasonable and enforceable.

E. <u>"At Will" and Termination</u>. This Agreement does not alter in any way the at-will nature of employment between Executive and the Company, which may be terminated by the Company or by Executive in accordance with the terms of Executive Employment Agreement of even date herewith.

F. <u>Choice of Law</u>. This Agreement shall be construed and interpreted in accordance with the laws of the State of Texas. All obligations payable or performable hereunder shall be payable and performable at the Company's offices in Houston, Harris County, Texas.

G. <u>Enforceability</u>. This Agreement shall be enforceable by the Company, and any of its successors, assigns, affiliates, subsidiaries, parent or related corporations or entities, including any person or entity to which the Company sells, transfers or assigns all or any part of its assets, or any entity to, in or with which the Company may hereafter enter into a merger transaction of any kind. Executive shall have no right to transfer or assign Executive's rights or obligations hereunder.

H. <u>Survival</u>. The provisions of this Agreement shall survive the termination of Executive's employment by the Company, regardless of how such employment is terminated.

BY SIGNING BELOW, EXECUTIVE REPRESENTS THAT EMPLOYEE HAS READ THIS AGREEMENT CAREFULLY AND UNDERSTANDS AND AGREES TO ITS TERMS, INCLUDING THOSE LIMITING EXECUTIVE'S RIGHTS TO SOLICIT CUSTOMERS OR EXECUTIVES OF THE COMPANY.

BY EXECUTIVE:

May 8, 2012 Date

/s/ Milburn E. Honeycutt Milburn E. Honeycutt

Vice President & Chief Accounting Officer

BY POWELL INDUSTRIES, INC.:

May 8, 2012 Date

/s/ Thomas W. Powell Thomas W. Powell President & Chief Executive Officer Chairman of the Board

TENTH AMENDMENT TO CREDIT AGREEMENT

THIS TENTH AMENDMENT TO CREDIT AGREEMENT (this "*Amendment*") is entered into as of March 26, 2012, among Powell Industries, Inc., a Delaware corporation ("*Parent*" or "*Borrower*"), Bank of America, N.A., a national banking association, as Agent, Swing Line Lender and L/C Issuer under the Credit Agreement (in such capacity as administrative agent, together with its successors in such capacity, "*Agent*"), and each lender from time to time party to the Credit Agreement (collectively, "*Lenders*" and individually, a "*Lender*"). Capitalized terms used but not defined in this Amendment have the meaning given them in the Credit Agreement (defined below).

RECITALS

A. Parent, as borrower, Agent and Lenders are parties to that certain Credit Agreement dated as of June 29, 2005 (as amended by the First Amendment to Credit Agreement dated November 7, 2005, the Second Amendment to Credit Agreement dated January 11, 2006, the Third Amendment to Credit Agreement dated December 7, 2006, the Fifth Amendment to Credit Agreement dated December 4, 2007, the Sixth Amendment to Credit Agreement dated December 14, 2007, the Seventh Amendment to Credit Agreement dated December 10, 2008, the Eighth Amendment to Credit Agreement dated March 11, 2010, the Ninth Amendment to Credit Agreement dated May 11, 2011, and as further amended, restated or supplemented, the "*Credit Agreement*").

B. Parent has requested, and Agent and Lenders have agreed, subject to the terms and conditions herein, to amend certain covenants in the Credit Agreement.

C. NOW THEREFORE, for good and valuable consideration, the receipt and sufficiency of which are acknowledged, the undersigned hereby agree as follows:

1. <u>Amendments to Credit Agreement</u>. The Credit Agreement is amended as set forth below as of the Effective Date:

(a) The definitions of "*Applicable Rate*" and "*Consolidated Fixed Charge Coverage Ratio*" in *Section 1.01* of the Credit Agreement are deleted in their entireties and are replaced with the following:

"*Applicable Rate* means, except as specified below, the following percentages per annum, based upon the Consolidated Leverage Ratio as set out in the most recent Compliance Certificate received by Agent pursuant to *Section 6.02(b)*:

Rudo	Rate	Rate	Credit Fee	Fee
<1.25:1.00	+1.00%	-0.50%	1.00%	0.1875%
1.25:1.00 but <2.25:1.00	+1.25%	-0.25%	1.00%	0.1875%
>2.25:1.00	+1.75%	+0.00%	1.75%	0.1875%
	1.25:1.00 but <2.25:1.00	<1.25:1.00 +1.00% 1.25:1.00 but <2.25:1.00 +1.25%	<1.25:1.00 +1.00% -0.50% 21.25:1.00 but <2.25:1.00	<1.25:1.00+1.00%-0.50%1.00%21.25:1.00 but <2.25:1.00

Any increase or decrease in the Applicable Rate resulting from a change in the Consolidated Leverage Ratio shall become effective as of the first Business Day immediately following the date a Compliance Certificate is delivered pursuant to *Section 6.02(b)*; *provided that*, if a Compliance Certificate is not delivered when due in accordance with such Section, then Pricing Level 3 shall apply as of the first Business Day after the date on which such Compliance Certificate was required to have been delivered until the first Business Day after a Compliance Certificate establishing a lower applicable Pricing Level is delivered pursuant to *Section 6.02(b)*.

Consolidated Fixed Charge Coverage Ratio means, as of any date of determination, the ratio of (a) the difference of (i) Consolidated EBITDA for the period of the four fiscal quarters most recently ended for which Borrower has delivered Current Financial Statements pursuant to **Section 6.01(a) or (b)** minus (ii) cash federal, state, local, and foreign income taxes for Parent and its Subsidiaries for such period, minus actual cash Maintenance Capital Expenditures paid during such period, to (b) Debt Service.

(b) Section 1.01 of the Credit Agreement is amended to add the following new definition in its appropriate alphabetical order:

Maintenance Capital Expenditures means Capital Expenditures made by Parent and its Subsidiaries that constitute normal replacements and maintenance which are properly charged to current operations.

(c) *Section 7.11(f)* (Capital Expenditures) of the Credit Agreement is deleted in its entirety and replaced with the following:

"(f) **Capital Expenditures**. Make or become legally obligated to make any expenditure in respect of the purchase or other acquisition of any fixed or capital asset (excluding normal replacements and maintenance which are properly charged to current operations), except for Capital Expenditures that (i) do not exceed \$60,000,000 in the aggregate, during the two year period commencing October 1, 2011, and ending on September 30, 2013, and (ii) do not exceed \$20,000,000 per Fiscal Year, commencing with the Fiscal Year ending September 30, 2014, and for each Fiscal Year ending thereafter.

2. <u>Delta-Unibus Corp. Letter of Credit</u>. Borrower, Agent and Lenders agree that Irrevocable Letter of Credit No. 3041587 (as amended), issued by Bank of America, N.A., to Bank of New York as trustee, as beneficiary, for Delta-Unibus Corp. (now known as Powell Electrical Systems, Inc.) as letter of credit applicant, shall be deemed a Letter of Credit issued under *Section 2.03* of the Credit Agreement, and shall be subject to and governed by the terms and conditions of the Credit Agreement. Borrower and Powell Electrical Systems, Inc. each agree that the Letter of Credit and Reimbursement Agreement dated October 25, 2001 (as amended prior to the date hereof), is terminated and is of no further force and effect.

3. <u>Conditions</u>. This Amendment shall be effective as of the date first set forth above (the "*Effective Date*") once each of the following have been delivered to Agent:

- (a) this Amendment executed by Borrower, Agent and Lenders;
- (b) Guarantors' Consent and Agreement executed by the Guarantors; and
- (c) such other documents as Agent or Lenders may reasonably request.

4. <u>Representations and Warranties</u>. Parent represents and warrants to Agent and Lenders that (a) it possesses all requisite power and authority to execute, deliver and comply with the terms of this Amendment, (b) this Amendment has been duly authorized and approved by all requisite corporate action on the part of such parties, (c) no other consent of any Person (other than Lenders) is required for this Amendment to be effective, (d) the execution and delivery of this Amendment does not violate its organizational documents, (e) the representations and warranties in each Loan Document to which it is a party are true and correct in all material respects on and as of the date of this Amendment as though made on the date of this Amendment (*except* to the extent that such representations and warranties speak to a specific date), (f) it is in full compliance with all covenants and agreements contained in each Loan Document (as amended by this Amendment) to which it is a party, and (g) no Default or Event of Default has occurred and is continuing. The representations and warranties made in this Amendment shall survive the execution and delivery of this Amendment. No investigation by Agent or Lenders is required for Agent or Lenders to rely on the representations and warranties in this Amendment.

5. Scope of Amendment; Reaffirmation; RELEASE. All references to the Credit Agreement shall refer to the Credit Agreement as amended by this Amendment. Except as affected by this Amendment, the Loan Documents are unchanged and continue in full force and effect. However, in the event of any inconsistency between the terms of the Credit Agreement (as amended by this Amendment) and any other Loan Document, the terms of the Credit Agreement shall control and such other document shall be deemed to be amended to conform to the terms of the Credit Agreement. Parent hereby reaffirms its obligations under the Loan Documents to which each is a party and agrees that all Loan Documents to which it is a party remain in full force and effect and continue to be legal, valid, and binding obligations enforceable in accordance with their terms (as the same are affected by this Amendment). As A MATERIAL PART OF THE CONSIDERATION FOR AGENT AND LENDERS ENTERING INTO THIS AMENDMENT, PARENT HEREBY RELEASES AND FOREVER DISCHARGES AGENT AND LENDERS (AND THEIR RESPECTIVE SUCCESSORS, ASSIGNS, AFFILIATES, OFFICERS, MANAGERS, DIRECTORS, EMPLOYEES, AND AGENTS) FROM ANY AND ALL CLAIMS, DEMANDS, DAMAGES, CAUSES OF ACTION, OR LIABILITIES FOR ACTIONS OR OMISSIONS (WHETHER ARISING AT LAW OR IN EQUITY, AND WHETHER DIRECT OR INDIRECT) IN CONNECTION WITH THE CREDIT AGREEMENT AND THE OTHER LOAN DOCUMENTS PRIOR TO THE DATE OF THIS AMENDMENT, WHETHER OR NOT HERETOFORE ASSERTED, AND WHICH BORROWER OR ANY COMPANY MAY HAVE OR CLAIM TO HAVE AGAINST AGENT AND LENDERS.

6. Miscellaneous.

(a) <u>Waiver</u>. This Amendment does not constitute (i) a waiver of, or a consent to (A) any provision of the Credit Agreement or any other Loan Document not expressly referred to in this Amendment, or (B) any present or future violation of, or default under, any provision of the Loan Documents, or (ii) a waiver of Agent's or Lenders' right to insist upon future compliance with each term, covenant, condition and provision of the Loan Documents.

(b) <u>Form</u>. Each agreement, document, instrument or other writing to be furnished to Lenders under any provision of this Amendment must be in form and substance satisfactory to Agent and its counsel.

(c) <u>Headings</u>. The headings and captions used in this Amendment are for convenience only and will not be deemed to limit, amplify or modify the terms of this Amendment, the Credit Agreement, or the other Loan Documents.

(d) <u>Costs, Expenses and Attorneys' Fees</u>. Parent agrees to pay or reimburse Agent on demand for all of their reasonable out-of-pocket costs and expenses incurred in connection with the preparation, negotiation, and execution of this Amendment, including, without limitation, the reasonable fees and disbursements of Agent's counsel.

(e) <u>Successors and Assigns</u>. This Amendment shall be binding upon and inure to the benefit of each of the undersigned and their respective successors and permitted assigns.

(f) <u>Multiple Counterparts</u>. This Amendment may be executed in any number of counterparts with the same effect as if all signatories had signed the same document. All counterparts must be construed together to constitute one and the same instrument. This Amendment may be transmitted and signed by facsimile. The effectiveness of any such documents and signatures shall, subject to applicable law, have the same force and effect as manually-signed originals and shall be binding on Parent, Agent and Lenders. Agent may also require that any such documents and signatures be confirmed by a manually-signed original; *provided that* the failure to request or deliver the same shall not limit the effectiveness of any facsimile document or signature.

(g) Governing Law. This Amendment and the other Loan Documents must be construed, and their performance enforced, under Texas law.

(h) <u>Entirety</u>. The LOAN DOCUMENTS (AS AMENDED HEREBY) REPRESENT THE FINAL AGREEMENT AMONG PARENT, AGENT AND LENDERS AND MAY NOT BE CONTRADICTED BY EVIDENCE OF PRIOR, CONTEMPORANEOUS, OR SUBSEQUENT ORAL AGREEMENTS BY THE PARTIES. THERE ARE NO UNWRITTEN ORAL AGREEMENTS AMONG THE PARTIES.

[Signatures appear on the following pages.]

The Amendment is executed as of the date set out in the preamble to this Amendment.

POWELL INDUSTRIES, INC., as Borrower

By: /s/ Don R. Madison

Don R. Madison Vice President, Secretary and Treasurer

Executed by Powell Electrical Systems, Inc. for purposes of *Section 2* of this Amendment.

POWELL ELECTRICAL SYSTEMS, INC.

By: /s/ Don R. Madison

Don R. Madison Vice President, Secretary, and Treasurer

Signature Page to Tenth Amendment to Credit Agreement

BANK OF AMERICA, N.A., as Agent

By: /s/ Ted Puckett

Ted Puckett Senior Vice President

BANK OF AMERICA, N.A., as a Lender, L/C Issuer and Swing Line Lender

By: /s/ Ted Puckett

Ted Puckett Senior Vice President

Signature Page to Tenth Amendment to Credit Agreement

GUARANTORS' CONSENT AND AGREEMENT TO TENTH AMENDMENT

As an inducement to Agent and Lenders to execute, and in consideration of Agent's and Lenders' execution of, this Amendment, the undersigned hereby consents to this Amendment and agrees that this Amendment shall in no way release, diminish, impair, reduce or otherwise adversely affect the obligations and liabilities of the undersigned under the Guaranty executed by each of the undersigned in connection with the Credit Agreement, or under any Loan Documents, agreements, documents or instruments executed by the undersigned to create liens, security interests or charges to secure any of the Obligations (as defined in the Credit Agreement), all of which are in full force and effect. The undersigned further represents and warrants to Agent and Lenders that (a) the representations and warranties in each Loan Document to which it is a party are true and correct in all material respects on and as of the date of this Amendment as though made on the date of this Amendment (except to the extent that such representations and warranties speak to a specific date), (b) it is in full compliance with all covenants and agreements contained in each Loan Document to which it is a party, and (c) no Default or Event of Default has occurred and is continuing. Guarantors hereby release Agent and Lenders from any liability for actions or omissions in connection with the Loan Documents prior to the date of this Amendment. This Guarantors' Consent and Agreement shall be binding upon each of the undersigned, and its permitted assigns, and shall inure to the benefit of Agent, Lenders, and its successors and assigns.

GUARANTORS:

TRANSDYN, INC., a Delaware corporation

By: /s/ Don R. Madison Don R. Madison Vice President, Secretary, and Treasurer

POWELL INDUSTRIES INTERNATIONAL, INC., a Delaware corporation

By: /s/ Don R. Madison

Don R. Madison Vice President, Secretary, and Treasurer POWELL INDUSTRIES ASIA, INC., a Delaware corporation

By: /s/ Don R. Madison Don R. Madison Vice President, Secretary, and Treasurer

POWELL ELECTRICAL SYSTEMS, INC., a Delaware corporation

By: /s/ Don R. Madison

Don R. Madison Vice President, Secretary, and Treasurer

Guarantors' Consent and Agreement to Tenth Amendment to Credit Agreement

AMENDED AND RESTATED CREDIT AGREEMENT

BETWEEN

POWELL CANADA INC. as Borrower

AND

POWELL INDUSTRIES, INC., NEXTRON LIMITED, and PPC TECHNICAL SERVICES INC. as Guarantors

AND

HSBC BANK CANADA, as Lender

MADE AS OF APRIL 26, 2012

ARTICLE 1—INTERPRETATION	2
1.1 Definitions	2
1.2 Headings; Articles and Sections	21
1.3 Number; persons; including	21
1.4 Application of Accounting Principles	22
1.5 References to Agreements and Enactments	22
1.7 Per Annum Calculations	22
1.8 References to Borrower	22
1.9 Schedules	22
ARTICLE 2 THE CREDIT FACILITIES	23
2.1 The Credit Facilities	23
2.2 Types of Availments	23
2.3 Purpose	23
2.4 Availability and Nature of the Credit Facilities	24
2.5 Margin Requirements	24
2.6 Notice Periods for Drawdowns, Conversions and Rollovers	24
2.7 Currency Determination	25
2.8 Conversion Option	25
2.9 Rollovers and Conversions not Repayments	25
2.10 Irrevocability	25
2.11 Optional Cancellation or Reduction of the Revolving Facility	25
2.12 Optional Repayment of Credit Facilities	26
2.13 Mandatory Repayment of Credit Facilities	26
2.14 Cash Collateral	26
2.15 Commitment and Renewal Fees	27
ARTICLE 3 CONDITIONS PRECEDENT TO DRAWDOWN	27
3.1 Conditions for All Drawdowns	27
3.2 Conditions for Closing	28
3.3 Waiver	29
ARTICLE 4 EVIDENCE OF DRAWDOWNS	29
4.1 Account of Record	29
ARTICLE 5 PAYMENTS OF INTEREST AND FEES	30
5.1 Interest on Canadian Prime Rate Loans	30
5.2 Interest on U.S. Base Rate Loans	30
5.3 Standby Fees for Revolving Facility	30
5.4 Stamping Fees	31
5.5 Fees Relating to Letters of Credit	31
5.6 Interest Act (Canada)	31
5.7 Nominal Rates; No Deemed Reinvestment	31
5.8 Interest on Overdue Amounts	32
5.9 Waiver	32
5.10 Maximum Rate Permitted by Law	32
ARTICLE 6 BANKERS' ACCEPTANCES	32
6.1 Bankers Acceptances	32
6.2 Fees	32
6.3 Form and Execution of Bankers' Acceptances	32
6.4 Power of Attorney; Provision of Bankers' Acceptances to Lender	33
6.5 Mechanics of Issuance	36
6.6 Rollover, Conversion or Payment on Maturity	36

6.6 Rollover, Conversion or Payment on Maturity

6.7 Restriction on Rollovers and Conversions	37
6.8 Rollovers	37
6.9 Conversion into Bankers' Acceptances	37
6.10 Conversion from Bankers' Acceptances	37
ARTICLE 7 LETTERS OF CREDIT	37
7.1 Availability	37
7.2 Currency, Type and Expiry	37
7.3 Reimbursement or Conversion on Presentation	38
7.4 Uniform Customs and Practice	38
ARTICLE 8 PLACE AND APPLICATIONS OF PAYMENTS	38
8.1 Place of Payment of Principal, Interest and Fees; Payments to Lender	38
8.2 Absolute and Unconditional Obligation to Pay	38
8.3 Funds	38
8.4 Application of Payments	39
8.5 Payments Clear of Taxes	39
8.6 Set Off	40
8.7 INTENTIONALLY DELETED	41
ARTICLE 9 REPRESENTATIONS AND WARRANTIES	41
9.1 Representations and Warranties	41
9.2 Deemed Repetition	47
9.3 Other Documents	47
9.4 Effective Time of Repetition	47
9.5 Nature of Representations and Warranties	47
ARTICLE 10 GENERAL COVENANTS	47
10.1 Affirmative Covenants of the Borrower	47
10.2 Negative Covenants of the Borrower	53
10.3 Financial Covenants	56
10.4 Lender May Perform Covenants	56
ARTICLE 11 SECURITY	57
11.1 Security on all Assets	57
11.2 Registration	58
11.3 Forms	58
11.4 Continuing Security	58
11.5 Dealing with Security	58
11.6 Effectiveness	59
11.7 Release and Discharge of Security	59
ARTICLE 12 EVENTS OF DEFAULT AND ACCELERATION	59
12.1 Events of Default	59
12.2 Acceleration	63
12.3 Set Off; Cash Collateral Accounts	63
12.4 Remedies Cumulative and Waivers	64
12.5 Termination of Lender's Obligations	64
12.6 Acceleration of All Lender Obligations	64
12.7 Application of Payments Following Acceleration	65
12.8 Calculations as at the Adjustment Time	65
ARTICLE 13 CHANGE OF CIRCUMSTANCES	65
13.1 Market Disruption Respecting Bankers' Acceptances	65
13.2 Illegality	66
ARTICLE 14 COSTS, EXPENSES AND INDEMNIFICATION	66
14.1 Costs and Expenses	66
14.2 General Indemnity	67

- 2 -

14.3 Environmental Indemnity	
14.4 Judgment Currency	
ARTICLE 15 GENERAL	
15.1 Exchange and Confidentiality of Information	
15.2 Notices	
15.3 Governing Law	
15.4 Benefit of the Agreement	
15.5 Assignments and Participations	
15.6 Severability	
15.7 Whole Agreement	
15.8 Amendments and Waivers	
15.9 Further Assurances	
15.10 Attornment	
15.11 Time of the Essence	
15.12 Credit Agreement Governs	
15.13 Counterparts	

- 3 -

68 68 69

OF THE SECOND PART,

OF THE FIRST PART,

HSBC BANK CANADA, a Canadian Chartered Bank (hereinafter referred to as a "Lender"),

WHEREAS the Borrower and Powell have requested the Lender to provide the Credit Facilities to the Borrower for the purposes set forth herein and the Lender has agreed to provide the Credit Facilities to the Borrower on the terms and conditions hereinafter set forth;

NOW THEREFORE THIS AGREEMENT WITNESSES that in consideration of the covenants and agreements herein contained and other good and valuable consideration, the receipt and sufficiency of which is hereby conclusively acknowledged by each of the parties hereto, the parties hereto covenant and agree as follows:

- 1 -

AMENDED AND RESTATED CREDIT AGREEMENT

THIS AGREEMENT is made as of April 26, 2012

BETWEEN:

POWELL CANADA INC., a Canada Corporation (hereinafter sometimes referred to as the "Borrower"),

POWELL INDUSTRIES, INC., a Delaware Corporation, (hereinafter sometimes referred to as "Powell" and sometimes referred to as the "Guarantor")

-and-

NEXTRON LIMITED, a Canada Corporation, and PPC TECHNICAL SERVICES INC., a Canada Corporation (hereinafter sometimes referred to as "Nextron" and "PPC Technical", respectively, and sometimes referred to as the "Subsidiaries")

-and-

-and-

OF THE THIRD PART,

OF THE FOURTH PART

ARTICLE 1—INTERPRETATION

1.1 Definitions

In this Agreement, unless something in the subject matter or context is inconsistent therewith:

"Acceleration Notice" means a written notice delivered by the Lender to the Borrower pursuant to Section 12.2 declaring all Obligations of the Borrower outstanding hereunder to be due and payable.

"Adjustment Time" means the time of occurrence of the last event necessary (being either the delivery of a Demand for Payment or the occurrence of a Termination Event) to ensure that all Obligations and the Financial Instrument Obligations under any Lender Financial Instruments are thereafter due and payable.

"Advance" means an advance of funds made by the Lender to the Borrower, but does not include any Conversion or Rollover.

"Affiliate" means any person which, directly or indirectly, controls, is controlled by or is under common control with another person; and, for the purposes of this definition, "control" (including, with correlative meanings, the terms "controlled by" or "under common control with") means the power to direct or cause the direction of the management and policies of any person, whether through the ownership of shares or other economic interests, the holding of voting rights or contractual rights or otherwise.

"Agreement" means this credit agreement, as the same may be amended, modified, supplemented or restated from time to time in accordance with the provisions hereof.

"Applicable Laws" means, in relation to any person, transaction or event:

- (a) all applicable provisions of laws, statutes, rules and regulations from time to time in effect of any Governmental Authority; and
- (b) all Governmental Authorizations to which the person is a party or by which it or its property is bound or having application to the transaction or event.

"Applicable Pricing Rate", with respect to any Loan, means, the percentage rate per annum in the column applicable to the type of Loan in question:

Canadian				
Prime Rate				
Loans –	U.S. Base	Bankers'		
Canadian	Rate Loans –	Acceptance Based		
Prime Rate	U.S. Base	Loans Stamping	Letters of	
Plus	Rate Plus	Fee	Credit	Standby Fee
50BPS	50BPS	175BPS	150BPS	18.75BPS

^{- 2 -}

"Approved Securities" means obligations maturing within one year from their date of purchase or other acquisition by the Borrower or Subsidiary and which are:

- (a) issued by the Government of Canada or an instrumentality or agency thereof and guaranteed fully as to principal, premium, if any, and interest by the Government of Canada; or
- (b) term deposits, guaranteed investment certificates, certificates of deposit, bankers' acceptances or bearer deposit notes, in each case, of any Canadian chartered bank or other Canadian financial institution which has a long term debt rating of at least A+ by S&P, A1 by Moody's, or A (high) by DBRS.

"Attributable Debt" means, in respect of any lease (whether characterized as an operating lease under GAAP or not) entered into by a person or a Subsidiary thereof as lessee, the present value (discounted at the rate of interest implicit in such transaction, determined in accordance with GAAP) of the lease payments of the lessee, including all rent and payments to be made by the lessee in connection with the return of the leased property, during the remaining term of the lease (including any period for which such lease has been extended or may, at the option of the lessor, be extended) but excluding for certainty, (a) amounts required to be paid on account of insurance, taxes, assessments, utility, operating and labour costs and similar charges and (b) amounts payable by the lessee in connection with the exercise of any end-of-term purchase option, early buyout option or any similar amounts payable at the election of the lessee.

"**BA Rate**" means, on any date which Bankers' Acceptances are to be issued pursuant hereto, the per annum rate of interest which is the rate established from time to time by the Lender as the annual yield rate applicable to Canadian Dollar bankers' acceptances having identical issue and comparable maturity dates as the Bankers' Acceptances proposed to be issued by the Borrower.

"Bankers' Acceptance" means a draft in Canadian Dollars drawn by the Borrower, accepted by the Lender and issued for value pursuant to this Agreement.

"Banking Day" means a day on which banks are open for business in Edmonton, Alberta, Toronto, Ontario and New York, New York, but does not in any event include a Saturday or a Sunday.

"BPS" means basis points, with one basis point equal to one one hundredth of one percent (0.01%).

"Canadian Dollars" and "Cdn.\$" mean the lawful money of Canada.

"**Canadian Prime Rate**" means, for any day, the variable annual rate of interest per annum established from time to time by the Lender as the reference rate of interest for the determination of interest rates that the Lender will charge to customers in Canada for Canadian Dollar prime rate loans and which was 3.00%pa on March 1, 2012.

"Canadian Prime Rate Loan" means an Advance in, or Conversion into, Canadian Dollars made by the Lender to the Borrower with respect to which the Borrower has specified or a provision hereof requires that interest is to be calculated by reference to the Canadian Prime Rate.

"clearing house" has the meaning set out in Section 6.4(a)(i).

- 3 -

"Closing Date" means March 1, 2012.

"Commitments" means:

- (a) the Revolving Facility Commitment,
- (b) the EFT Facility Commitment,
- (c) the MC Facility Commitment,
- (d) the F/X Facility Commitment, and
- (e) the DSL Facility Commitment, and

"Commitment" means any of the Commitments.

"**Compliance Certificate**" means a certificate of the Borrower signed on its behalf by the president, chief financial officer, vice president – financial or treasurer of the Borrower, substantially in the form annexed hereto as Schedule A, to be given to the Lender by the Borrower pursuant hereto.

"**Consolidated Tangible Net Worth**" means, with respect to any person, all Shareholders' Equity of such person and its Subsidiaries which would, in accordance with GAAP, be classified upon a consolidated balance sheet of such person as Shareholders' Equity of such person and its Subsidiaries, and whether or not so classified, shall include, without duplication, preferred capital, <u>plus</u> (i) Permitted Subordinated Debt, <u>less</u> (ii) intangible assets including, without limitation, good will, trademarks and loans to and investments in Affiliates.

"**Conversion**" means a conversion or deemed conversion of a Loan under the Credit Facilities into another type of Loan under the Credit Facilities pursuant to the provisions hereof, subject to Section 2.7 and to Article 6 with respect to Bankers' Acceptances.

"**Conversion Date**" means the date specified by the Borrower as being the date on which the Borrower has elected to convert, or this Agreement requires the conversion of, one type of Loan into another type of Loan and which shall be a Banking Day.

"Conversion Notice" means a notice substantially in the form annexed hereto as Schedule B to be given to the Lender by the Borrower pursuant hereto.

"Credit Facilities" means, collectively, the Revolving Facility, EFT Facility, MC Facility, F/X Facility and DSL Facility, and "Credit Facility" means anyone of them.

"DBNA" has the meaning set out in Section 6.4.

"DBRS" means Dominion Bond Rating Service Limited and any successors thereto.

"**Debt**" means, with respect to any person, all obligations, liabilities and indebtedness of such person and its Subsidiaries which would, in accordance with GAAP, be classified upon a consolidated balance sheet of such person as liabilities of such person and its Subsidiaries and, whether or not so classified, shall include (without duplication):

(a) indebtedness of such person and its Subsidiaries for borrowed money;

- 4 -

- (b) obligations of such person and its Subsidiaries arising pursuant or in relation to: (i) bankers' acceptances (including payment and reimbursement obligations in respect thereof), or (ii) letters of credit and letters of guarantee supporting obligations which would otherwise constitute Debt within the meaning of this definition or indemnities issued in connection therewith;
- (c) obligations of such person and its Subsidiaries with respect to drawings under all other letters of credit and letters of guarantee;
- (d) obligations of such person and its Subsidiaries under Guarantees, indemnities, assurances, legally binding comfort letters or other contingent obligations relating to the indebtedness or other obligations of any other person which would otherwise constitute Debt within the meaning of this definition and all Financial Assistance including endorsements of bills of exchange (other than for collection or deposit in the ordinary course of business);
- (e) (i) all indebtedness of such person and its Subsidiaries representing the deferred purchase price of any property to the extent that such indebtedness is or remains unpaid after the expiry of the customary time period for payment, provided however that such time period shall in no event exceed 90 days, and (ii) all obligations of such person and its Subsidiaries created or arising under any: (A) conditional sales agreement or other title retention agreement or (B) capital lease;
- (f) all Attributable Debt of such person and its Subsidiaries other than in respect of (i) leases of office space or (ii) operating leases, in each case entered into in the ordinary course of business (and, for certainty, no Sale-Leaseback shall be considered to be entered into in the ordinary course of business);
- (g) all other long-term obligations (including the current portion thereof) upon which interest charges are customarily paid prior to a default by such person; and
- (h) all indebtedness of other persons secured by a Security Interest on any asset of such person and its Subsidiaries, whether or not such indebtedness is assumed thereby; provided that the amount of such indebtedness shall be the lesser of (i) the fair market value of such asset at such date of determination, and (ii) the amount of such indebtedness shall only be Debt to the extent recorded as a liability in accordance with GAAP;

but shall exclude each of the following, determined (as required) in accordance with GAAP:

- (i) accounts payable to trade creditors and accrued liabilities incurred in the ordinary course of business;
- (j) deferred and future taxes;
- (k) dividends or other equity distributions payable;
- (l) accrued interest not yet due and payable;
- (m) liabilities in respect of deferred credits and liabilities;

- 5 -

- (n) for certainty, any Debt owing to the Borrower or a Subsidiary; and
- (o) such other similar liabilities as may be agreed by the Lender from time to time.

"Debt to EBITDA Ratio" means, as at any date of determination, the ratio of (a) Debt as at such date to (b) EBITDA for the 12 months ending at such Quarter End, provided that, unless otherwise expressly provided or the context otherwise requires, references to "Debt to EBITDA Ratio" shall be and shall be deemed to be references to the Debt and EBITDA of Powell, on a consolidated basis.

"**Default**" means any event or condition which, with the giving of notice, lapse of time or upon a declaration or determination being made (or any combination thereof), would constitute an Event of Default.

"Demand for Payment" means an Acceleration Notice or a Financial Instrument Demand for Payment.

"Disclosure Schedule" means the disclosure schedule attached hereto as Schedule E.

"Discount Proceeds" means the net cash proceeds to the Borrower from the sale of a Bankers' Acceptance pursuant hereto before deduction or payment of the fees to be paid to the Lender under Section 6.2.

"Distribution" means:

- (a) the declaration, payment or setting aside for payment of any distribution on or in respect of any Ownership Interest in the Borrower or any Subsidiary which is not a Wholly-Owned Subsidiary (including any return of capital), but excluding to Powell or any of its Subsidiaries;
- (b) the redemption, retraction, purchase, retirement or other acquisition, in whole or in part, of any Ownership Interest in the Borrower or any Subsidiary which is not a Wholly-Owned Subsidiary, but excluding by Powell or any of its Subsidiaries;
- (c) the making of any loan or advance or any other provision of credit or Financial Assistance by the Borrower or any Subsidiary to any Related Party other than to the Borrower or a Wholly-Owned Subsidiary or Powell or any of its Subsidiaries;
- (d) the payment of any principal, interest, fees or other amounts on or in respect of any loans, advances or other Debt owing at any time by the Borrower or any Subsidiary to any Related Party, other than to the Borrower or a Wholly-Owned Subsidiary or Powell or any of its Subsidiaries;
- (e) (i) the payment of any amount, (ii) the sale, transfer, lease or other disposition of any property or assets, or (iii) any granting or creation of any rights or interests, at any time, by the Borrower or any Subsidiary to or in favour of any Related Party, other than to or in favour of the Borrower or a Wholly-Owned Subsidiary or Powell or any of its Subsidiaries; or
- (f) the payment of any principal, interest fees or other amounts on or in respect of any loans, advances of other Debt owing at any time by the Borrower or any subsidiary to any third party if such payment would cause an Event of Default, and whether any of the foregoing is made, paid or satisfied in or for cash, property or any combination thereof.

- 6 -

"**Documents**" means this Agreement, the Security Documents and all certificates, notices, instruments and other documents delivered or to be delivered to the Lender in relation to the Credit Facilities pursuant hereto or thereto and, when used in relation to any person, the term "Documents" shall mean and refer to the Documents executed and delivered by such person.

"Drawdown" means:

- (a) an Advance of a Canadian Prime Rate Loan;
- (b) an Advance of a U.S. Base Rate Loan;
- (c) the issuance of a Letter of Credit; or
- (d) the issuance of Bankers' Acceptances;

other than as a result of Conversions or Rollovers.

"Drawdown Date" means the date on which a Drawdown is made by the Borrower pursuant to the provisions hereof and which shall be a Banking Day.

"Drawdown Notice" means a notice substantially in the form annexed hereto as Schedule C to be given to the Lender by the Borrower pursuant hereto.

"DSL Facility" means the foreign exchange daily settlement facility in the maximum principal amount of Cdn. \$2,000,000.00 (or the Equivalent Amount thereof in United States Dollars) to be made available to the Borrower by the Lender in accordance with the provisions hereof, subject to any reduction in accordance with the provisions hereof.

"DSL Facility Commitment" means the maximum principal amount the Lender has agreed to make available to the Borrower under the DSL Facility, being Cdn. \$2,000,000.00, subject to reductions or adjustments pursuant to the terms hereof.

"EBITDA" means, with respect to any fiscal period of Powell, the Net Income for such period plus (on a consolidated basis and without duplication):

- (a) Interest Expense for such period, to the extent deducted in determining any portion of the Net Income;
- (b) all amounts deducted in the calculation of any portion of the Net Income for such period in respect of the provision for cash taxes (in accordance with GAAP); and
- (c) all amounts deducted in the calculation of any portion of the Net Income for such period in respect of non-cash items, including depreciation, amortization, deferred/future taxes, and other non-cash items approved by the Lender.

"EFT Facility" means the electronic funds transfer facility in the maximum principal amount of Cdn. \$2,000,000.00 per day, to be made available to the Borrower by the Lender in accordance with the provisions hereof, subject to any reduction in accordance with the provisions hereof, and to be governed by the terms of the Lender's Electronic Funds Transfer (EFT) Application Form.

- 7 -

"**EFT Facility Commitment**" means the maximum principal amount the Lender has agreed to make available to the Borrower under the EFT Facility, being Cdn. \$2,000,000.00 per day, subject to reductions or adjustments pursuant to the terms hereof.

"Environmental Claims" means any and all administrative, regulatory or judicial actions, suits, demands, claims, liens, notices of non-compliance or violation, investigations, inspections, inquiries or proceedings relating in any way to any Environmental Laws or to any permit issued under any such Environmental Laws including, without limitation:

- (a) any claim by a Governmental Authority for enforcement, clean up, removal, response, remedial or other actions or damages pursuant to any Environmental Laws; and
- (b) any claim by a person seeking damages, contribution, indemnification, cost recovery, compensation or injunctive or other relief resulting from or relating to Hazardous Materials, including any Release thereof, or arising from alleged injury or threat of injury to human health or safety (arising from environmental matters) or the environment.

"Environmental Laws" means all Applicable Laws with respect to the environment or environmental or public health and safety matters contained in statutes, regulations, rules, ordinances, orders, judgments, approvals, notices, permits or policies, guidelines or directives having the force of law.

"**Equity Interests**" means shares of capital stock, partnership interests, membership interests in a limited liability company, beneficial interests in a trust or other equity ownership interests in a Person, and any warrants, options or other rights entitling the holder thereof to purchase or acquire any such Equity Interest.

"Equivalent Amount" means, on any date, the equivalent amount of currency after giving effect to a conversion of a specified amount of currency at the noon rate of exchange for Canadian interbank transactions established by the Bank of Canada for the day in question, or, if such rate is for any reason unavailable, at the spot rate quoted for wholesale transactions by the Lender at approximately noon (Edmonton time) on that date in accordance with its normal practice.

"Event of Default" has the meaning set out in Section 12.1.

"Financial Assistance" means, with respect to any person and without duplication, any loan, guarantee, indemnity, assurance, acceptance, extension of credit, loan purchase, share purchase, equity or capital contribution, investment or other form of direct or indirect financial assistance or support of any other person or any obligation (contingent or otherwise) primarily for the purpose of enabling another person to incur or pay any Debt or to comply with agreements relating thereto or otherwise to assure or protect creditors of the other person against loss in respect of Debt of the other person and includes any guarantee of or indemnity in respect of the Debt of the other person and any absolute or contingent obligation to (directly or indirectly):

- 8 -

- (a) advance or supply funds for the payment or purchase of any Debt of any other person;
- (b) purchase, sell or lease (as lessee or lessor) any property, assets, goods, services, materials or supplies primarily for the purpose of enabling any person to make payment of Debt or to assure the holder thereof against loss;
- (c) guarantee, indemnify, hold harmless or otherwise become liable to, any creditor of any other person from or against any losses, liabilities or damages in respect of Debt;
- (d) make a payment to another for goods, property or services regardless of the non-delivery or non-furnishing thereof to the Borrower or any Subsidiary (as applicable); or
- (e) make an advance, loan or other extension of credit to another person, or make any subscription for equity, capital contribution, or investment in another person, or maintain the capital, working capital, solvency or general financial condition of another person.

The amount of any Financial Assistance is the amount of any loan or direct or indirect financial assistance or support, without duplication, given, or all Debt of the obligor to which the Financial Assistance relates, unless the Financial Assistance is limited to a determinable amount, in which case the amount of the Financial Assistance is the determinable amount.

"Financial Instrument" means any currency swap agreement, cross currency agreement, forward agreement, floor, cap or collar agreement, futures or options, insurance or other similar agreement or arrangement, or any combination thereof, entered into by the Borrower or a Subsidiary where the subject matter of the same is currency exchange rates or the price, value or amount payable thereunder is dependent or based upon currency exchange rates or fluctuations in currency exchange rates as in effect from time to time.

"Financial Instrument Demand for Payment" means a demand made by the Lender pursuant to a Lender Financial Instrument demanding payment of the Financial Instrument Obligations which are then due and payable relating thereto and shall include, without limitation, any notice under any agreement evidencing a Lender Financial Instrument which, when delivered, would require an early termination thereof and a payment by the Borrower or a Subsidiary in settlement of obligations thereunder as a result of such early termination.

"Financial Instrument Obligations" means obligations arising under Financial Instruments entered into by the Borrower or a Subsidiary to the extent of the net amount due or accruing due by the Borrower or Subsidiary thereunder (determined by marking to market the same in accordance with their terms).

"fiscal quarter" means the three month period commencing on the first day of each fiscal year and each successive three month period thereafter during such fiscal year.

"fiscal year" means the fiscal year of the Borrower which commences on October 1 of each year and ends on September 30 of each subsequent year.

- 9 -

"**Fixed Charge Coverage Ratio**" means, as at a Quarter End, the ratio of (a) EBITDA for the 12 months ending at such Quarter End <u>less</u> (i) maintenance capital expenditures <u>plus</u> (ii) amounts paid or the provision for federal state, local or foreign income taxes for such period, on a consolidated basis, to (b) the Fixed Charges for the 12 months ending at such Quarter End; provided that, if any Debt has created, issued, incurred, assumed, Guaranteed, repaid, repurchased or redeemed during such 12 month period (other than Debt under a revolving facility), then the Fixed Charge Coverage Ratio will be calculated giving pro forma effect thereto and the use of proceeds therefrom as if the same had occurred at the beginning of such 12 month period.

"Fixed Charges" means, for any period of determination, the sum of the following determined on a consolidated basis and without duplication:

- (a) the Interest Expense for such period; plus
- (b) the amount of Debt which has a scheduled due date or is otherwise required to be repaid or paid, as the case may be, during such period (which, for certainty, does not include voluntary prepayments);

provided that, if Powell has guaranteed any Debt of any person which is not the Borrower or a Subsidiary, then the Fixed Charges shall be determined as if such Debt was directly created, issued, incurred or assumed by Powell.

"**F/X Facility**" means the foreign exchange forward contract facility in the maximum principal amount of U.S. \$2,000,000.00 to be made available to the Borrower by the Lender in accordance with the provisions hereof, subject to any reduction in accordance with the provisions hereof, and to be governed by the terms of the Lender's Agreement for Foreign Exchange Contracts.

"F/X Facility Commitment" means the maximum principal amount the Lender has agreed to make available to the Borrower under the F/X Facility, being U.S. \$2,000,000.00, subject to reductions or adjustments pursuant to the terms hereof.

"GAAP" means generally accepted accounting principles in the United States set out in the opinions and pronouncements of the Accounting Principles Board and the American Institute of Certified Public Accountants and statements and pronouncements of the Financial Accounting Standards Board or such other principles as may be approved by a significant segment of the accounting profession in the United States, that are applicable to the circumstances as of the date of determination, consistently applied.

"Governmental Authority" means any federal, provincial, state, regional, municipal or local government or any department, agency, board, tribunal or authority thereof or other political subdivision thereof and any entity or person exercising executive, legislative, judicial, regulatory or administrative functions of, or pertaining to, government or the operation thereof.

"Governmental Authorization" means an authorization, order, permit, approval, grant, license, consent, right, franchise, privilege, certificate, judgment, writ, injunction, award, determination, direction, decree or demand or the like issued or granted by law or by rule or regulation of any Governmental Authority.

"Guarantee" means any guarantee, undertaking to assume, endorse, contingently agree to purchase or to provide funds for the payment of, or otherwise become liable in respect of, any obligation of any person; provided that the amount of each Guarantee shall be deemed to be the amount of the obligation guaranteed thereby, unless the Guarantee is limited to a determinable amount in which case the amount of such Guarantee shall be deemed to be the lesser of such determinable amount or the amount of such obligation.

- 10 -

"Hazardous Materials" means any substance or mixture of substances which, if released into the environment, would likely cause, immediately or at some future time, harm or degradation to the environment or to human health or safety and includes any substance defined as or determined to be a pollutant, contaminant, waste, hazardous waste, hazardous chemical, hazardous substance, toxic substance or dangerous good under any Environmental Law.

"Hedging Affiliate" means any Affiliate of a Lender which enters into Financial Instrument.

"HSBC" means HSBC Bank Canada, a Canadian chartered bank.

"Indemnified Parties" means, collectively, the Lender, including a receiver, receiver manager or similar person appointed under Applicable Laws, and their respective shareholders, Affiliates, officers, directors, employees and agents, and "Indemnified Party" means anyone of the foregoing.

"Indemnified Third Party" has the meaning set out in Section 14.3.

"Information" has the meaning set out in Section 15.1.

"Intellectual Property" means, collectively, patents, patents pending, copyrights, proprietary processes or programs, industrial designs, trademarks, trademark applications, tradenames and other intellectual property of every nature and kind.

"Interest Expense" means, for any period, without duplication, interest expense determined on a consolidated basis in accordance with GAAP as the same would be set forth or reflected in a consolidated statement of income and, in any event and without limitation, shall include:

- (a) all interest accrued or payable in respect of such period, including capitalized interest;
- (b) all fees (including standby, commitment and stamping fees and fees payable in respect of letters of credit and letters of guarantee supporting obligations which constitute Debt, but excluding the fees payable on the date hereof in relation to the establishment of the Credit Facilities accrued or payable in respect of such period and which relate to any indebtedness or credit agreement, prorated (as required) over such period);
- (c) any difference between the face amount and the discount proceeds of any bankers' acceptances, commercial paper and other obligations issued at a discount, prorated (as required) over such period.

"Interest Payment Date" means with respect to each Canadian Prime Rate Loan or U.S. Base Rate Loan, the first Banking Day of each calendar month provided that, in any case, the Maturity Date or, if applicable, any earlier date on which the Credit Facilities are fully cancelled or permanently reduced in full, shall be an Interest Payment Date with respect to all Loans then outstanding under the Credit Facilities.

- 11 -

"Interest Period" means:

- (a) with respect to each Canadian Prime Rate Loan or U.S. Base Rate Loan, the period commencing on the applicable Drawdown Date or Conversion Date, as the case may be, and terminating on the date selected by the Borrower hereunder for the Conversion of such Loan into another type of Loan or for the repayment of such Loan;
- (b) with respect to each Bankers' Acceptance, the period selected by the Borrower hereunder and being of one (I), two (2) or three (3) months' duration, subject to market availability, (or, subject to the agreement of the Lender, a longer or shorter period) commencing on the Drawdown Date, Rollover Date or Conversion Date of such Loan, provided that the last day of all Interest Periods for Bankers' Acceptances outstanding under the Revolving Facility shall expire on or prior to the Maturity Date; and
- (c) with respect to each Letter of Credit, the period commencing on the date of issuance of such Letter of Credit and terminating on the last day the Letter of Credit is outstanding.

"**Investment**" means (a) any purchase or other acquisition of shares or other securities of any person, (b) any form of Financial Assistance to or for the benefit of any person, (c) any capital contribution to any other person and (d) any purchase or other acquisition of any assets, property or undertaking other than an acquisition in the ordinary course of business of the purchaser.

"Judgment Conversion Date" has the meaning set out in Section 14.4.

"Judgment Currency" has the meaning set out in Section 14.4.

"LC Fee" means the fee charged by the Lender for issuing a Letter of Credit under the Revolving Facility, which shall be calculated by the Lender in accordance with Section 5.5.

"Lender's Account" means the account maintained by the Lender to which payments and transfers under this Agreement are to be effected as designated by written notice to the Borrower from time to time.

"Lender Financial Instrument" means a Financial Instrument entered into between the Lender or a Hedging Affiliate and the Borrower or a Subsidiary.

"Lender Financial Instrument Obligations" means, collectively, all of the obligations, indebtedness and liabilities (present or future, absolute or contingent, mature or not) of the Borrower and its Subsidiaries under, pursuant or relating to any and all Lender Financial Instruments.

"Letter of Credit" or "LC" means a Canadian Dollar financial or non-financial standby letter of credit, documentary letter of credit or letter of guarantee issued by and in a form satisfactory to the Lender, which Letters of Credit shall be issued at the request of and for the account of the Borrower pursuant to Article 7 for the purpose of supporting the Borrower's and its Subsidiaries' ordinary course business operations.

"Loans" means all Canadian Prime Rate Loans, U.S. Base Rate Loans, Bankers' Acceptances and Letters of Credit outstanding hereunder as well as the amounts owing under the EFT Facility, MC Facility, F/X Facility and DSL Facility, and "Loan" means anyone of the foregoing.

- 12 -

"Material Adverse Effect" means a material adverse effect on:

- (a) the financial condition of Powell or on the Borrower and its Subsidiaries on a consolidated basis, as the case may be;
- (b) the ability of Powell, the Borrower or any of its Subsidiaries to observe or perform its obligations under the Documents to which it is a party or the validity or enforceability of such Documents or any material provision thereof;
- (c) the property, business, operations, expected net cash flows of Powell or the Borrower, liabilities or capitalization of Powell or of the Borrower and its Subsidiaries on a consolidated basis, as the case may be; or
- (d) the Security, the priority thereof or any right or remedy of the Lender thereunder.

"Material Agreements" means, collectively, each agreement, lease, instrument, indenture or other document to which the Borrower or any Subsidiary is a party and which if terminated or released (without replacement) or if the counterparty thereto defaulted in its performance thereof, such termination, release or default would have or would reasonably be expected to have a Material Adverse Effect.

"Maturity Date" means March 1, 2015.

"**MC Facility**" means the Mastercard facility in the maximum principal amount of Cdn. \$500,000.00 to be made available to the Borrower by the Lender in accordance with the provisions hereof, subject to any reduction in accordance with the provisions hereof, and to be governed by the terms of the Lender's Mastercard Program Application and the Lender's Mastercard Cardholder Application.

"**MC Facility Commitment**" means the maximum principal amount the Lender has agreed to make available to the Borrower under the MC Facility, being Cdn. \$500,000.00, subject to reductions or adjustments pursuant to the terms hereof.

"Moody's" means Moody's Investors Service, Inc. and any successors thereto.

"Net Income" means, in respect of the period for which it is being determined, the net income determined on a consolidated basis in accordance with GAAP.

"**Obligations**" means, at any time and from time to time, all of the obligations, indebtedness and liabilities (present or future, absolute or contingent, matured or not) of the Borrower and its Subsidiaries to the Lender under, pursuant or relating to the Documents, the Credit Facilities and whether the same are from time to time reduced and thereafter increased or entirely extinguished and thereafter incurred again and including, without limitation, all principal, interest, fees, legal and other costs, charges and expenses, and other amounts payable by the Borrower under this Agreement.

"Officer's Certificate" means a certificate or notice (other than a Compliance Certificate) signed by anyone of the president, chief financial officer, a vice president, treasurer, assistant treasurer, controller, corporate secretary or assistant secretary of the Borrower or Subsidiary, as the case may be, (including, in the case of a partnership a certificate or notice signed by such an officer of a general partner of such partnership); provided, however, that Drawdown Notices, Conversion Notices, Rollover Notices and Repayment Notices shall be executed on behalf of the Borrower by anyone of the foregoing persons or such other persons as may from time to time be designated by written notice from the Borrower to the Lender.

- 13 -

"**Outstanding Principal**" means, at any time, the aggregate of (a) the principal amount of all outstanding Canadian Prime Rate Loans, (b) the principal amount of all outstanding U.S. Base Rate Loans, (c) the amounts payable at maturity of all outstanding Bankers' Acceptances, (d) the face amount of all outstanding and undrawn Letters of Credit, as well as the principal amount outstanding under the EFT Facility, MC Facility, F/X Facility and the DSL Facility if not otherwise hereinbefore referred to.

"Ownership Interest" means:

- (a) in respect of a body corporate, any shares in the capital of such body corporate;
- (b) in respect of a partnership, any partnership units or interests in such partner and includes any other income, capital, beneficial or ownership interest (however designated) in such partnership; and
- (c) in respect of any other person, any income, capital, beneficial or ownership interest (however designated) in such person (including, for certainty, any trust units in the case of a trust),

in each case, whether any of the foregoing are voting or non-voting and including any securities, instruments or contractual rights capable of being converted into, exchanged or exercised for any of the foregoing (including options, warrants, conversion or exchange privileges and similar rights).

"Permitted Contest" means action taken by or on behalf the Borrower or Subsidiary in good faith by appropriate proceedings diligently pursued to contest a Tax, claim or Security Interest, provided that:

- (a) the person to which the Tax, claim or Security Interest being contested is relevant (and, in the case of a Subsidiary, the Borrower on a consolidated basis) has established reasonable reserves therefore if and to the extent required by GAAP;
- (b) proceeding with such contest does not have, and would not reasonably be expected to have, a Material Adverse Effect; and
- (c) proceeding with such contest will not create a material risk of sale, forfeiture or loss of, or interference with the use or operation of, a material part of the property, assets and undertaking of the Borrower and its Subsidiaries.

"Permitted Debt" means the following:

- (a) the Obligations;
- (b) any Debt owing by the Borrower or any of its Subsidiaries to Powell;
- (c) any Debt owing by a Subsidiary to the Borrower, by a Subsidiary to a Subsidiary which is a Wholly-Owned Subsidiary and by the Borrower to a Subsidiary which is a Wholly-Owned Subsidiary;

- 14 -

- (d) Attributable Debt of the Borrower or any Subsidiary arising in connection with operating leases entered into in the ordinary course of business (which, for certainty, shall not include any operating leases entered into in connection with any Sale-Leaseback);
- (e) Debt arising pursuant to the indemnification, purchase price adjustment or similar provisions of agreements entered into by the Borrower or any Subsidiary in connection with acquisitions, Investments or dispositions permitted hereunder, or pursuant to guarantees, letters of credit, surety bonds or performance bonds provided to secure the performance of the Borrower or any Subsidiary pursuant to such agreements; provided that, for certainty, in no event shall such Debt include any Debt specified in subparagraphs (a) to (h), inclusive, of the definition thereof; and
- (f) Debt consisting of Financial Assistance permitted under Section 10.2(i).

"Permitted Disposition" means, in respect of the Borrower or any of its Subsidiaries, any of the following:

- (a) a sale or disposition by the Borrower or such Subsidiary in the ordinary course of business and in accordance with sound industry practice of tangible personal property that is obsolete, no longer useful for its intended purpose or being replaced in the ordinary course of business;
- (b) a sale or disposition of assets (including shares or ownership interests) by a Subsidiary to the Borrower, by a Subsidiary to a Subsidiary which is a Wholly-Owned Subsidiary and by the Borrower to a Subsidiary which is a Wholly-Owned Subsidiary;
- (c) a sale or disposition by the Borrower or any Subsidiary of its interest in machinery, equipment or other tangible personal property for which Purchase Money Obligations were incurred and (i) such Purchase Money Obligations are fully repaid concurrently with such sale or disposition and (ii) such sale or disposition is made in the ordinary course of business at fair market value to a person at arm's length from the Borrower and its Subsidiaries; and
- (d) any sale or disposition of inventory of the Borrower or such Subsidiary, provided that such sale or disposition is made in the ordinary course of business for cash consideration not less than the fair market value of the inventory being sold or otherwise disposed of.

"Permitted Encumbrances" means as at any particular time any of the following encumbrances on the property or any part of the property of the Borrower or any Subsidiary:

- (a) liens for taxes, assessments or governmental charges not at the time due or delinquent or, if due or delinquent, the validity of which is being contested at the time by a Permitted Contest;
- (b) deemed liens and trusts arising by operation of law in connection with workers' compensation, employment insurance and other social security legislation, in each case, which secure obligations not at the time due or delinquent or, if due or delinquent, the validity of which is being contested at the time by a Permitted Contest;

- 15 -

- (c) liens under or pursuant to any judgment rendered, or claim filed, against the Borrower or Subsidiary, which the Borrower or Subsidiary (as applicable) shall be contesting at the time by a Permitted Contest;
- (d) undetermined or inchoate liens and charges incidental to construction or current operations which have not at such time been filed pursuant to law
 against the Borrower or Subsidiary or which relate to obligations not due or delinquent or, if due or delinquent, the validity of which is being
 contested at the time by a Permitted Contest;
- (e) easements, rights of way, servitudes or other similar rights in land (including, without in any way limiting the generality of the foregoing, rights of way and servitudes for railways, sewers, drains, gas and oil and other pipelines, gas and water mains, electric light and power and telecommunication, telephone or telegraph or cable television conduits, poles, wires and cables) granted to or reserved or taken by other persons which individually or in the aggregate do not materially detract from the value of the land concerned or materially impair its use in the operation of the business of the Borrower and its Subsidiaries, taken as a whole;
- (f) security given by the Borrower or Subsidiary to a public utility or any municipality or governmental or other public authority when required by such utility or municipality or other authority in connection with the operations of the Borrower or Subsidiary (as applicable), all in the ordinary course of its business which individually or in the aggregate do not materially detract from the value of the asset concerned or materially impair its use in the operation of the business of the Borrower and its Subsidiaries, taken as a whole;
- (g) the reservation in any original grants from the Crown of any land or interests therein and statutory exceptions and reservations to title;
- (h) Security Interests in favour of the Lender;
- (i) the Security;
- (j) any operating lease entered into in the ordinary course of business (which, for certainty, shall not include any operating leases entered into in connection with any Sale-Leaseback);
- (k) bankers' liens, rights of set-off and other similar liens existing solely with respect to cash and Approved Securities on deposit in one or more accounts maintained by the Borrower or any of its Subsidiaries, in each case, granted in the ordinary course of business in favour of the Lender with which such accounts are maintained, securing amounts owing to the Lender with respect to cash management and operating account arrangements, including those involving pooled accounts and netting arrangements;
- (l) to the extent constituting Security Interests, Financial Assistance permitted under this Agreement;

- 16 -

- (m) Security Interests securing Attributable Debt, provided that such Security Interests shall attach only to the property subject to the lease giving rise to such Attributable Debt and provided further that such Attributable Debt is Permitted Debt;
- Security Interests securing a Purchase Money Obligation, provided that such Security Interests shall attach only to the property acquired in connection with which such Purchase Money Obligation was incurred (and proceeds thereof) and provided further that such Purchase Money Obligation is Permitted Debt;
- (o) landlords' liens or any other rights of distress reserved in or exercisable under any lease of real property for rent and for compliance with the terms of such lease; provided that such lien does not attach generally to all or substantially all of the undertaking, assets and property of the Borrower or any Subsidiary;
- (p) pledges or deposits to secure performance of (i) bids, tenders, contracts (other than contracts for the payment of money) or (ii) leases of real property, in each case, to which the Borrower or a Subsidiary is a party;
- (q) Security Interests resulting from the deposit of cash or Approved Securities as security when the Borrower or a Subsidiary is required to do so by a Governmental Authority or by normal business practice in connection with contracts, licenses or tenders or similar matters in the ordinary course of business and for the purpose of carrying on the same, or to secure workers' compensation, surety or appeal bonds or to secure costs of litigation when required by Applicable Law;
- (r) minor defects of title which, individually and in the aggregate, do not materially affect the right of ownership of the Borrower or its Subsidiary in and to the property affected thereby or the right of the Borrower or its Subsidiary to utilize such property to conduct its business; and
- (s) any extension, renewal or replacement (or successive extensions, renewals or replacements), as a whole or in part, of any Security Interest referred to in the preceding subparagraphs (a) to (r) inclusive of this definition, so long as any such extension, renewal or replacement of such Security Interest is limited to all or any part of the same property that secured the Security Interest extended, renewed or replaced (plus improvements on such property) and the indebtedness or obligation secured thereby is not increased;

provided that nothing in this definition shall in and of itself cause the Obligations hereunder or the other obligations secured by the Security to be subordinated in priority of payment to any such Permitted Encumbrance or cause any Security Interests in favour of the Lender to rank subordinate to any such Permitted Encumbrance.

"**Permitted Hedging**" means Financial Instruments which are entered into in the ordinary course of business and for hedging purposes and not for speculative purposes (determined, where relevant, by reference to GAAP).

"**Permitted Subordinated Debt**" means any Debt owed by Powell to another person which Debt is specifically postponed and assigned to the Lender as security for the payment of the Obligations, or any of them.

- 17 -

"**Person**" means an individual, a partnership, a corporation, a limited or unlimited liability company, a joint venture, a trust, an unincorporated organization, a union, a government or any department or agency thereof and the heirs, executors, administrators or other legal representatives of an individual, and words importing persons have a similar meaning.

"**Power of Attorney**" means a power of attorney provided by the Borrower to the Lender with respect to Bankers' Acceptances in accordance with and pursuant to Section 6.4 hereof.

"**Purchase Money Obligation**" means any monetary obligation created or assumed as part of the purchase price of real or tangible personal property, whether or not secured, any extensions, renewals or refundings of any such obligation, provided that the principal amount of such obligation outstanding on the date of such extension, renewal or refunding is not increased and further provided that any security given in respect of such obligation shall not extend to any property other than the property acquired in connection with which such obligation was created or assumed and fixed improvements, if any, erected or constructed thereon and the proceeds thereof.

"Quarter End" means September 30, December 31, March 31 and June 30, in each year.

"Realization Proceeds" has the meaning set out in Section 12.7.

"Related Party" means any person which is anyone or more of the following:

- (a) an Affiliate of the Borrower or any Subsidiary thereof;
- (b) a shareholder or partner of the Borrower or any Subsidiary which, together with all Affiliates of such person, owns or controls, directly or indirectly, more than 10% of the Ownership Interests of the Borrower or any Subsidiary, or an Affiliate of any such shareholder or partner;
- (c) an officer or director of any of the foregoing; and
- (d) a person which is not at arm's length from the Borrower and its Subsidiaries.

"**Release**" means any release, spill, emission, leak, pumping, injection, deposit, disposal, discharge, dispersal, leaching or migration into the environment including, without limitation, the movement of Hazardous Materials through ambient air, soil, surface water, ground water, wetlands, land or sub-surface strata.

"**Required Permits**" means all Governmental Authorizations which are necessary at any given time for the Borrower and each of its Subsidiaries to own and operate its property, assets, rights and interests or to carry on its business and affairs.

"Revolving Facility" means the revolving credit facility in the maximum principal amount of Cdn. \$10,000,000.00 (or the Equivalent Amount thereof in United States Dollars) to be made available to the Borrower by the Lender in accordance with the provisions hereof, subject to any reduction in accordance with the provisions hereof.

"Revolving Facility Commitment" means the maximum principal amount the Lender has agreed to make available to the Borrower under the Revolving Facility, being Cdn. \$10,000,000.00, subject to reductions or adjustments pursuant to the terms hereof.

- 18 -

"Rollover" means with respect to Bankers' Acceptances, the issuance of new Bankers' Acceptances in respect of all or any portion of Bankers' Acceptances maturing at the end of the Interest Period applicable thereto, all in accordance with Article 6 hereof.

"Rollover Date" means the date of commencement of a new Interest Period applicable to a Loan and which shall be a Banking Day.

"Rollover Notice" means a notice substantially in the form annexed hereto as Schedule D to be given to the Lender by the Borrower pursuant hereto.

"S&P" means the Standard & Poor's Rating Group (a division of The McGraw-Hill Companies, Inc.) and any successors thereto.

"Sale-Leaseback" means an arrangement, transaction or series of arrangements or transactions under which title to any real property, tangible personal property or fixture is transferred by the Borrower or a Subsidiary (a "transferor") to another person which leases or otherwise grants the right to use such property to the transferor (or nominee of the transferor) and, whether or not in connection therewith, the transferor also acquires a right or is subject to an obligation to acquire such property or a material portion thereof, and regardless of the accounting treatment of such arrangement, transaction or series of arrangements or transactions.

"Security" means the security and Security Interests created under the Security Documents.

"Security Documents" means the security and related agreements executed and delivered, or required to be executed and delivered, in favour of the Lender by Powell, the Borrower and the Subsidiaries pursuant to Section 11.1, as amended, supplemented, replaced or otherwise modified from time to time.

"Security Interest" means mortgages, charges, pledges, hypothecs, assignments by way of security, conditional sales or other title retentions, security created under the *Bank Act* (Canada), liens, encumbrances, security interests or other interests in property, howsoever created or arising, whether fixed or floating, perfected or not, which secure payment or performance of an obligation and, including, in any event:

- (a) deposits or transfers of cash, marketable securities or other financial assets under any agreement or arrangement whereby such cash, securities or assets may be withdrawn, returned or transferred only upon fulfillment of any condition as to the discharge of any other indebtedness or other obligation to any creditor;
- (b) (i) rights of set-off or (ii) any other right of or arrangement of any kind with any creditor, which in any case are made, created or entered into, as the case may be, for the purpose of or having the effect (directly or indirectly) of (A) securing Debt, (B) preferring some holders of Debt over other holders of Debt or (C) having the claims of any creditor be satisfied prior to the claims of other creditors with or from the proceeds of any properties, assets or revenues of any kind now owned or later acquired (other than, with respect to (C) only, rights of set-off granted or arising in the ordinary course of business);
- (c) the rights of lessors under capital leases, operating leases and any other lease financing; and
- (d) absolute assignments of accounts receivable.

- 19 -

"Shareholders' Equity" means with respect to any person, all equity of such person and its Subsidiaries which would, in accordance with GAAP, be classified upon a consolidated balance sheet of such person as equity of such person and its Subsidiaries and, whether or not so classified, shall include (without duplication) preferred capital.

"Subsidiary" means, with respect to any person ("X"):

- (a) any corporation of which at least a majority of the outstanding shares having by the terms thereof ordinary voting power to elect a majority of the board of directors of such corporation (irrespective of whether at the time shares of any other class or classes of such corporation might have voting power by reason of the happening of any contingency, unless the contingency has occurred and then only for as long as it continues) is at the time directly, indirectly or beneficially owned or controlled by X or one or more of its Subsidiaries, or X and one or more of its Subsidiaries;
- (b) any partnership of which, at the time, X, or one or more of its Subsidiaries, or X and one or more of its Subsidiaries: (i) directly, indirectly or beneficially own or control more than 50% of the income, capital, beneficial or ownership interests (however designated) thereof; and (ii) is a general partner, in the case of limited partnerships, or is a partner or has authority to bind the partnership, in all other cases; or
- (c) any other person of which at least a majority of the income, capital, beneficial or ownership interests (however designated) are at the time directly, indirectly or beneficially owned or controlled by X, or one or more of its Subsidiaries, or X and one or more of its Subsidiaries,

provided that, unless otherwise expressly provided or the context otherwise requires, references herein to "Subsidiary" or "Subsidiaries" shall be and shall be deemed to be references to Subsidiaries of the Borrower, and shall include Nextron and PPC Technical.

"**Taxes**" means all taxes, levies, imposts, stamp taxes, duties, fees, deductions, withholdings, charges, compulsory loans or restrictions or conditions resulting in a charge which are imposed, levied, collected, withheld or assessed by any country or political subdivision or taxing authority thereof now or at any time in the future, together with interest thereon and penalties, charges or other amounts with respect thereto, if any, and "Taxa" and "Taxation" shall be construed accordingly.

"Termination Event" means an automatic early termination of obligations relating to a Lender Financial Instrument under any agreement relating thereto without any notice being required from a Lender.

"**U.S. Base Rate**" means the variable annual rate of interest per annum established from time to time by the Lender as the reference rate of interest for the determination of interest rates that the Lender will charge to customers in Canada for United States Dollar base rate loans in Canada and which was 3.75%pa on March 1, 2012.

- 20 -

"**U.S. Base Rate Loan**" means an Advance in United States Dollars made by the Lender to the Borrower under the Revolving Facility with respect to which the Borrower has specified or a provision hereof requires that interest is to be calculated by reference to the U.S. Base Rate.

"U.S. Dollars" and "US\$" means the lawful money of the United States of America.

"Uniform Customs" has the meaning set out in Section 7.4.

"Wholly-Owned Subsidiary" means:

- (a) a corporation, all of the issued and outstanding shares in the capital of which are beneficially held by:
 - (i) the Borrower or Powell, as the context requires;
 - (ii) the Borrower or Powell, as the context requires, and one or more corporations, all of the issued and outstanding shares in the capital of which are held by the Borrower or Powell, as the context requires; or
 - (iii) two or more corporations, all of the issued and outstanding shares in the capital of which are held by the Borrower or Powell, as the context requires;
- (b) a corporation which is a Wholly-Owned Subsidiary of a corporation that is a Wholly-Owned Subsidiary of the Borrower or Powell, as the context requires;
- (c) a partnership, all of the partners of which are the Borrower or Powell, as the context requires, and/or Wholly-Owned Subsidiaries of the Borrower or Powell, as the context requires; or
- (d) any person of which all of the income, capital, beneficial and ownership interests (however designated) are beneficially owned and controlled by the Borrower or Powell, as the context requires, and/or Wholly-Owned Subsidiaries of the Borrower or Powell, as the context requires.

1.2 Headings; Articles and Sections

The division of this Agreement into Articles and Sections and the insertion of headings are for convenience of reference only and shall not affect the construction or interpretation of this Agreement. The terms "this Agreement", "hereof", "hereunder" and similar expressions refer to this Agreement and not to any particular Article, Section or other portion hereof and include any agreement supplemental hereto. Unless something in the subject matter or context is inconsistent therewith, references herein to Articles and Sections are to Articles and Sections of this Agreement.

1.3 Number; persons; including

Words importing the singular number only shall include the plural and vice versa, words importing the masculine gender shall include the feminine and neuter genders and vice versa, words importing persons shall include individuals, partnerships, associations, trusts, unincorporated organizations and corporations and *vice versa* and words and terms denoting inclusiveness (such as "include" or "includes" or "including"), whether or not so stated, are not limited by their context or by the words or phrases which precede or succeed them.

- 21 -

1.4 Application of Accounting Principles

Where the character or amount of any asset or liability or item of revenue or expense or amount of equity is required to be determined, or any consolidation or other accounting computation is required to be made for the purpose of this Agreement or any other Document, such determination or calculation shall, to the extent applicable and except as otherwise specified herein or as otherwise agreed in writing by the parties, be made in accordance with GAAP applied on a consistent basis.

1.5 <u>References to Agreements and Enactments</u>

Reference herein to any agreement, instrument, licence or other document shall be deemed to include reference to such agreement, instrument, licence or other document as the same may from time to time be amended, modified, supplemented or restated in accordance with the provisions of this Agreement if and to the extent such provisions are applicable; and reference herein to any enactment shall be deemed to include reference to such enactment as re-enacted, amended or extended from time to time and to any successor enactment.

1.6 Per Annum Calculations

Unless otherwise stated, wherever in this Agreement reference is made to a rate "per annum" or a similar expression is used, such rate is expressed on the basis of, and shall be calculated on the basis of, a year of 365 days.

1.7 References to Borrower

References in this Agreement to actions and steps by, or the performance of the terms and conditions hereof by, the Borrower shall, if the context requires, be and shall be construed as being by the general partner on behalf of and in respect of the Borrower, if the Borrower is a limited partnership.

1.8 Schedules

The following are the Schedules annexed hereto and incorporated by reference and deemed to be part hereof:

Schedule A	-	Compliance Certificate
Schedule B	-	Conversion Notice
Schedule C	-	Drawdown Notice
Schedule D	-	Rollover Notice
Schedule E	-	Disclosure Schedule

- 22 -

ARTICLE 2 THE CREDIT FACILITIES

2.1 The Credit Facilities

(a) <u>INTENTIONALLY DELETED</u>

- (b) <u>Revolving Facility.</u> Subject to the terms and conditions hereof, the Lender shall make available to the Borrower the Revolving Facility. The Outstanding Principal under the Revolving Facility shall not exceed the Revolving Facility Commitment.
- (c) <u>EFT Facility.</u> Subject to the terms and conditions hereof, the Lender shall make available to the Borrower the EFT Facility. The Outstanding Principal under the EFT Facility shall not exceed the EFT Facility Commitment.
- (d) <u>MC Facility.</u> Subject to the terms and conditions hereof, the Lender shall make available to the Borrower the MC Facility. The Outstanding Principal under the MC Facility shall not exceed the MC Facility Commitment.
- (e) <u>F/X Facility.</u> Subject to the terms and conditions hereof, the Lender shall make available to the Borrower the F/X Facility. The Outstanding Principal under the F/X Facility shall not exceed the F/X Facility Commitment.
- (f) <u>DSL Facility.</u> Subject to the terms and conditions hereof, the Lender shall make available to the Borrower the DSL Facility. The Outstanding Principal under the DSL Facility shall not exceed the DSL Facility Commitment.

2.2 Types of Availments

(a) <u>INTENTIONALLY DELETED</u>

(b) <u>Revolving Facility.</u> The Borrower may make Drawdowns, Conversions and Rollovers under the Revolving Facility of (1) Canadian Prime Rate Loans, (2) U.S. Base Rate Loans, (3) Bankers' Acceptances, and (4) Letters of Credit with terms of up to one year in Canadian Dollars. The Borrower shall have the option, subject to the terms and conditions hereof, to determine which types of Loans shall be drawn down and in which combinations or proportions.

2.3 Purpose

- (a) <u>INTENTIONALLY DELETED</u>
- (b) <u>Revolving Facility</u>. The Revolving Facility is being made available for working capital and other general corporate purposes of the Borrower.
- (c) <u>EFT Facility.</u> The EFT Facility is being made available to assist the Borrower with the payment of payroll and other payables.
- (d) <u>MC Facility.</u> The MC Facility is being made available for the issuance of expense cards to key employees of the Borrower.

- 23 -

- (e) <u>F/X Facility.</u> The F/X Facility is being made available for the purchase of forward contracts of U.S. Dollars with a maximum maturity of twelve (12) months to hedge against currency fluctuations.
- (f) <u>DSL Facility.</u> The DSL Facility is being made available for spot foreign exchange transactions.

2.4 Availability and Nature of the Credit Facilities

- (a) <u>INTENTIONALLY DELETED</u>
- (b) <u>Revolving Facility</u>. The Revolving Facility is a revolving credit facility: prior to the applicable Maturity Date or the occurrence of an Event of Default, the Outstanding Principal under the Revolving Facility may revolve and Borrower may borrow, repay and re-borrow up to the Revolving Facility Commitment, but only after the completion and registration, if applicable, of the Documents and the satisfaction of the Conditions Precedent to Drawdown as set out in Article 3 hereof.
- (c) <u>EFT Facility, MC Facility, F/X Facility and DSL Facility</u>. The EFT Facility, MC Facility, F/X Facility and DSL Facility shall be available to the Borrower after the completion and registration, if applicable, of the Documents and the satisfaction of the Conditions Precedent to Drawdown as set out in Article 3 hereof.

2.5 Margin Requirements

The obligations outstanding under the Revolving Facility (including the undrawn face amount of Letters of Credit) shall at no time exceed the Revolving Facility Commitment, or

2.6 Notice Periods for Drawdowns, Conversions and Rollovers

Subject to the provisions hereof, the Borrower may make a Drawdown, Conversion or Rollover under the Revolving Facility by delivering a Drawdown Notice, Conversion Notice or Rollover Notice, as the case may be (executed in accordance with the definition of Officer's Certificate), with respect to a specified type of Loan to the Lender not later than:

- (a) 10:30 a.m. (Edmonton time) three (3) Banking Days prior to the proposed Drawdown Date, Conversion Date or Rollover Date, as the case may be, for the Drawdown of, Conversion into or Rollover of Bankers' Acceptances;
- (b) 10:30 a.m. (Edmonton time) three (3) Banking Day prior to the proposed Drawdown Date or Conversion Date, as the case may be, for Drawdowns of or Conversions into Canadian Prime Rate Loans or U.S. Base Rate Loans; and
- (c) 10:30 a.m. (Edmonton time) three (3) Banking Days prior to the proposed Drawdown Date for the issuance of a Letter of Credit, which shall be accompanied by the appropriate application and indemnity in the applicable Lender's customary form.

- 24 -

2.7 Currency Determination

In respect of any Drawdown, Conversion or Rollover under the Revolving Facility in United States Dollars, the Lender will determine the Equivalent Amount in Canadian Dollars for such Loan as of the date of the Drawdown Notice, Conversion Notice or Rollover Notice, as applicable.

2.8 Conversion Option

Subject to the provisions of this Agreement, the Borrower may convert the whole or any part of any type of Loan under the Revolving Facility into any other type of permitted Loan under the Revolving Facility by giving the Lender a Conversion Notice in accordance herewith; provided that:

- (a) Conversions of Bankers' Acceptances may only be made on the last day of the Interest Period applicable thereto;
- (b) a Conversion shall not result in an increase in Outstanding Principal; increases in Outstanding Principal may only be effected by Drawdowns;
- (c) in respect of Conversions of a Loan denominated in one currency to a Loan denominated in another currency, the Borrower shall at the time of the Conversion repay the Loan or portion thereof being converted in the currency in which it was denominated; and
- (d) a Conversion of a Letter of Credit may occur only in accordance with Section 7.3.

2.9 Rollovers and Conversions not Repayments

Any amount converted shall be a Loan of the type converted to upon such Conversion taking place, and any amount rolled over shall continue to be the same type of Loan under the applicable Credit Facility as before the Rollover, but such Conversion or Rollover (to the extent of the amount converted or rolled over) shall not of itself constitute a repayment or a fresh utilization of any part of the amount available under the applicable Credit Facility.

2.10 Irrevocability

A Drawdown Notice, Rollover Notice, Conversion Notice or Repayment Notice given by the Borrower hereunder shall be irrevocable and, subject to any options the Lender may have hereunder in regard thereto and the Borrower's rights hereunder in regard thereto, shall oblige the Borrower to take the action contemplated on the date specified therein.

2.11 Optional Cancellation or Reduction of the Revolving Facility

The Borrower may, at any time, upon giving at least five (5) Banking Days' prior written notice to the Lender, cancel in full or, from time to time, permanently reduce in part the unutilized portion of Revolving Facility; provided, however, that any such reduction shall be in a minimum amount of Cdn. \$500,000.

- 25 -

2.12 Optional Repayment of Credit Facilities

The Borrower may at any time and from time to time repay, without penalty, to the Lender the whole or any part of any Loan owing by it under the Revolving Facility together with accrued interest thereon to the date of such repayment provided that:

- (a) repayments pursuant to this Section may only be made on a Banking Day;
- (b) a Bankers' Acceptance may only be repaid on its maturity unless collateralized in accordance with Section 2.14; and
- (c) a Letter of Credit may only be repaid to the extent it is drawn or returned for cancellation.
- (d) **INTENTIONALLY DELETED**.

2.13 Mandatory Repayment of Credit Facilities

- (a) <u>All Facilities.</u> Subject to Section 12.2, the Borrower shall repay or pay, as the case may be, to the Lender all Loans and other Obligations outstanding under each of the Credit Facilities on or before the Maturity Date.
- (b) **INTENTIONALLY DELETED**

2.14 Cash Collateral

- (a) With respect to the prepayment or cash collateralization of unmatured Bankers' Acceptances required or contemplated pursuant to Section 1.12(b), Section 12.3 or elsewhere in this Agreement, the Borrower shall provide for the funding in full of such unmatured Bankers' Acceptances by paying to and depositing with the Lender cash collateral for each such unmatured Bankers' Acceptances equal to the face amount payable at maturity thereof; such cash collateral deposited by the Borrower shall be held by the Lender in a cash collateral account. Such cash collateral account shall be assigned to the Lender as security for the obligations of the Borrower in relation to such Bankers' Acceptances and the security of the Lender thereby created shall rank in priority to all other Security Interests and adverse claims against such cash collateral. Such cash collateral shall be applied to satisfy the obligations of the Borrower for such Bankers' Acceptances as they mature and the Lender is hereby irrevocably directed by the Borrower to apply any such cash collateral to such maturing Bankers' Acceptances. Amounts held in such cash collateral accounts may not be withdrawn by the Borrower without the consent of the Lender. If after maturity of the Bankers' Acceptances for which such funds are held and application by the Lender of the amounts in such cash collateral accounts to satisfy the obligations of the Borrower hereunder with respect to the Bankers' Acceptances being repaid, any excess remains, such excess shall be promptly paid by the Lender to the Borrower so long as no Default or Event of Default is then continuing.
- (b) With respect to funding the cash collateralization of unexpired Letters of Credit as a result of Section 12.3 or elsewhere in this Agreement, the Borrower shall provide for the funding in full of the unexpired Letters of Credit by paying to and depositing with the Lender, cash collateral for each such unexpired Letter of Credit, equal to the undrawn face amount thereof. Such cash collateral deposited by the Borrower shall be held by the Lender in a cash collateral account, which

- 26 -

shall be assigned to the Lender as security for the obligations of the Borrower in relation to such Letters of Credit and the security of the Lender thereby created in such cash collateral shall rank in priority to all other Security Interests and adverse claims against such cash collateral. Such cash collateral shall be applied to satisfy the obligations of the Borrower for such Letters of Credit if drawn upon and the Lender is hereby irrevocably directed by the Borrower to so apply any such cash collateral. Amounts held in such cash collateral accounts may not be withdrawn by the Borrower without the consent of the Lender. If after the drawing in full or expiration of the Letters of Credit for which such funds are held and application by the Lender of the amounts in such cash collateral accounts to satisfy the obligations of the Borrower hereunder with respect thereto, any excess remains, such excess shall be promptly paid by the Lender to the Borrower so long as no Default or Event of Default is then continuing.

2.15 Commitment and Renewal Fees

The Borrower will pay to the Lender:

- (a) on or before the Closing Date, an up-front commitment fee equal to 0.40% of the Revolving Facility Commitment; and
- (b) on each anniversary of the Closing Date on which any amount remains outstanding or available under any of the Credit Facilities, an annual renewal fee of Cdn. \$5,000.00.

Fees collected by the Lender shall be its property as consideration for the time, effort and expense incurred by it in the review and administration of documents and financial statements, and the Borrower acknowledges and agrees that the determination of these costs is not feasible and that the fees set out herein represent a reasonable estimate of such costs.

ARTICLE 3 CONDITIONS PRECEDENT TO DRAWDOWN

3.1 Conditions for All Drawdowns

On or before each Drawdown hereunder, the following conditions shall be satisfied:

- (a) the Lender shall have received a proper and timely Drawdown Notice from the Borrower requesting the Drawdown;
- (b) the representations and warranties set forth in Section 9.1 shall be true and accurate in all respects on and as of the date of the requested Drawdown;
- (c) no Default or Event of Default shall have occurred and be continuing on and as of the date of the requested Drawdown nor shall the Drawdown result in the occurrence of a Default or Event of Default;
- (d) after giving effect to the proposed Drawdown, the Outstanding Principal of all Loans outstanding under the applicable Credit Facility shall not exceed the maximum amount of such Credit Facility; and
- (e) the Borrower shall not be in receipt of a Demand for Payment

- 27 -

3.2 Conditions for Closing

In addition to the conditions set forth in Section 3.1, the effectiveness of this Agreement and the obligation of the Lender to advance the Drawdowns hereunder are subject to the satisfaction of the following additional conditions:

- (f) the Lender, Borrower and Powell shall have fully executed and delivered this Agreement, in form and substance satisfactory to the Lender (acting reasonably);
- (g) INTENTIONALLY DELETED
- (h) all reasonable fees and expenses previously agreed in writing between the Borrower and the Lender shall be paid by the Borrower to the Lender (including the payment of all of the fees, charges and expenses of Lender's legal counsel);
- (i) the Borrower, Powell and each corporate Subsidiary of the Borrower which is executing and delivering Documents shall have delivered to the Lender a current certificate of status, compliance or good standing, as the case may be, in respect of its jurisdiction of incorporation and certified copies of its constating documents, by-laws, shareholders agreement (if any) and the resolutions authorizing the Documents to which it is a party and transactions hereunder and an Officer's Certificate as to the incumbency of the officers of Powell, the Borrower or the Subsidiary, as the case may be, signing the Documents to which it is a party;
- (j) INTENTIONALLY DELETED
- (k) the Borrower shall have delivered to the Lender an Officer's Certificate detailing ownership structure of the Borrower and its Subsidiaries as of the Closing Date, which certificate shall be in form and substance satisfactory to the Lender (acting reasonably);
- (1) the Security and other Documents shall have been fully executed and delivered, each in form and substance satisfactory to the Lender (acting reasonably), and all registrations, filings and recordings necessary or desirable (as determined by the Lender's legal counsel, acting reasonably) in connection with the Security shall have been made and completed,
- (m) the Lender shall have received (i) a legal opinion from legal counsel to each of Powell, the Borrower and its Subsidiaries in each applicable jurisdiction and (ii) a legal opinion from Lender's legal counsel in each applicable jurisdiction, each in form and substance as may be required by the Lender;
- (n) the Lender shall have received current certificates of insurance evidencing that the Borrower and the Subsidiaries maintain insurance in amounts, terms and coverage in accordance with prudent industry practices and to the extent available on commercially reasonable terms, as required by the Lender, acting reasonably;

- 28 -

(o) <u>INTENTIONALLY DELETED</u>

- (p) the Lender shall have received from Powell an executed Compliance Certificate, *inter alia*, evidencing compliance with the financial covenants set forth in Section 10.3;
- (q) as at the date of such Drawdown, no material adverse change in the business, affairs, assets, properties, operations, or condition, financial or otherwise, of Powell or of the Borrower and its Subsidiaries taken as a whole, shall have occurred since December 31, 2011;

(r) INTENTIONALLY DELETED

(s) the Lender shall have received company prepared year end consolidated financial statements of Powell, which are to the satisfaction of the Lender;

(t) INTENTIONALLY DELETED

- (u) the Lender shall have received the fiscal 2012 consolidated cash flow projections for Powell, which shall include a pro-forma balance sheet, income statement and cash flow statement, and a capital expenditure budget for each quarter during such fiscal year, and shall otherwise be in a form and substance satisfactory to the Lender, acting reasonably; and
- (v) INTENTIONALLY DELETED
- (w) the Lender shall have received the corporate organizational chart for Powell and all of its Subsidiaries, including the Borrower;
- (x) INTENTIONALLY DELETED
- (y) INTENTIONALLY DELETED
- (z) the Lender shall have received such other certificates, instruments, documents and information as the Lender may reasonably request, including but not limited to any internally or independently prepared environmental assessment reports for the Borrower and its Subsidiaries.

3.3 Waiver

The conditions set forth in Sections 3.1 and 3.2 are inserted for the sole benefit of the Lender and may be waived by the Lender, in whole or in part (with or without terms or conditions).

ARTICLE 4 EVIDENCE OF DRAWDOWNS

4.1 Account of Record

The Lender shall open and maintain books of account evidencing all Loans and all other amounts owing by the Borrower to the Lender hereunder. The Lender shall enter in the foregoing accounts details of all amounts from time to time owing, paid or repaid by the Borrower hereunder. The information entered in the foregoing accounts shall, absent manifest error, constitute *prima facie* evidence of the obligations of the Borrower to the Lender hereunder with respect to all Loans and all other amounts owing by the Borrower to the Lender hereunder. After a request by the Borrower, the Lender shall promptly advise the Borrower of such entries made in the Lender's books of account.

- 29 -

ARTICLE 5 PAYMENTS OF INTEREST AND FEES

5.1 Interest on Canadian Prime Rate Loans

The Borrower shall pay interest on each Canadian Prime Rate Loan owing by it under the Revolving Facility during each Interest Period applicable thereto in Canadian Dollars at a rate per annum equal to the Canadian Prime Rate in effect from time to time during such Interest Period plus the Applicable Pricing Rate. Each determination by the Lender of the Canadian Prime Rate applicable from time to time during an Interest Period shall, in the absence of manifest error, be *prima facie* evidence thereof. Such interest shall accrue daily and shall be payable in arrears on each Interest Payment Date for such Loan for the period from and including the Drawdown Date or the preceding Conversion Date or Interest Payment Date, as the case may be, for such Loan to and including the day preceding such Interest Payment Date and shall be calculated on the principal amount of the Canadian Prime Rate Loan outstanding during such period and on the basis of the actual number of days elapsed in a year of 365 days. Changes in the Canadian Prime Rate shall cause an immediate adjustment of the interest rate applicable to such Loans without the necessity of any notice to the Borrower.

5.2 Interest on U.S. Base Rate Loans

The Borrower shall pay interest on each U.S. Base Rate Loan owing by it under the Operating Facility during each Interest Period applicable thereto in United States Dollars at a rate per annum equal to the U.S. Base Rate in effect from time to time during such Interest Period plus the Applicable Pricing Rate. Each determination by the Lender of the U.S. Base Rate applicable from time to time during an Interest Period shall, in the absence of manifest error, be *prima facie* evidence thereof. Such interest shall accrue daily and shall be payable monthly in arrears on each Interest Payment Date for such Loan for the period from and including the Drawdown Date or the preceding Conversion Date or Interest Payment Date, as the case may be, for such Loan to and including the day preceding such Interest Payment Date and shall be calculated on the principal amount of the U.S. Base Rate shall cause an immediate adjustment of the interest rate applicable to such Loans without the necessity of any notice to the Borrower.

5.3 Standby Fees for Revolving Facility

- (a) The Borrower shall pay to the Lender a standby fee in Canadian Dollars calculated at a rate per annum equal to the Applicable Pricing Rate (based on a year of 365 days) for standby fees on the amount, if any, for each day by which the amount of the Outstanding Principal owing to the Lender under the Revolving Facility is less than the Revolving Facility Commitment. Fees determined in accordance with this Section shall be payable by the Borrower in accordance with Section 5.3(b) until the earlier of cancellation in full of the undrawn portion of the Revolving Facility and the Maturity Date under the Revolving Facility.
- (b) The standby fees referred to in Section 5.3(a) shall accrue daily and be payable monthly in arrears on each Interest Payment Date applicable to Canadian Prime Rate Loans for the period from the date hereof or the preceding Interest Payment Date, as the case may be, to and including the day before such Interest Payment Date.
- (c) In order to calculate the daily Outstanding Principal under this Section 5.3 for any day in a calendar month, the Lender shall convert any U.S. Base Rate Loans into the Equivalent Amount thereof in Canadian dollars.

- 30 -

5.4 Stamping Fees

Upon the acceptance by the Lender of a Bankers' Acceptance, the Borrower shall pay to the Lender a stamping fee in Canadian Dollars equal to the Applicable Pricing Rate for stamping fees for Bankers' Acceptances calculated on the principal amount at maturity of such Bankers' Acceptance and for the period of time from and including the date of acceptance to but excluding the maturity date of such Bankers' Acceptance, and calculated on the basis of the number of days elapsed in a year of 365 days.

5.5 Fees Relating to Letters of Credit

- (a) The Borrower shall pay to the Lender in respect of Letters of Credit issued hereunder, an LC Fee payable annually on the date of issuance, calculated at a rate per annum equal to the Lender's Applicable Pricing Rate and on the average daily amount of such Letter of Credit for the number of days such Letter of Credit was outstanding for the period from and including the date of issuance or the date of the immediately preceding determination of the LC Fee (as the case may be) to but excluding that date of determination, in each case, in a year of 365 days.
- (b) In addition, with respect to all Letters of Credit, the Borrower shall from time to time pay to the Lender its usual and customary fees and charges (at the then prevailing rates) for the amendment, delivery and administration of letters of credit such as the Letters of Credit and shall pay and reimburse the Lender for any reasonable out-of-pocket costs and expenses incurred in connection with any Letter of Credit, including in connection with any payment thereunder.

5.6 Interest Act (Canada)

Whenever a rate of interest hereunder is calculated on the basis of a year (the **"deemed year**") which contains fewer days than the actual number of days in the calendar year of calculation, such rate of interest shall be expressed as a yearly rate for purposes of the *Interest Act* (Canada) by multiplying such rate of interest by the actual number of days in the calendar year of calculation and dividing it by the number of days in the deemed year.

5.7 Nominal Rates; No Deemed Reinvestment

The principle of deemed reinvestment of interest shall not apply to any interest calculation under this Agreement; all interest payments to be made hereunder shall be paid without allowance or deduction for deemed reinvestment or otherwise, before and after maturity, default and judgment. The rates of interest specified in this Agreement are intended to be nominal rates and not effective rates. Interest calculated hereunder shall be calculated using the nominal rate method and not the effective rate method of calculation.

- 31 -

5.8 Interest on Overdue Amounts

Notwithstanding any other provision hereof, in the event that any amount due hereunder (including, without limitation, any interest payment) is not paid when due (whether by acceleration or otherwise), the Borrower shall pay interest on such unpaid amount (including, without limitation, interest on interest), if and to the fullest extent permitted by Applicable Law, from the date that such amount is due until the date that such amount is paid in full (but excluding the date of such payment if the payment is received for value at the required place of payment on the date of such payment), and such interest shall accrue daily, be calculated and compounded monthly and be payable on demand, after as well as before maturity, default and judgment, at a rate per annum that is equal to (i) in respect of amounts due in Canadian Dollars, the rate of interest then payable on Canadian Prime Rate Loans plus 3.0% per annum and (ii) in respect of amounts due in United States Dollars, the rate of interest then payable on U.S. Base Rate Loans plus 3.0% per annum.

5.9 <u>Waiver</u>

To the extent permitted by Applicable Law, the covenant of the Borrower to pay interest at the rates provided herein shall not merge in any judgment relating to any obligation of the Borrower to the Lender and any provision of the *Interest Act* (Canada) or *Judgment Interest Act* (Alberta) which restricts any rate of interest set forth herein shall be inapplicable to this Agreement and is hereby waived by the Borrower.

5.10 Maximum Rate Permitted by Law

No interest or fee to be paid hereunder shall be paid at a rate exceeding the maximum rate permitted by Applicable Law. In the event that such interest or fee exceeds such maximum rate, such interest or fees shall be reduced or refunded, as the case may be, so as to be payable at the highest rate recoverable under Applicable Law.

ARTICLE 6 BANKERS' ACCEPTANCES

6.1 Bankers Acceptances

The Borrower may give the Lender notice that Bankers' Acceptances will be required under the Revolving Facility pursuant to a Drawdown, Rollover or Conversion.

6.2 <u>Fees</u>

Upon the acceptance by the Lender of a Bankers' Acceptance, the Borrower shall pay to the Lender a fee in Canadian Dollars equal to the Applicable Pricing Rate calculated on the principal amount at maturity of such Bankers' Acceptance and for the period of time from and including the date of acceptance to but excluding the maturity date of such Bankers' Acceptance and calculated on the basis of the number of days elapsed in a year of 365 days.

6.3 Form and Execution of Bankers' Acceptances

The following provisions shall apply to each Bankers' Acceptance hereunder:

(a) the face amount at maturity of each draft drawn by the Borrower to be accepted as a Bankers' Acceptance shall be Cdn. \$100,000 and integral multiples thereof;

- 32 -

- (b) the term to maturity of each draft drawn by the Borrower to be accepted as a Bankers' Acceptance shall, subject to market availability as determined by the Lender, be one (I), two (2) or three (3) months (or such other longer or shorter term as agreed by the Lender), as selected by the Borrower in the relevant Drawdown, Rollover or Conversion Notice, and each Bankers' Acceptance shall be payable and mature on the last day of the Interest Period selected by the Borrower for such Bankers' Acceptance (which, for certainty, pursuant to the definition of "Interest Period" shall be on or prior to the Maturity Date);
- (c) each draft drawn by the Borrower and presented for acceptance by the Lender shall be drawn on the standard form of the Lender in effect at the time; provided, however, that the Lender may use a generic form of Bankers' Acceptance, in a form satisfactory to the Lender, acting reasonably;
- (d) subject to Section 6.3(e) below, Bankers' Acceptances shall be signed by duly authorized officers of the Borrower or, in the alternative, the signatures of such officers may be mechanically reproduced in facsimile thereon and Bankers' Acceptances bearing such facsimile signatures shall be binding on the Borrower as if they had been manually executed and delivered by such officers on behalf of the Borrower; notwithstanding that any person whose manual or facsimile signature appears on any Bankers' Acceptance may no longer be an authorized signatory for the Borrower on the date of issuance of a Bankers' Acceptance, such signature shall nevertheless be valid and sufficient for all purposes as if such authority had remained in force at the time of such issuance and any such Bankers' Acceptance shall be binding on the Borrower; and
- (e) in lieu of signing Bankers' Acceptances in accordance with Section 6.3(d) above, the Borrower may provide a Power of Attorney to the Lender; for so long as a Power of Attorney is in force with respect to the Lender, the Lender shall execute and deliver Bankers' Acceptances on behalf of the Borrower in accordance with the provisions thereof and, for certainty, all references herein to drafts drawn by the Borrower, Bankers' Acceptances executed by the Borrower or similar expressions shall be deemed to include Bankers' Acceptances executed in accordance with a Power of Attorney, unless the context otherwise requires.

6.4 Power of Attorney; Provision of Bankers' Acceptances to Lender

- (a) Unless revoked in accordance herewith, the Borrower hereby appoints the Lender, acting by any authorized signatory of the Lender in question, the attorney of the Borrower:
 - to sign for and on behalf and in the name of the Borrower as drawer, drafts in the Lender's standard form which are depository bills as defined in the *Depository Bills and Notes Act* (Canada) (the "DBNA"), payable to a "clearing house" (as defined in the DBNA) including, without limitation, The Canadian Depository For Securities Limited or its nominee, CDS & Co. (the "clearing house");
 - (ii) for drafts which are not depository bills, to sign for and on behalf and in the name of the Borrower as drawer and to endorse on its behalf, Bankers' Acceptances drawn on the Lender payable to the order of the undersigned or payable to the order of the Lender;

- 33 -

- (iii) to fill in the amount, date and maturity date of such Bankers' Acceptances; and
- (iv) to deposit and/or deliver such Bankers' Acceptances which have been accepted by the Lender,

provided that such acts in each case are to be undertaken by the Lender in question strictly in accordance with instructions given to the Lender by the Borrower as provided in this Section. For certainty, signatures of any authorized signatory of the Lender may be mechanically reproduced in facsimile on Bankers' Acceptances in accordance herewith and such facsimile signatures shall be binding and effective as if they had been manually executed by such authorized signatory of the Lender.

Instructions from the Borrower to the Lender relating to the execution, completion, endorsement, deposit and/or delivery by the Lender on behalf of the Borrower of Bankers' Acceptances which the Borrower wishes to submit to the Lender for acceptance by the Lender shall be communicated by the Borrower in writing to the Lender by delivery to the Lender of Drawdown Notices, Conversion Notices and Rollover Notices, as the case may be, in accordance with this Agreement.

The communication in writing by the Borrower to the Lender of the instructions set out in the Drawdown Notices, Conversion Notices and Rollover Notices referred to above shall constitute (a) the authorization and instruction of the Borrower to the Lender to sign for and on behalf and in the name of the Borrower as drawer the requested Bankers' Acceptances and to complete and/or endorse Bankers' Acceptances in accordance with such information as set out above and (b) the request of the Borrower to the Lender to accept such Bankers' Acceptances and deposit the same with the clearing house or deliver the same, as the case may be, in each case in accordance with this Agreement and such instructions. The Borrower acknowledges that the Lender shall not be obligated to accept any such Bankers' Acceptances except in accordance with the provisions of this Agreement.

The Lender shall be and it is hereby authorized to act on behalf of the Borrower upon and in compliance with instructions communicated to the Lender as provided herein if the Lender reasonably believes such instructions to be genuine.

This Power of Attorney may be revoked by the Borrower with respect to the Lender at any time upon not less than 5 Banking Days' prior written notice served upon the Lender, provided that no such revocation shall reduce, limit or otherwise affect the obligations of the Borrower in respect of any Bankers' Acceptance executed, completed, endorsed, deposited and/or delivered in accordance herewith prior to the time at which such revocation becomes effective.

- (b) Unless the Borrower has provided Powers of Attorney to the Lender, to facilitate Drawdowns, Rollovers or Conversions of Bankers' Acceptances, the Borrower shall, upon execution of this Agreement and thereafter from time to time as required by the Lender, provide to the Lender drafts drawn in blank by the Borrower (pre-endorsed and otherwise in fully negotiable form, if applicable) in quantities sufficient for the Lender to fulfill its obligations hereunder. Any such pre signed drafts which are delivered by the Borrower to the Lender shall be held in safekeeping by the Lender with the same degree of care as if they were the Lender's property, and shall only be dealt with by the Lender in accordance herewith. The Lender shall not be responsible or liable for its failure to make any Drawdown, Rollover or Conversion of Bankers' Acceptances required hereunder if the cause of such failure is, in whole or in part, due to the failure of the Borrower to provide such pre-signed drafts to the Lender on a timely basis.
- (c) By 10:00 a.m. (Edmonton time) on the applicable Drawdown Date, Conversion Date or Rollover Date for which Borrower has given notice hereunder that Bankers' Acceptances have been elected or will be required by Borrower, the Borrower shall (i) either deliver to the Lender, or, if previously delivered, be deemed to have authorized the Lender to complete and accept, or (ii) where the Borrower has previously executed and delivered a Power of Attorney to the Lender, be deemed to have authorized the Lender to sign on behalf of the Borrower, complete and accept, drafts drawn by the Borrower on the Lender in a principal amount at maturity equal to the Bankers' Acceptances specified by the Borrower in the relevant Drawdown Notice, Conversion Notice or Rollover Notice, as the case may be, as notified to the Lender by the Borrower.

- 35 -

6.5 Mechanics of Issuance

- (a) The Lender may at any time and from time to time hold, sell, rediscount or otherwise dispose of any or all Bankers' Acceptances accepted and purchased by it for its own account.
- (b) On each such Drawdown Date, Rollover Date or Conversion Date involving the issuance of Bankers' Acceptances:
 - (i) Lender shall complete and accept, in accordance with the Drawdown Notice, Conversion Notice or Rollover Notice delivered by the Borrower the Bankers' Acceptances to be issued on such date and shall purchase such Bankers' Acceptances for its own account at a purchase price which reflects the BA Rate applicable to such issue; and
 - (ii) in the case of a Drawdown, the Lender shall, for same day value on the Drawdown Date, remit the Discount Proceeds payable by the Lender (net of the acceptance fee payable to the Lender pursuant to Section 6.2) to the Borrower.

6.6 Rollover, Conversion or Payment on Maturity

In anticipation of the maturity of Bankers' Acceptances, the Borrower shall, subject to and in accordance with the requirements hereof, do one or a combination of the following with respect to the aggregate face amount at maturity of all such Bankers' Acceptances:

- (a) (i) deliver to the Lender a Rollover Notice that the Borrower intends to draw and present for acceptance on the maturity date new Bankers' Acceptances in an aggregate face amount up to the aggregate amount of the maturing Bankers' Acceptances and (ii) on the maturity date pay to the Lender an additional amount equal to the difference between the aggregate face amount of the maturing Bankers' Acceptances and the Discount Proceeds of such new Bankers' Acceptances;
- (b) (i) deliver to the Lender a Conversion Notice requesting a Conversion of the maturing Bankers' Acceptances to another type of Loan under the Revolving Facility as the maturing Bankers' Acceptances and (ii) on the maturity date pay to the Lender an amount equal to the difference, if any, between the aggregate face amount of the maturing Bankers' Acceptances and the amount of the Loans into which Conversion is requested; or
- (c) on the maturity date of the maturing Bankers' Acceptances, pay to the Lender an amount equal to the aggregate face amount of such Bankers' Acceptances.

If the Borrower fails to so notify the Lender or make such payments on maturity, the Lender shall effect a Conversion into a Canadian Prime Rate Loan of the entire amount of such maturing Bankers' Acceptances as if a Conversion Notice had been given by the Borrower to the Lender to that effect.

- 36 -

6.7 <u>Restriction on Rollovers and Conversions</u>

Subject to the other provisions hereof, Conversions and Rollovers of Bankers' Acceptances may only occur on the maturity date thereof.

6.8 <u>Rollovers</u>

In order to satisfy the continuing liability of the Borrower to the Lender for the face amount of maturing Bankers' Acceptances accepted by the Lender, the Lender shall receive and retain for its own account the Discount Proceeds of new Bankers' Acceptances issued on a Rollover, and the Borrower shall on the maturity date of the Bankers' Acceptances being rolled over pay to the Lender an amount equal to the difference between the face amount of the maturing Bankers' Acceptances and the Discount Proceeds from the new Bankers' Acceptances, together with the acceptance fees to which the Lender is entitled pursuant to Section 6.2.

6.9 Conversion into Bankers' Acceptances

In respect of Conversions into Bankers' Acceptances, in order to satisfy the continuing liability of the Borrower to the Lender for the amount of the converted Loan, the Lender shall receive and retain for its own account the Discount Proceeds of the Bankers' Acceptances issued upon such Conversion, and the Borrower shall on the Conversion Date pay to the Lender an amount equal to the difference between the principal amount of the converted Loan and the aggregate Discount Proceeds from the Bankers' Acceptances issued on such Conversion, together with the acceptance fees to which the Lender is entitled pursuant to Section 6.2.

6.10 Conversion from Bankers' Acceptances

In order to satisfy the continuing liability of the Borrower to the Lender for an amount equal to the aggregate face amount of the maturing Bankers' Acceptances converted to another type of Loan, the Lender shall record the obligation of the Borrower as a Loan of the type into which such continuing liability has been converted.

ARTICLE 7 LETTERS OF CREDIT

7.1 <u>Availability</u>

Subject to the provisions hereof, the Borrower may require that Letters of Credit be issued under the Revolving Facility by delivering a Drawdown Notice in accordance with Section 2.6. The issuance of Letters of Credit shall constitute a Drawdown hereunder and shall reduce the availability of the Revolving Facility by the aggregate Outstanding Principal of Letters of Credit under the Revolving Facility.

7.2 <u>Currency, Type and Expiry</u>

Letters of Credit issued pursuant hereto shall be denominated in Canadian Dollars and amounts payable thereunder shall be paid in the currency in which the Letter of Credit is denominated. A Letter of Credit issued hereunder shall be issued by the Lender; <u>provided</u> that the Lender shall have no obligation to issue any Letter of Credit unless and until it has received such ancillary documents, including applications and indemnities, as the Lender normally requires for similar transactions. Letters of Credit shall be in a form satisfactory to the Lender, acting reasonably, and shall have an expiration date not in excess of one year from the date of issue and, in any event, not later than the Maturity Date.

- 37 -

7.3 <u>Reimbursement or Conversion on Presentation</u>

Upon presentation of a Letter of Credit and payment thereunder by the Lender, the Borrower shall (at its option) either forthwith pay to and reimburse the Lender all amounts paid pursuant to such Letter of Credit or, failing such payment, the Borrower shall be deemed to have effected a Conversion of such Letter of Credit into a Canadian Prime Rate Loan.

7.4 Uniform Customs and Practice

The Uniform Customs and Practice for Loan Documentary Credits as most recently published by the International Chamber of Commerce (the "**Uniform Customs**") shall in all respects apply to each Letter of Credit unless expressly provided to the contrary therein and shall be deemed for such purpose to be a part of this Agreement as if fully incorporated herein. In the event of any conflict or inconsistency between the Uniform Customs and the governing law of this Agreement, the Uniform Customs shall, to the extent permitted by Applicable Law, prevail to the extent necessary to remove the conflict or inconsistency.

ARTICLE 8 PLACE AND APPLICATIONS OF PAYMENTS

8.1 Place of Payment of Principal, Interest and Fees; Payments to Lender

All payments of principal, interest, fees and other amounts to be made by the Borrower to the Lender pursuant to this Agreement shall be made to the Lender in the currency in which the Loan is outstanding for value on the day such amount is due, and if such day is not a Banking Day on the Banking Day next following, by deposit or transfer thereof to the Lender's Account or at such other place as the Lender may specify in writing from time to time. Any payment delivered or made to the Lender by 1:00 p.m. local time at the place where such payment is to be made shall be credited as of that day, but if made afterwards shall be credited as of the next Banking Day thereafter.

8.2 Absolute and Unconditional Obligation to Pay

The obligation of the Borrower to make all payments pursuant to this Agreement and the Security Documents shall be absolute and unconditional and shall not be limited or affected by any circumstance, including, without limitation:

- (a) any set-off, compensation, counter-claim, recoupment, defense or other right which the Borrower may have against the Lender or anyone else for any reason whatsoever; or
- (b) any insolvency, bankruptcy, reorganization or similar proceedings by or against the Borrower.

8.3 <u>Funds</u>

Each amount advanced, disbursed or paid hereunder shall be advanced, disbursed or paid, as the case may be, in such form of funds as may from time to time be customarily used in Edmonton, Alberta in the settlement of banking transactions similar to the banking transactions

required to give effect to the provisions of this Agreement on the day such advance, disbursement or payment is to be made (for certainty, each such amount advanced, disbursed or paid hereunder shall be advanced, disbursed or paid, as the case may be, in immediately available funds to the extent possible).

8.4 Application of Payments

Except as otherwise agreed in writing by the Lender, if any Event of Default shall occur and be continuing, all payments made by the Borrower to the Lender shall be applied in the following order:

- (a) to amounts due hereunder as fees other than acceptance fees for Bankers' Acceptances;
- (b) to amounts due hereunder as costs and expenses;
- (c) to amounts due hereunder as default interest;
- (d) to amounts due hereunder as interest and acceptance fees for Bankers' Acceptances; and
- (e) to amounts due hereunder as principal (including reimbursement obligations in respect of Bankers' Acceptances).

8.5 Payments Clear of Taxes

(a) Any and all payments by the Borrower to the Lender hereunder shall be made free and clear of, and without deduction or withholding for or on account of, any and all present or future Taxes and all liabilities with respect thereto imposed, levied, collected, withheld or assessed by any Governmental Authority or under the laws of any international tax authority imposed on the Lender, or by or on behalf of the foregoing (and, for greater certainty, nothing in this Section 8.5(a) shall make the Borrower liable for any taxes imposed on or measured by the Lender's overall net income or franchise taxes imposed on it (in lieu of net income taxes) by the jurisdiction (or any political subdivision thereof) under the laws of which the Lender is organized, in which its applicable lending office is maintained or in which its principal office is located). In addition, the Borrower agrees to pay any present or future stamp, transfer, registration, excise, issues, documentary or other taxes, charges or similar levies which arise from any payment made under this Agreement or the Loans or in respect of the execution, delivery or registration or the compliance with this Agreement or the other Documents contemplated hereunder other than taxes imposed on or measured by the Lender's overall net income and franchise taxes imposed on it (in lieu of net income taxes) by the jurisdiction (or any political subdivision thereof) under the laws of which the Lender is organized, in which its applicable lending office is maintained or in which its principal office is located. The Borrower shall indemnify and hold harmless the Lender for the full amount of any Taxes or other amounts paid or payable by the Lender and any liability (including penalties, interest, additions to tax and reasonable out-of-pocket expenses) resulting therefrom or with respect thereto which arise from any payment made under or pursuant to this Agreement or the Loans or in respect of the execution, delivery or registration of, or compliance with, this Agreement o

- 39 -

than taxes imposed on or measured by the Lender's overall net income and franchise taxes imposed on it (in lieu of net income taxes) by the jurisdiction (or any political subdivision thereof) under the laws of which the Lender is organized, in which its applicable lending office is maintained or in which its principal office is located.

(b) If the Borrower shall be required by law to deduct or withhold any amount from any payment or other amount required to be paid to the Lender hereunder, or if any liability therefor shall be imposed or shall arise from or in respect of any sum payable hereunder, then the sum payable to the Lender hereunder shall be increased as may be necessary so that after making all required deductions, withholdings, and additional income tax payments attributable thereto (including deductions, withholdings or income tax payable for additional sums payable under this provision) the Lender receives an amount equal to the amount it would have received had no such deductions or withholdings been made or if such additional taxes had not been imposed; in addition, the Borrower shall pay the full amount deducted or withheld for such liabilities to the relevant taxation authority or other authority in accordance with Applicable Law, such payment to be made (if the liability is imposed on the Lender, the Borrower shall deliver to the Lender. If the liability is imposed on the Lender, the Borrower shall deliver to the Lender evidence satisfactory to the Lender, acting reasonably, of the payment to the relevant taxation authority or other authority of the full amount deducted or withheld.

8.6 <u>Set Off</u>

- (a) In addition to any rights now or hereafter granted under Applicable Law and not by way of limitation of any such rights, upon the occurrence of an Event of Default which remains unremedied (whether or not the Loans have been accelerated hereunder), the Lender shall have the right (and are hereby authorized by the Borrower) at any time and from time to time to combine all or any of the Borrower's accounts with the Lender and to set off and to appropriate and to apply any and all deposits (general or special, term or demand) including, but not limited to, indebtedness evidenced by certificates of deposit whether matured or unmatured, and any other indebtedness at any time held by the Borrower or owing by the Lender to or for the credit or account of the Borrower against and towards the satisfaction of any Obligations owing by the Borrower, and may do so notwithstanding that the balances of such accounts and the liabilities are expressed in different currencies, and the Lender is hereby authorized to effect any necessary currency conversions at the noon spot rate of exchange announced by the Bank of Canada on the Banking Day before the day of conversion.
- (b) The Lender shall notify the Borrower of any such set off from the Borrower's accounts within a reasonable period of time thereafter, although the Lender shall not be liable to the Borrower for its failure to so notify.

- 40 -

8.7 INTENTIONALLY DELETED

ARTICLE 9 REPRESENTATIONS AND WARRANTIES

9.1 <u>Representations and Warranties</u>

Each of Powell and the Borrower represents and warrants as follows to the Lender and acknowledges and confirms that the Lender is relying upon such representations and warranties:

(a) <u>Existence - The Borrower</u>

The Borrower is a corporation duly incorporated and validly existing under the laws of Canada and is duly registered in all other jurisdictions where the nature of its property or character of its business requires registration, except for jurisdictions where the failure to be so registered or qualified would not have a Material Adverse Effect, and has all necessary power and authority to own its properties and carry on its business as presently carried on or as contemplated by the Documents.

(b) <u>Existence – Powell</u>

Powell is a corporation duly incorporated and validly existing under the laws of the State of Delaware and is duly registered in all other jurisdiction where the nature of its property or character of its business requires registration, except for jurisdictions where the failure to be so registered or qualified would not have a Material Adverse Effect, and has all necessary power and authority to own its properties and carry on its business as presently carried on.

(c) Existence and Good Standing -Corporations

Each of the Subsidiaries is a corporation incorporated and validly existing and in good standing under the laws of Canada; each is duly registered in all other jurisdictions where the nature of its property or character of its business requires registration, except for jurisdictions where the failure to be so registered or qualified would not have a Material Adverse Effect, and has all necessary power and authority to own its properties and carry on its business as presently carried on or as contemplated by the Documents.

(d) Existence -Partnerships and Trusts

Each Subsidiary that is a partnership or a trust is validly existing under the laws of the Province of Alberta and is duly registered in all other jurisdictions where the nature of its property or character of its business requires registration, except for jurisdictions where the failure to be so registered or qualified would not have a Material Adverse Effect, and has all necessary power and authority to own its properties and carry on its business as presently carried on or as contemplated by the Documents.

- 41 -

(e) <u>Authority</u>

Each of the Borrower, Powell and each Subsidiary have full power, legal right and authority to enter into the Documents to which it is a party and do all such acts and things as are required by such Documents to be done, observed or performed, in accordance with the terms thereof.

(f) Valid Authorization and Execution

Each of the Borrower, Powell and each Subsidiary has taken all necessary corporate, partnership and other action (as applicable) of its directors, shareholders, partners, trustees and other persons (as applicable) to authorize the execution, delivery and performance of the Documents to which it is a party and to observe and perform the provisions thereof in accordance with the terms therein contained.

(g) Validity of Agreement Non-Conflict

None of the authorization, execution or delivery of this Agreement or performance of any obligation pursuant thereto requires or will require, pursuant to Applicable Law now in effect, any approval or consent of any Governmental Authority having jurisdiction (except such as has already been obtained and are in full force and effect) nor is in conflict with or contravention of (i) the Borrower's, (ii) Powell's or (iii) any Subsidiary's articles, by laws or other constating documents or any resolutions of directors or shareholders or the provisions of its partnership agreement or declaration of trust or trust indenture (as applicable) or (iv) the provisions of any other indenture, instrument, undertaking or other agreement to which Powell, the Borrower or any of its Subsidiaries is a party or by which they or their properties or assets are bound, the contravention of which would have or would reasonably be expected to have a Material Adverse Effect. The Documents when executed and delivered will constitute valid and legally binding obligations of each of Powell, the Borrower and each of its Subsidiaries which is a party thereto enforceable against each such party in accordance with their respective terms, subject to applicable bankruptcy, insolvency and other laws of general application limiting the enforceability of creditors' rights and to the fact that equitable remedies are only available in the discretion of the court.

(h) Ownership of Property

Each of the Borrower, Powell, and each Subsidiary has good and marketable title to its material property, subject to Permitted Encumbrances which, individually and in the aggregate, do not materially affect their respective rights of ownership to such property, the value thereof or their right or ability to utilize the same in the conduct of their business and affairs.

(i) **INTENTIONALLY DELETED**

(j) <u>Debt</u>

Neither the Borrower nor any Subsidiary has created, incurred, assumed, suffered to exist, or entered into any contract, instrument or undertaking pursuant to which, the Borrower or any Subsidiary is now or may hereafter become liable for Debt other than Permitted Debt, in the aggregate, in excess of \$1,000,000.00.

- 42 -

(k) Encumbrances

Neither the Borrower nor any Subsidiary has created, incurred, assumed, suffered to exist, or entered into any contract, instrument or undertaking pursuant to which, any person may have or be entitled to any Security Interest on or in respect of its property and assets or any part thereof except for Permitted Encumbrances.

(l) <u>No Material Adverse Effect</u>

No event or circumstance has occurred or is continuing which has had or would reasonably be expected to have a Material Adverse Effect.

(m) <u>No Omissions</u>

The Borrower has made available to the Lender all material information necessary to make any representations, warranties and statements contained in this Agreement not misleading in any material respect in light of the circumstances in which they are given.

(n) (n) Non-Default

No Default or Event of Default has occurred or is continuing or would occur following any Drawdown hereunder.

(o) <u>Financial Condition</u>

The audited and unaudited consolidated financial statements of Powell delivered to the Lender pursuant hereto present fairly, in all material respects, the consolidated financial condition of Powell as at the date thereof and the results of the consolidated operations thereof for the fiscal year or fiscal quarter (as applicable) then ending, all in accordance with GAAP consistently applied.

(p) Information Provided

All information, materials and documents, including all cash flow projections, economic models, capital and operating budgets and other information and data:

(i) prepared and provided to the Lender by Powell, the Borrower or any Subsidiary in respect of the transactions contemplated by this Agreement, or as required by the terms of this Agreement, were, in the case of financial projections, prepared in good faith based upon reasonable assumptions at the date of preparation, and, in all other cases, true, complete and correct in all material respects as of the respective dates thereof; and



(ii) to the extent prepared by persons other than Powell, the Borrower or a Subsidiary and provided to the Lender by or on behalf of Powell, the Borrower or any Subsidiary in respect of the transactions contemplated by this Agreement, or as required by the terms of this Agreement, were, to the best of the knowledge of Powell and the Borrower after due inquiry, in the case of financial projections, prepared in good faith based upon reasonable assumptions at the date of preparation, and, in all other cases, true, complete and correct in all material respects as of the respective dates thereof.

(q) <u>Absence of Litigation</u>

There are no actions, suits or proceedings pending or, to the knowledge of Powell and the Borrower, threatened against or affecting Powell, the Borrower or any of its Subsidiaries, their property or any of their undertakings and assets, at law, in equity or before any arbitrator or before or by any Governmental Authority having jurisdiction in the premises in respect of which there is a reasonable possibility of a determination adverse to Powell, the Borrower or any Subsidiary and which, if determined adversely, would have or would reasonably be expected to have a Material Adverse Effect.

(r) Compliance with Applicable Laws, Court Orders and Agreements

Powell, the Borrower and each of its Subsidiaries and their respective property, businesses and operations are in compliance with all Applicable Laws (including, without limitation, all applicable Environmental Laws), all applicable directives, judgments, decrees, injunctions and orders rendered by any Governmental Authority or court of competent jurisdiction, its articles, by laws and other constating documents, all agreements or instruments to which it is a party or by which its property or assets are bound, and any employee benefit plans, except to the extent that failure to so comply would not have and would not reasonably be expected to have a Material Adverse Effect.

(s) <u>Required Permits in Effect</u>

All Required Permits are in full force and effect, except to the extent that the failure to have or maintain the same in full force and effect would not, when taken in the aggregate, have or reasonably be expected to have a Material Adverse Effect.

(t) <u>Remittances Up to Date</u>

All of the material remittances required to be made by Powell, the Borrower and its Subsidiaries to Governmental Authorities which are due and payable have been made, are currently up to date and there are no outstanding arrears, other than those which are being contested by Permitted Contest.

(u) <u>Environmental</u>

(i) To the best of the knowledge and belief of Powell and the Borrower, after due inquiry, Powell, the Borrower, its Subsidiaries and their respective properties, assets and undertakings taken as a whole comply in all respects and the businesses, activities and operations of same and the use of such properties, assets and undertakings and the processes and undertakings performed thereon comply in all respects

- 44 -

with all Environmental Laws except to the extent that failure to so comply would not have and would not reasonably be expected to have a Material Adverse Effect; further, neither Powell nor the Borrower knows, and has no reasonable grounds to know, of any facts which result in or constitute or are likely to give rise to non-compliance with any Environmental Laws, which facts or non-compliance have or would reasonably be expected to have a Material Adverse Effect.

(ii) Neither Powell nor the Borrower have received written notice and, except as previously disclosed to the Lender in writing, has no knowledge after due inquiry, of any facts which could give rise to any notice of non-compliance with any Environmental Laws, which non-compliance has or would reasonably be expected to have a Material Adverse Effect and has not received any notice that Powell, the Borrower or any of its Subsidiaries is a potentially responsible party for a federal, provincial, regional, municipal or local clean up or corrective action in connection with their respective properties, assets and undertakings where such clean up or corrective action has or would reasonably be expected to have a Material Adverse Effect.

(v) <u>(v) Taxes</u>

Each of Powell, the Borrower and each of its Subsidiaries has duly filed on a timely basis all tax returns required to be filed and have paid all material Taxes which are due and payable, and have paid all material assessments and reassessments, and all other material Taxes, governmental charges, governmental royalties, penalties, interest and fines claimed against them, other than those which are being contested by them by Permitted Contest; they have made adequate provision for, and all required installment payments have been made in respect of, Taxes payable for the current period for which returns are not yet required to be filed; there are no agreements, waivers or other arrangements providing for an extension of time with respect to the filing of any tax return by them or the payment of any Taxes; there are no actions or proceedings being taken by any taxation authority in any jurisdictions where Powell, the Borrower or any Subsidiary carries on business to enforce the payment of any Taxes by them other than those which are being contested by them by Permitted Contest.

(w) Subsidiaries and Assets; Documents of Title

The Disclosure Schedule (as it may be supplemented from time to time by the Borrower) sets forth as at the date hereof:

- the corporate organizational chart of Powell and its Subsidiaries and the names and jurisdictions of incorporation of each of Powell, the Borrower and Subsidiaries, their respective share capital or other ownership interests and the outstanding shares or ownership interests, and any trade names used by such entities;
- (ii) the jurisdictions in which Powell, the Borrower and the Subsidiaries conduct business;

- 45 -

- (iii) the location of Powell's, the Borrower's and the Subsidiaries' respective places of business, locations where inventory or other assets are held and the locations of their respective chief executive offices;
- (iv) all fee interests in any real property, and all material leases, subleases or assignments of leases (together with all amendments, modifications, supplements, renewals or extensions of any thereof) in real property, regardless of whether Powell, the Borrower or any Subsidiary is the landlord or tenant (whether directly or as an assignee or successor in interest) under such lease, sublease or assignment;
- (v) all Material Agreements (of the Borrower and its Subsidiaries only); and
- (vi) all trademarks, patents and other material intellectual property.

No negotiable documents of title, bills of lading or warehouse receipts have been issued in respect of the Borrower's or any Subsidiary's inventory or assets.

(x) (x) Intellectual Property

Each of the Borrower and its Subsidiaries has or has the legal right to use all Intellectual Property necessary for the operation and conduct of their business, affairs, operations and processes and, to the best of their knowledge and belief, no person has asserted any claim or taken any step or proceedings to prohibit or limit the use of such Intellectual Property by the Borrower and its Subsidiaries.

(y) <u>Performance of Material Agreements</u>

Neither the Borrower nor any of its Subsidiaries is in default in the performance, observance or fulfillment of any of the obligations, covenants or conditions contained in any of its Material Agreements, and no condition exists that, with the giving of notice or the lapse of time or both, would constitute such a default, except in either case where the consequences, direct or indirect, of such default or defaults, if any, could not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect **("Material Defaults")**. All Material Agreements are in full force and effect and no dispute which could reasonably be expected to have a Material Adverse Effect currently exist thereunder, and neither Powell nor the Borrower has knowledge of any Material Defaults by any third party currently existing thereunder.

(z) Books and Records

All books and records of Powell, the Borrower and the Subsidiaries have been fully, properly and accurately kept and completed in all material respects and there are no material inaccuracies or material discrepancies of any kind contained or reflected therein.

- 46 -

(aa) Employee Matters

Each of Powell, the Borrower and its Subsidiaries has made full payment when due of all required contributions to any employee benefit plan except where failure to do so, individually or in the aggregate, could not reasonably be expected to have a Material Adverse Effect. There is no strike or work stoppage in existence or threatened involving Powell, the Borrower or any of its Subsidiaries that could, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect.

9.2 Deemed Repetition

On the date of delivery by the Borrower of a Drawdown Notice, Conversion Notice or Rollover Notice to the Lender, and again on the date of any Drawdown, Conversion or Rollover made by the Borrower pursuant thereto, as well as on each anniversary of the Closing Date:

- (a) Each of the representations and warranties contained in Section 9.1 shall be deemed to be repeated; and
- (b) Each of the Borrower and Powell shall be deemed to have represented to the Lender that no event has occurred and remains outstanding which would constitute a Default or an Event of Default nor will any such event occur as a result of the aforementioned Drawdown, Conversion or Rollover.

9.3 Other Documents

All representations, warranties, certifications and statements of Powell, the Borrower or any Subsidiary contained in any other Document delivered pursuant hereto or thereto shall be deemed to constitute representations and warranties made by Powell and the Borrower to the Lender under Section 9.1 of this Agreement.

9.4 Effective Time of Repetition

All representations and warranties, when repeated or deemed to be repeated hereunder, shall be construed with reference to the facts and circumstances existing at the time of repetition, unless they are stated herein to be made as at the date hereof or as at another date.

9.5 Nature of Representations and Warranties

The representations and warranties set out in this Agreement or deemed to be made pursuant hereto shall survive the execution and delivery of this Agreement and the making of each Drawdown, notwithstanding any investigations or examinations which may be made by the Lender or Lender's legal counsel. Such representations and warranties shall survive until this Agreement has been terminated, provided that the representations and warranties relating to environmental matters shall survive the termination of this Agreement.

ARTICLE 10 GENERAL COVENANTS

10.1 Affirmative Covenants of the Borrower

So long as any Obligation is outstanding or any Credit Facility is available hereunder, the Borrower covenants and agrees with the Lender that, unless the Lender otherwise consent in writing:



(a) <u>Punctual Payment and Performance</u>

It shall duly and punctually pay the principal of all Loans, all interest thereon and all fees and other amounts required to be paid by the Borrower hereunder in the manner specified hereunder and the Borrower shall perform and observe all of its obligations under this Agreement and under any other Document to which it is a party.

(b) Books and Records

It shall keep proper books of record and account in which complete and correct entries will be made of its transactions in accordance with GAAP.

(c) <u>Maintenance and Operation</u>

It shall do or cause to be done, and will cause each Subsidiary to do or cause to be done, all things necessary or required to have all its properties, assets and operations owned, operated and maintained in accordance with diligent and prudent industry practice and Applicable Laws except to the extent that the failure to do or cause to be done the same would not have and would not reasonably be expected to have a Material Adverse Effect, and at all times cause the same to be owned, operated, maintained and used in compliance with all terms of any applicable insurance policy to the extent necessary to ensure that coverage under any such policy cannot be denied by the insurers thereunder.

(d) Maintain Existence; Compliance with Legislation Generally; Required Permits

Except as otherwise permitted by Section 10.2(d), the Borrower shall, and shall cause each of its Subsidiaries, to preserve and maintain its corporate, partnership or trust existence (as the case may be) as a corporation, partnership or trust existing under the laws of Canada or any province thereof. The Borrower shall do or cause to be done, and shall cause its Subsidiaries to do or cause to be done, all acts necessary or desirable to comply with all Applicable Laws, except where such failure to comply does not and would not reasonably be expected to have a Material Adverse Effect, and to preserve and keep in full force and effect all Required Permits and all other franchises, licences, rights, privileges, permits and Governmental Authorizations necessary to enable the Borrower and each of its Subsidiaries to operate and conduct their respective businesses in accordance with prudent industry practice, except to the extent that the failure to have any of the same does not and would not reasonably be expected to have a Material Adverse Effect.

(e) Budgets, Financial Statements, Engineering Reports and Other Information

The Borrower shall deliver to the Lender:

(i) <u>Annual Financials</u>—as soon as available and, in any event, within 120 days after the end of each of its fiscal years, copies of Powell's audited annual financial statements on a consolidated basis with a comparison to the forecast for such year, and, if requested, copies of the unaudited annual financial statements on an unconsolidated basis of the Borrower and each Subsidiary, each consisting of a balance sheet, statement of

- 48 -

income, statement of cash flows and statement of shareholders' equity for each such year, together with the notes thereto in the case of the audited annual financial statements, all prepared in accordance with GAAP consistently applied, together with a report and unqualified opinion of Powell's auditors thereon in the case of audited annual financial statements of Powell and including any management letters provided by the auditors in connection with such audit;

- (ii) <u>Quarterly Financials</u>—as soon as available and, in any event within 60 days after the end of each of the first three (3) fiscal quarters, copies of Powell's unaudited financial statements on a consolidated basis with a comparison to the approved forecast for such fiscal quarter, in each case consisting of a balance sheet, statement of income and statement of cash flows for each such period all in reasonable detail and stating in comparative form the figures for the corresponding date and fiscal quarter in the previous fiscal year, all prepared in accordance with GAAP consistently applied;
- (iii) <u>Compliance Certificates</u>—concurrently with furnishing the annual financial statements pursuant to Section 10.1 (e)(i) and the quarterly financial statements pursuant to Section 10.1 (e)(ii) after each Quarter End (except for the fourth fiscal quarter), a Compliance Certificate signed by anyone of the president, chief financial officer, vice president -finance or treasurer of the Borrower and stating that, *inter alia*, the representations and warranties in Section 9.1 are true and accurate in all respects (or, if applicable, specifying those representations and warranties that are not), that no Default or Event of Default has occurred and is continuing (or, if applicable, specifying those defaults or events notified in accordance with Section 10.1(h) below) and demonstrating compliance with all covenants of the Borrower in Sections 10.1 and 10.2, and all covenants of the Borrower and Powell in Section 10.3;
- (iv) <u>Financial Statements of the Borrower</u> as soon as available and, in any event within sixty (60) days after the end of each of the first three (3) fiscal quarters and after the end of each of its fiscal years, copies of Borrower's unaudited financial statements on a consolidated basis for each of the first three (3) fiscal quarters and for the fiscal year, respectively (in-house statements are acceptable), with a comparison to the approved forecast for such fiscal quarters and fiscal year, and in each case consisting of a balance sheet, statement of income and statement of cash flows for each such period all in reasonable detail and stating in comparative form the figures for the corresponding fiscal quarters in the previous fiscal year and the immediately preceding fiscal year, respectively, all prepared in accordance with GAAP consistently applied.
- (v) <u>Forecasts</u>—within 120 days after the end of each of its fiscal years, the Borrower shall provide the Lender with an annual business and finance plan including a consolidated financial forecast for the next fiscal year (quarter by quarter for the first year), including a balance sheet, statement of income, statement of cash flows, statement of shareholders' equity, a statement of changes from the prior forecasts, a capital expenditure

budget and a *pro forma* calculation of the financial covenants set forth in Section 10.3, each containing reasonable detail, together with a statement of anticipated significant events for the Borrower and Subsidiaries;

- 49 -

(vi) INTENTIONALLY DELETED

- (vii) <u>Financial Instruments</u>—concurrently with furnishing the Compliance Certificates pursuant to Section 10.1 (e)(iii), a report on the status of all outstanding Financial Instruments, such report to be in a form and containing such information as may be required by the Lender; and
- (viii) <u>Other</u>—at the request of the Lender, such other information, reports, engineering data, certificates, projections of income and cash flow or other matters affecting the business, affairs, financial condition, property or assets of the Borrower or the business, affairs, financial condition, property or assets of any of its Subsidiaries as the Lender may reasonably request.

(f) <u>Rights of Inspection</u>

At any reasonable time and from time to time upon reasonable prior notice, the Borrower shall permit the Lender or any representative thereof (at the expense of the Borrower during the continuance of a Default or Event of Default and, otherwise, at the expense of the Lender, as applicable) to (i) examine and make copies of and abstracts from the records and books of account of the Borrower or any of its Subsidiaries, (ii) visit and inspect the premises and properties of the Borrower or any of its Subsidiaries (in each case at the risk of the Borrower, except for the gross negligence or willful misconduct of the inspecting party or the failure of any such inspecting party to comply with the Borrower's or any such Subsidiary's health and safety requirements, as advised to such inspecting party), and (iii) discuss the affairs, operations, finances and accounts of the Borrower or any of the Subsidiaries or directors of the Borrower or any of its Subsidiaries.

(g) Notice of Material Litigation

The Borrower shall promptly give written notice to the Lender of any litigation, proceeding or dispute affecting the Borrower or any of its Subsidiaries in respect of a demand or claim in respect of which there is a reasonable possibility of an adverse determination and which if adversely determined would reasonably be expected to result in a liability, obligation or judgment in excess of Cdn. \$250,000 (or the Equivalent Amount thereof in any other currency) or to have a Material Adverse Effect, and shall from time to time furnish to the Lender all reasonable information requested by the Lender concerning the status of any such litigation, proceeding or dispute.

(h) Notice of Default or Event of Default

The Borrower shall deliver to the Lender, as soon as reasonably practicable and in any event no later than three (3) Banking Days after becoming aware of a Default or an Event of Default, an Officer's Certificate describing in detail such Default or such Event of Default and specifying the steps, if any, being taken to cure or remedy the same.

- 50 -

(i) Notice of Material Adverse Effect

The Borrower shall, as soon as reasonably practicable, promptly notify the Lender of any event, circumstance or condition that has had or is reasonably likely to have a Material Adverse Effect.

(j) <u>New Subsidiaries and Security Related Notices.</u>

The Borrower shall promptly give written notice to the Lender of:

- (i) the acquisition, creation or existence of each new Subsidiary and the information set forth in the Disclosure Schedule as described in Section 9.1(w) with respect to such Subsidiary, which notice shall be provided at least fifteen (15) days before such event;
- (ii) any name change of the Borrower or any Subsidiary, which notice shall be provided at least fifteen (15) days before such name change;
- (iii) any change in the location of the Borrower's or any Subsidiary's chief executive office, which notice shall be provided at least fifteen(15) days before the change in the location;
- (iv) any acquisition (whether by purchase, lease or otherwise) of any real or personal property or assets by the Borrower or any Subsidiary which are intended to be used or kept in any jurisdiction or location not identified in the Disclosure Schedule, or any relocation of existing assets outside said jurisdictions or locations; and
- (v) any of the Borrower or Subsidiary acquiring (whether by purchase, lease or otherwise) an interest in real property where such property has a fair market value in excess of Cdn. \$500,000 (or the Equivalent Amount thereof in any other currency).

(j) Documents of Title

If any negotiable document of title is issued in respect of the inventory or other assets of the Borrower or any Subsidiary, the Borrower shall notify the Lender of such negotiable document of title and the details regarding same and the Borrower or Subsidiary shall at all times maintain possession of such negotiable document of title; provided that, upon the occurrence of an Event of Default, such negotiable document of title shall be delivered to the Lender.

(k) Payment of Taxes, Withholdings, etc.

The Borrower shall, and shall cause its Subsidiaries to, from time to time pay or cause to be paid all material Taxes, rents, rates, levies or assessments, ordinary or extraordinary, governmental fees or dues, and to make and remit all withholdings, lawfully levied, assessed or imposed upon the Borrower or its Subsidiaries or any of the assets of the Borrower or its Subsidiaries, as and when the same become due and payable, except when and so long as the validity of any such material Taxes, rents, rates, levies, assessments, fees, dues or withholdings is being contested by the Borrower or its Subsidiaries by a Permitted Contest.

- 51 -

(l) Payment of Preferred Claims

The Borrower shall, and shall cause its Subsidiaries to, from time to time pay when due or cause to be paid when due all material amounts related to wages, workers' compensation obligations, government royalties or pension fund obligations and any other amount which may result in a lien, charge, Security Interest or similar encumbrance against the assets of the Borrower or such Subsidiary arising under statute or regulation, except when and so long as the validity of any such material amounts or other obligations is being contested by the Borrower or its Subsidiaries by a Permitted Contest.

(m) <u>Environmental Covenants</u>

- (i) Without limiting the generality of Section 10.1(d) above, the Borrower shall, and shall cause its Subsidiaries and any other party acting under their direction to, conduct their business and operations so as to comply at all times with all Environmental Laws.
- (ii) If the Borrower or its Subsidiaries shall:

(A) receive or give any notice that a violation of any Environmental Law has or may have been committed or is about to be committed by the same;

(B) receive any notice that a complaint, proceeding or order has been filed or is about to be filed against the same alleging a violation of any Environmental Law; or

(C) receive any notice requiring the Borrower or a Subsidiary, as the case may be, to take any action in connection with the Release of Hazardous Materials into the environment or alleging that the Borrower or the Subsidiary may be liable or responsible for costs associated with a response to or to clean up a Release of Hazardous Materials into the environment or any damages caused thereby,

if such violation, action or liability could reasonably be expected to give rise to liability of the Borrower and/or any of its Subsidiaries, in the aggregate, in any fiscal year exceeding \$250,000.00, the Borrower shall promptly provide the Lender with a copy of such notice and shall, or shall cause its Subsidiary to, furnish to the Lender from time to time all reasonable information requested by the Lender relating to the same.

(n) Use of Loans

The Borrower shall use all Loans and the proceeds thereof solely for the purposes set forth in Section 2.3 hereof.

- 52 -

(o) <u>Required Insurance</u>

The Borrower shall, and shall cause each of its Subsidiaries to, maintain, with financially sound and reputable insurers, insurance with respect to their respective properties and business and against such casualties and contingencies and in such types and such amounts as shall be required by the Lender (acting reasonably) and, in any event, as shall be in accordance with prudent business practices for persons of the size and type of business and operations as the Borrower and its Subsidiaries.

(p) <u>Compliance With Material Agreements</u>

The Borrower shall, and shall cause each of its Subsidiaries to, comply in all material respects with their respective Material Agreements.

(q) <u>Intellectual Property</u>

The Borrower and its Subsidiaries shall have, or shall have the legal right to use, all Intellectual Property necessary for the operation and conduct of their business, affairs, operations and processes.

10.2 <u>Negative Covenants of the Borrower</u>

So long as any Obligation is outstanding or any Credit Facility is available hereunder, the Borrower covenants and agrees with the Lender that, unless the Lender otherwise consents in writing:

(a) <u>Change of Business</u>

The Borrower shall not, and shall not permit any Subsidiary to, change in any material respect the nature of its business or operations from the types of businesses and intended operations carried on by the Borrower and its Subsidiaries as set out in the business plan presented to the Lender with the Borrower's application or request for the Credit Facilities.

(b) <u>No Change of Control</u>

The Borrower shall remain a Wholly-Owned Subsidiary of Powell.

(c) <u>Negative Pledge</u>

The Borrower shall not, nor shall it permit any of its Subsidiaries to, create, issue, incur, assume or permit to exist any Security Interests on any of their property, undertakings or assets other than Permitted Encumbrances.

(d) <u>No Dissolution</u>

The Borrower shall not, nor shall it permit any of its Subsidiaries to, liquidate, dissolve or wind up or take any steps or proceedings in connection therewith except, in the case of Subsidiaries, where the successor thereto or transferee thereof is the Borrower or another Wholly-Owned Subsidiary of the Borrower.

- 53 -

(e) Limit on Purchase or Sale of Assets

Except for Permitted Dispositions, the Borrower shall not, and shall not permit its Subsidiaries to:

- (i) sell, transfer or otherwise dispose of any of their respective property or assets during the continuance of a Default or Event of Default; or
- purchase, sell, transfer or otherwise dispose of property or assets in any period of twelve consecutive months, whether in one or a series of transactions, having an aggregate fair market value in excess of Cdn. \$500,000 (or the Equivalent Amount thereof in any other currency), other than in the ordinary course of its business.

Notwithstanding this Section 10.2(e), it is understood and agreed that the Borrower may purchase property and equipment for the purposes of establishing its new Canadian Headquarters to a maximum aggregate fair market value of not more that FORTY MILLION (\$40,000,000.00) DOLLARS without being found in violation of this Section 10.2.

(f) Limitation on Debt

The Borrower shall not have or incur, or permit any Subsidiary to have or incur, any Debt other than Permitted Debt, in the aggregate, in excess of \$1,000,000.00 at any time.

(g) Limit on Investment

The Borrower shall not, nor shall it permit any Subsidiary to, make Investments other than:

- (i) Investments consisting of Financial Assistance permitted under Section 10.2(i);
- (ii) Investments in Approved Securities; and
- (iii) Investments consisting of the acquisition of all of the shares of another Person or substantially all of the business assets, property and undertaking of another Person, not exceeding, in the aggregate in any fiscal year, Cdn. \$2,000,000.00 (or the Equivalent Amount thereof in any other currency), provided that in any event any such acquisition shall not be a hostile acquisition or takeover.

(h) Limits on Distributions

The Borrower shall not make, or permit any Subsidiary to make any Distributions.

(i) Limit on Financial Assistance

The Borrower and its Subsidiaries shall not provide any Financial Assistance to or in favour of any person except:

- 54 -

- (i) in favour of the Lender and their respective Hedging Affiliates for or in respect of the Obligations or Lender Financial Instrument Obligations;
- (ii) for the benefit of the Borrower or a Wholly-Owned Subsidiary in connection with Permitted Debt;
- (iii) in favour of the Borrower or a Wholly-Owned Subsidiary;
- (iv) for Investments permitted pursuant to Section 10.2(g); and
- (v) Powell or any of its Subsidiaries

(j) No Financial Instruments Other Than Permitted Hedging

The Borrower and its Subsidiaries shall not enter into, transact or have outstanding any Financial Instruments or Financial Instrument Obligations other than Permitted Hedging.

(k) Non-Arm's Length Transaction

Except in respect of transactions between or among the Borrower and/or one or more of its Wholly-Owned Subsidiaries, the Borrower shall not, nor shall it permit any Subsidiary to, enter into any contract, agreement or transaction whatsoever, including for the sale, purchase, lease or other dealing in any property or the provision of any services (other than office and administration services provided in the ordinary course of business), with any Related Party except upon fair and reasonable terms, which terms are not less favourable to the Borrower or a Subsidiary than it would obtain in an arm's length transaction and, if applicable, for consideration which equals the fair market value of such property or other than at a fair market rental as regards leased property.

(l) <u>No Merger, Amalgamation, etc.</u>

The Borrower shall not, nor shall it permit any of its Subsidiaries to, enter into any transaction whereby all or substantially all of its undertaking, property and assets would become the property of any other person whether by way of reconstruction, reorganization, recapitalization, consolidation, amalgamation, merger, transfer, sale or otherwise except, in the case of Subsidiaries, where the successor thereto or transferee thereof is the Borrower or another Wholly-Owned Subsidiary of the Borrower.

(m) <u>Material Agreements</u>

The Borrower shall not, nor shall it permit any Subsidiary to, take any steps to terminate (without replacement), forfeit, surrender, amend, supplement or modify any Material Agreement (or provide any waiver or consent to like effect) or waive any failure of any counterparty thereto perform its obligations thereunder, if any of the foregoing would have or reasonably be expected to have a Material Adverse Effect or if the amendment, supplement, modification, waiver or consent relates to the assignment provisions of such agreement.

- 55 -

10.3 Financial Covenants

So long as any Obligation is outstanding or any Credit Facility is available hereunder, Powell and the Borrower each covenants and agrees with the Lender that, unless the Lender otherwise consent in writing:

- (a) <u>Maximum Debt to EBITDA Ratio</u>
 - (i) As of the Closing Date and as of each Quarter End thereafter, the Debt to EBITDA Ratio for Powell shall not exceed 2.75:1.0.
- (b) Minimum Fixed Charge Coverage Ratio

As at each Quarter End, the Fixed Charge Coverage Ratio for Powell shall be equal to or greater than 1.25:1.0.

(c) <u>Consolidated Tangible Net Worth</u>

As at each Quarter End, commencing with the Quarter ending March 31, 2012, the Consolidated Tangible Net Worth of Powell shall be equal or greater than the sum of:

- (i) U.S. \$172,500,000, plus
- (ii) an amount equal to 50% of the Net Income for each fiscal Quarter, commencing with the fiscal quarter ended March 31, 2012, and for each fiscal quarter thereafter (with no deduction for any net loss in any fiscal quarter), plus
- (iii) an amount equal to 100% of the aggregate increase in Shareholders' Equity of Powell and its Subsidiaries after the date hereof by reason of the issuance and sale of any Equity Interests of Powell or any of its Subsidiaries (other than issuances to Powell or a Wholly-Owned Subsidiary), including upon any conversion of any debt securities of Powell into capital stock or other equity interests.

10.4 Lender May Perform Covenants

If Powell or the Borrower fails to perform any covenants on its part herein contained, subject to any consents or notice or cure periods required by Section 12.1, the Lender may give notice to Powell and the Borrower of such failure and if such covenant remains unperformed, the Lender may, in its discretion but need not, perform any such covenant capable of being performed by the Lender and if the covenant requires the payment or expenditure of money, the Lender may make such payments or expenditure and all sums so expended shall be forthwith payable by the Borrower to the Lender and shall bear interest at the applicable interest rate provided in Section 5.8 for amounts due in Canadian Dollars or United States Dollars, as the case may be. No such performance, payment or expenditure by the Lender shall be deemed to relieve the Borrower of any default hereunder or under the other Documents.

- 56 -

ARTICLE 11 SECURITY

11.1 Security on all Assets

The (i) Obligations and (ii) Lender Financial Instrument Obligations shall be secured by first priority perfected Security Interests on, to and against all present and future property, assets and undertaking of the Borrower and each of its Subsidiaries. As continuing collateral security for the Obligations and the Lender Financial Instrument Obligations, the Borrower has delivered or shall deliver to the Lender the following Security (unless expressly indicated otherwise) completed in form and manner satisfactory to the Lender or its solicitors:

- (a) line of credit by way of current account overdraft agreement executed by the Borrower respecting the Revolving Facility;
- (b) general security agreement executed by the Borrower creating a first priority security interest in all present and after acquired personal property of the Borrower and a floating charge over all of the Borrower's present and after acquired real property (with a floating charge to be registered against the Borrower at Personal Property Registry and against titles to Borrower's real property at Alberta Land Titles);
- (c) guarantee of the indebtedness of the Borrower to the Lender executed by Powell, limited to Cdn. \$12,500,000.00 plus interest and charges as provided in the guarantee;
- (d) unlimited guarantee of the indebtedness of the Borrower to the Lender executed by each of the Subsidiaries of the Borrower from time to time, including Nextron and PPC Technical, supported by a general security agreement creating a first priority security interest in all present and after acquired personal property of each such Subsidiary and a floating charge over all of each such Subsidiary's present and after acquired real property;
- (e) the Lender's standard application and indemnity agreement with respect to the issuance of Letters of Credit, executed by the Borrower;
- (f) Banker's Acceptances agreement, executed by the Borrower;
- (g) the Bank's standard electronic funds transfer agreement respecting the EFT Facility, executed by the Borrower;
- (h) the Bank's standard Mastercard agreements respecting the MC Facility, executed by the Borrower and cardholders;
- (i) the Bank's standard foreign exchange contracts agreement respecting the F/X Facility, executed by the Borrower and cardholders; and
- (j) any related documents and registrations required by the Lender or its solicitors (acting reasonably), including, without limitation, all supporting certificates and opinions as the Lender may reasonably require.

- 57 -

11.2 <u>Registration</u>

The Borrower shall, at its expense, register, file or record the Security in all offices where such registration, filing or recording is necessary or of advantage to the creation, perfection and preserving of the security applicable to it. The Borrower shall amend and renew such registrations, filings and recordings from time to time as and when required to keep them in full force and effect or to preserve the priority established by any prior registration, filing or recording thereof. The Security shall rank in priority to all other mortgages, charges, liens, encumbrances and security interests unless otherwise specifically agreed to in writing by the Lender.

11.3 <u>Forms</u>

The forms of Security shall have been or be prepared based upon the laws of Alberta applicable thereto in effect at the date hereof. The Lender shall have the right to require that:

- (a) any such Security be amended to reflect any changes in such laws, whether arising as a result of statutory amendments, court decisions or otherwise, in order to confer upon the Lender the Security Interests intended to be created thereby, and
- (b) the Borrower and its Subsidiaries execute and deliver to the Lender such other and further debentures, mortgages, trust deeds, assignments and security agreements as may be reasonably required to ensure the Lender holds, subject to Permitted Encumbrances, first priority, perfected Security Interests on and against all of the property and assets of the Borrower and its Subsidiaries.

11.4 Continuing Security

Each item or part of the Security shall for all purposes be treated as a separate and continuing collateral security and shall be deemed to have been given in addition to and not in place of any other item or part of the Security or any other security now held or hereafter acquired by the Lender. No item or part of the Security shall be merged or be deemed to have been merged in or by this Agreement or any documents, instruments or acknowledgements delivered hereunder, or any simple contract debt or any judgment, and any realization of or steps taken under or pursuant to any security, instrument or agreement shall be independent of and not create a merger with any other right available to the Lender under any security, instruments or agreements held by it or at law or in equity.

11.5 Dealing with Security

The Lender may grant extensions of time or other indulgences, take and give up securities (including the Security or any part or parts thereof), accept compositions, grant releases and discharges and otherwise deal with the Borrower and other parties and with security (including without limitation, the Security and each part thereof) as the Lender may see fit, without prejudice to or in any way limiting the liability of the Borrower under this Agreement or the other Documents or under any of the Security or any other collateral security.

- 58 -

11.6 Effectiveness

The Security and the security created by any other Document constituted or required to be created shall be effective, and the undertakings as to the Security herein or in any other Document shall be continuing, whether any Loans or Lender Financial Instrument Obligations are then outstanding or any amounts thereby secured or any part thereof shall be owing before or after, or at the same time as, the creation of such Security Interests or before or after or upon the date of execution of any amendments to this Agreement.

11.7 <u>Release and Discharge of Security</u>

The Borrower and its Subsidiaries shall not be discharged from the Security or any part thereof, other than to the extent that such Security applies to a Permitted Disposition (in which case the Security shall cease to apply to the subject matter thereof) except by a written release and discharge signed by the Lender. If all of the Obligations and Lender Financial Instrument Obligations have been unconditionally and indefeasibly repaid, paid, satisfied and discharged, as the case may be, in full and the Credit Facilities have been fully cancelled, then the Lender shall release and discharge the Security, all at the expense of the Borrower.

ARTICLE 12 EVENTS OF DEFAULT AND ACCELERATION

12.1 Events of Default

The occurrence of anyone or more of the following events (each such event being herein referred to as an "Event of Default") shall constitute a default under this Agreement:

- (a) <u>Principal Default:</u> if the Borrower fails to pay the principal of any Loan hereunder when due and payable or fails to cash collateralize any Bankers' Acceptance or Letter of Credit when required to do so hereunder;
- (b) <u>Other Payment Default:</u> if the Borrower fails to pay:
 - (i) any interest (including, if applicable, default interest) accrued on any Loan;
 - (ii) any acceptance fee with respect to a Bankers' Acceptance; or
 - (iii) any other amount not specifically referred to in Section 12.1(a) or in this Section 12.1(b) payable by the Borrower hereunder;

in each case when due and payable, and such default is not remedied within three (3) Banking Days after written notice thereof is given by the Lender to the Borrower specifying such default and requiring the Borrower to remedy or cure the same;

- (c) <u>Certain Covenant Defaults:</u> if the Borrower fails to observe or perform any covenant in Sections 10.2(b) to (j), inclusive and Section 10.2(l), or should the Borrower or Powell fail to observe or perform any covenant in Section 10.3;
- (d) <u>Breach of Other Covenants</u>: if the Borrower or a Subsidiary fails to observe or perform any covenant or obligation herein or in any other Document required on its part to be observed or performed (other than a covenant or condition whose breach or default in performance is specifically dealt with elsewhere in this Section) and, after notice has been given by the Lender to the Borrower or Subsidiary specifying such default and requiring the Borrower or Subsidiary to remedy or cure the same, the Borrower or Subsidiary shall fail to remedy such default within a period of twenty (20) Banking Days after the giving of such notice;

- 59 -

- (e) <u>Incorrect Representations</u>: if any representation or warranty made by Powell, the Borrower or any Subsidiary herein or in any other Document shall prove to have been incorrect or misleading in any respect on and as of the date made and the facts or circumstances which make such representation or warranty incorrect or misleading are not remedied and the representation or warranty in question remains incorrect or misleading more than twenty (20) Banking Days after the Lender notifies the Borrower of the same;
- (f) <u>Involuntary Insolvency:</u> if a decree or order of a court of competent jurisdiction is entered adjudging the Borrower or a Subsidiary a bankrupt or insolvent under the *Companies' Creditors Arrangement Act* (Canada), the *Bankruptcy and Insolvency Act* (Canada), the *Winding-up and Restructuring Act* (Canada) or any other bankruptcy, insolvency or analogous laws or ordering the winding up or liquidation of its affairs; or if a decree or order of a court of competent jurisdiction is entered adjudging Powell a bankrupt or insolvent under any bankruptcy, insolvency or analogous laws or ordering the winding up for liquidation of its affairs;
- (g) <u>Idem:</u> if any case, proceeding or other action shall be instituted in any court of competent jurisdiction against Powell, the Borrower or any Subsidiary, seeking in respect of it an adjudication in bankruptcy, reorganization, dissolution, winding up, liquidation, a composition, proposal or arrangement with creditors, a readjustment of debts, the appointment of trustee in bankruptcy, receiver, receiver and manager, interim receiver, custodian, sequestrator or other person with similar powers with respect to Powell, the Borrower or any Subsidiary or of all or any substantial part of its assets, or any other like relief in respect of Powell, the Borrower or any Subsidiary under any bankruptcy or insolvency law and:
 - (i) such case, proceeding or other action results in an entry of an order for such relief or any such adjudication or appointment, or
 - (ii) such case, proceeding or other action shall continue undismissed, or unstayed and in effect, for any period of ten (10) consecutive Banking Days;
- (h) <u>Voluntary Insolvency</u>: if the Borrower or any Subsidiary makes any assignment in bankruptcy or makes any other assignment for the benefit of creditors or makes any proposal under the *Bankruptcy and Insolvency Act* (Canada) or any comparable law, seeks relief under the *Companies*' *Creditors Arrangement Act* (Canada), the *Winding-up and Restructuring Act* (Canada) or any other bankruptcy, insolvency or analogous law; or if Powell makes an assignment in bankruptcy or makes any other assignment for the benefit of creditors or makes any proposal under any bankruptcy, insolvency or analogous law; or if Powell, the Borrower or any Subsidiary files a petition or proposal to take advantage of any act of insolvency, consents to or acquiesces in the appointment of a trustee in bankruptcy, receiver, receiver and manager, interim receiver, custodian,

- 60 -

sequestrator or other person with similar powers of itself or of all or any substantial portion of its assets, or files a petition or otherwise commences any proceeding seeking any reorganization, arrangement, composition, administration or readjustment under any applicable bankruptcy, insolvency, moratorium, reorganization or other similar law affecting creditors' rights or consents to, or acquiesces in, the filing of such assignment, proposal, relief, petition, proposal, appointment or proceeding;

- (i) <u>Dissolution:</u> except as permitted by Section 10.2(d), if proceedings are commenced for the dissolution, liquidation or winding up of Powell, the Borrower or any Subsidiary unless such proceedings are being actively and diligently contested in good faith to the satisfaction of the Lender;
- (j) <u>Security Realization</u>: if creditors of Powell, the Borrower or any Subsidiaries having a Security Interest against or in respect of the property and assets thereof, or any part thereof, realize upon or enforce any such security against such property and assets or any part thereof having an aggregate fair market value in excess of Cdn. \$100,000 (or the Equivalent Amount thereof in United States Dollars or the equivalent thereof in any other currency) and such realization or enforcement shall continue in effect and not be released, discharged or stayed within the lesser of twenty (20) Banking Days and the period of time prescribed under Applicable Laws for the completion of the sale of or realization against the assets subject to such seizure or attachment;
- (k) Seizure: if property and assets of Powell, the Borrower and its Subsidiaries or any part thereof having an aggregate fair market value in excess of Cdn. \$100,000 (or the Equivalent Amount thereof in any other currency) are seized or otherwise attached by anyone pursuant to any legal process or other means, including, without limitation, distress, execution or any other step or proceeding with similar effect and such attachment, step or other proceeding shall continue in effect and not be released, discharged or stayed within the lesser of twenty (20) Banking Days and the period of time prescribed under Applicable Laws for the completion of the sale of or realization against the assets subject to such seizure or attachment;
- Judgment: if one or more final judgments, decrees or orders, after available appeals have been exhausted, shall be awarded against Powell, the Borrower or any Subsidiary for an aggregate amount in excess of Cdn. \$100,000 (or the Equivalent Amount thereof in any other currency) and Powell or the Borrower has not provided security for any of such judgments, decrees or orders within twenty (20) Banking Days of such judgment, decree or order being awarded;
- (m) <u>Payment Cross-Default:</u> if Powell, the Borrower or any of its Subsidiaries (or any combination thereof) defaults in the payment when due (whether at maturity, upon acceleration, or otherwise) of Debt or Financial Instrument Obligations thereof in aggregate in excess of Cdn. \$1,000,000 (or the Equivalent Amount thereof in any other currency);
- (n) <u>Event Cross Default:</u> if a default, event of default or other similar condition or event (however described) in respect of Powell, the Borrower or any of its Subsidiaries (or any combination thereof) occurs or exists under any indentures, credit agreements, agreements or other instruments evidencing or relating to

- 61 -

Debt or Financial Instrument Obligations thereof(individually or collectively) in an aggregate amount in excess of Cdn. \$1,000,000 (or the Equivalent Amount thereof in any other currency) and such default, event or condition has resulted in such Debt or Financial Instrument Obligations becoming, or becoming capable at such time of being declared, due and payable thereunder before it would otherwise have been due and payable (whether or not it is so declared), unless the default, event or condition has been remedied or waived in accordance with the provisions of the relevant indentures, credit agreements, agreements or other instruments;

- (o) <u>Cease to Carry on Business:</u> if Powell, the Borrower or any Subsidiary ceases to carry on business, except in the case of Subsidiaries in compliance with the Documents;
- (p) <u>Change of Control:</u> if the Borrower ceases to be a Wholly-Owned Subsidiary of Powell, or if any of the Subsidiaries of the Borrower cease to be Wholly-Owned Subsidiaries of the Borrower;
- (q) <u>Lender Financial Instruments:</u> if a Financial Instrument Demand for Payment has been delivered to the Borrower or any Subsidiary and such person fails to make payment thereunder within the time otherwise required for payment thereunder, or if a Termination Event occurs;
- (r) Loss and Priority of Security: except for Permitted Encumbrances, if any of the Security shall cease to be a valid first priority Security Interest against the property, assets and undertaking of the Borrower or any Subsidiary as against third parties (and the same is not forthwith effectively rectified or replaced by the Borrower upon becoming aware thereof);
- (s) <u>Invalidity</u>: if any of this Agreement or any Security or any material provision of any of the foregoing shall at any time for any reason cease to be in full force and effect, be declared to be void or voidable (and the same is not forthwith effectively rectified or replaced by Powell or the Borrower upon becoming aware thereof) or shall be repudiated, or the validity or enforceability thereof shall at any time be contested by Powell or the Borrower or any Subsidiary of the Borrower, or should Powell, the Borrower or any Subsidiary deny that it has any or any further liability or obligation thereunder, or at any time it shall be unlawful or impossible for them to perform any of their respective Obligations;
- (t) <u>Qualified Auditor's Report:</u> if the audited consolidated financial statements of Powell are issued with a report of Powell's auditors which is qualified in any material respect and such qualification is not removed within 20 Banking Days; and
- (u) <u>Material Adverse Effect:</u> if any event or circumstance has occurred and is continuing which, in the opinion of the Lender (acting reasonably), has had or would reasonably be expected to have a Material Adverse Effect.

- 62 -

12.2 Acceleration

If any Event of Default shall occur and for so long as it is continuing:

- (a) the entire principal amount of all Loans then outstanding from the Borrower and all accrued and unpaid interest thereon,
- (b) an amount equal to the face amount at maturity of all Bankers' Acceptances issued by the Borrower which are unmatured, and
- (c) all other Obligations outstanding hereunder,

shall, at the option of the Lender, become immediately due and payable upon written notice to that effect from the Lender to the Borrower, all without any other notice and without presentment, protest, demand, notice of dishonour or any other demand whatsoever (all of which are hereby expressly waived by the Borrower). In such event and if the Borrower does not immediately pay all such amounts upon receipt of such notice, the Lender may exercise any right or recourse and/or proceed by any action, suit, remedy or proceeding against the Borrower, Powell and other guarantors authorized or permitted by law for the recovery of all the indebtedness and liabilities of the Borrower to the Lender and proceed to exercise any and all rights hereunder and under the other Documents and no such remedy for the enforcement of the rights of the Lender shall be exclusive of or dependent on any other remedy but anyone or more of such remedies may from time to time be exercised independently or in combination.

12.3 Set Off; Cash Collateral Accounts.

- (a) Upon the occurrence of an Event of Default, the Lender may require the Borrower to forthwith pay funds in an amount sufficient to pay the maximum aggregate amount for which the Lender is or may become liable in respect of all outstanding Bankers' Acceptances and Letters of Credit into a cash collateral account in accordance with Section 2.14 and any amount not so paid by the Borrower may, at the option of the Lender and without notice to the Borrower, be paid by the Lender into a cash collateral account and shall be deemed to constitute a Canadian Prime Rate Loan; and
- (b) In addition to any rights now or hereafter granted under Applicable Law and not by way of limitation of any such rights, upon the occurrence of an Event of Default which remains unremedied or unwaived (whether or not the Loans have been accelerated hereunder), the Lender shall have the right (and is hereby authorized by the Borrower) at any time and from time to time to combine all or any of the Borrower's or Subsidiaries' accounts with the Lender and to set off and to appropriate and to apply any and all deposits (general or special, term or demand) including, but not limited to, indebtedness evidenced by certificates of deposit whether matured or unmatured, and any other indebtedness at any time held by the Borrower or owing by the Lender to or for the credit or account of the Borrower against and towards the satisfaction of any Obligations, and may do so notwithstanding that the balances of such accounts and the liabilities are expressed indifferent currencies, and the Lender is hereby authorized to effect any necessary currency conversions at the applicable Bank of Canada noon rate on the Banking Day before the day of conversion. The Lender shall notify the Borrower of any such set-off from the Borrower's accounts within a reasonable period of time thereafter.

- 63 -

12.4 <u>Remedies Cumulative and Waivers</u>

For greater certainty, it is expressly understood and agreed that the rights and remedies of the Lender hereunder or under any other Document are cumulative and are in addition to and not in substitution for any rights or remedies provided by law or by equity; and any single or partial exercise by the Lender of any right or remedy for a default or breach of any term, covenant, condition or agreement contained in this Agreement or other Document shall not be deemed to be a waiver of or to alter, affect or prejudice any other right or remedy or other rights or remedies to which the Lender may be lawfully entitled for such default or breach. Any waiver by, the Lender of the strict observance, performance or compliance with any term, covenant, condition or other matter contained herein and any indulgence granted, either expressly or by course of conduct, by the Lender shall be effective only in the specific instance and for the purpose for which it was given and shall be deemed not to be a waiver of any rights and remedies of the Lender under this Agreement or any other Document as a result of any other default or breach hereunder.

12.5 <u>Termination of Lender's Obligations</u>

The occurrence of a Default or Event of Default shall relieve the Lender of all obligations to provide any further Drawdowns, Rollovers or Conversions to the Borrower hereunder; provided that the foregoing shall not prevent the Lender from disbursing money or effecting any Conversion which, by the terms hereof, it is entitled to effect, or any Conversion or Rollover requested by the Borrower and acceptable to the Lender.

12.6 Acceleration of All Lender Obligations

- (a) If:
 - (i) a Termination Event has occurred;
 - (ii) a Financial Instrument Demand for Payment has been delivered to the Borrower or a Subsidiary by the Lender of Hedging Affiliate and the cure period provided in Section 12.1(q) has expired; or
 - (iii) an Acceleration Notice has been delivered to the Borrower,

then, to the extent that it is not already the case, all Obligations and all Financial Instrument Obligations under Lender Financial Instruments shall be immediately due and payable and the Lender shall (and shall be entitled to) deliver such other Demands for Payment and notices as may be necessary to ensure that all Obligations and Financial Instrument Obligations under Lender Financial Instruments are thereafter due and payable under this Agreement and the Lender Financial Instruments, as applicable.

(b) Each agreement, indenture, instrument or other document evidencing or relating to a Lender Financial Instrument shall, notwithstanding any provision thereof to the contrary, be deemed to be hereby amended to allow and permit the Lender to comply with the provisions of this Section 12.6.

- 64 -

12.7 Application of Payments Following Acceleration

All monies and property received by the Lender for application in respect of the Obligations and the Financial Instrument Obligations under Lender Financial Instruments subsequent to the Adjustment Time and all monies received as a result of a realization upon the Security (collectively, the "**Realization Proceeds**") shall be applied in the order and manner set forth below:

- (a) firstly, applied and distributed on account of the costs and expenses of enforcement and realization upon the Security or costs, expenses and fees otherwise owing to the Lender hereunder or under the Security; and
- (b) secondly, distributed to the Lender and Hedging Affiliates on account of the Obligations and the Financial Instrument Obligations under Lender Financial Instruments,

and the balance (if any) of the Realization Proceeds after the payment in full and in cash of all such obligations shall be paid to the Borrower or otherwise as may be required by Applicable Law.

12.8 Calculations as at the Adjustment Time

For the purposes of this Agreement, if:

- (a) a Financial Instrument Demand for Repayment has been delivered; or
- (b) a Termination Event has occurred under any agreement evidencing a Permitted Lender Financial Instrument;

then any amount which is payable by the Borrower or a Subsidiary under such Lender Financial Instrument in settlement of obligations arising thereunder as a result of the early termination of the Lender Financial Instrument shall be deemed to have become payable at the time of delivery of such Financial Instrument Demand for Repayment or the time of occurrence of such Termination Event, as the case may be, notwithstanding that the amount payable by the Borrower or a Subsidiary is to be subsequently calculated and notice thereof given to the Borrower or such Subsidiary in accordance with such Lender Financial Instrument.

ARTICLE 13 CHANGE OF CIRCUMSTANCES

13.1 Market Disruption Respecting Bankers' Acceptances

If the Lender (acting reasonably) makes a determination, which determination shall be conclusive and binding upon the Borrower, and notifies the Borrower, that:

- (a) there no longer exists an active market for bankers' acceptances accepted by the Lender; or
- (b) the bid rate does not accurately reflect the rate which would be applicable to a sale of Bankers' Acceptances in the market;

then:

- 65 -

- (c) the right of the Borrower to request Bankers' Acceptances from the Lender shall be suspended until the Lender determines that the circumstances causing such suspension no longer exist, and so notifies the Borrower;
- (d) any outstanding Drawdown Notice requesting a Loan by way of Bankers' Acceptances shall be deemed to be a Drawdown Notice requesting a Loan by way of Canadian Prime Rate Loans in the amount specified in the original Drawdown Notice; and
- (e) any outstanding Rollover Notice requesting a Rollover of a Loan by way of Bankers' Acceptances shall be deemed to be a Conversion Notice requesting a Conversion of such Loans into a Loan by way of Canadian Prime Rate Loans.

The Lender shall promptly notify the Borrower of any suspension of the Borrower's right to request the Bankers' Acceptances and of any termination of any such suspension.

13.2 Illegality

If a Lender determines, in good faith, that the adoption of any Applicable Law, regulation, treaty or official directive (whether or not having the force of law) or any change therein or in the interpretation or application thereof by any court or by any Governmental Authority or any other entity charged with the interpretation or administration thereof or compliance by the Lender with any request or direction (whether or not having the force of law) of any such authority or entity, now or hereafter makes it unlawful or impossible for the Lender to make, fund or maintain a Loan under a Credit Facility or to give effect to its obligations in respect of such a Loan, the Lender may, by written notice thereof to the Borrower declare its obligations under this Agreement in respect of such Loan to be terminated whereupon the same shall forthwith terminate, and the Borrower shall, within the time required by such law (or at the end of such longer period as the Lender at its discretion has agreed), either effect a Conversion of such Loan in accordance with the provisions hereof (if such Conversion would resolve the unlawfulness or impossibility) or prepay the principal of such Loan together with accrued interest, such additional compensation as may be applicable with respect to such Loan to the date of such payment and all costs, losses and expenses incurred by the Lender by reason of the liquidation or redeployment of deposits or other funds or for any other reason whatsoever resulting from the repayment of such Loan or any part thereof on other than the last day of the applicable Interest Period. If any such change shall only affect a portion of the Lender's obligations under this Agreement which is, in the opinion of the Lender, severable from the remainder of this Agreement so that the remainder of this Agreement may be continued in full force and effect without otherwise affecting any of the obligations of the Lender or the Borrower hereunder, such Lender shall only declare its obligations under that portion so te

ARTICLE 14 COSTS, EXPENSES AND INDEMNIFICATION

14.1 Costs and Expenses

The Borrower shall pay promptly upon notice from the Lender all reasonable out-of-pocket costs and expenses of the Lender in connection with the Documents and the establishment of the Credit Facilities, including in connection with preparation, printing, execution and delivery of this Agreement and the other Documents whether or not any Drawdown has been made hereunder, and also including, without limitation, the reasonable fees and out-of-pocket costs and expenses of Lender's legal counsel with respect thereto and with respect to advising the Lender as to their

- 66 -

rights and responsibilities under this Agreement and the other Documents. Except for ordinary expenses of the Lender relating to the day to day administration of this Agreement, the Borrower further agrees to pay within 30 days of demand by the Lender all reasonable out-of-pocket costs and expenses in connection with the preparation or review of waivers, consents and amendments pertaining to this Agreement, and in connection with the establishment of the validity and enforceability of this Agreement and the preservation or enforcement of rights of the Lender under this Agreement and other Documents, including, without limitation, all reasonable out-of-pocket costs and expenses sustained by the Lender as a result of any failure by the Borrower to perform or observe any of its obligations hereunder or in connection with any action, suit or proceeding (whether or not an Indemnified Party is a party or subject thereto), together with interest thereon from and after such 30th day if such payment is not made by such time, such costs to include Lender's legal costs on a solicitor and his own client basis.

14.2 General Indemnity

In addition to any liability of the Borrower to the Lender under any other provision hereof, the Borrower shall indemnify each Indemnified Party and hold each Indemnified Party harmless against any losses, claims, costs, damages or liabilities (including, without limitation, any expense or cost incurred in the liquidation and redeployment of funds acquired to fund or maintain any portion of a Loan and reasonable out-of-pocket expenses and reasonable legal fees on a solicitor and his own client basis) incurred by the same as a result of or in connection with the Credit Facilities or the Documents (including any use of the proceeds of any Loan), including as a result of or in connection with:

- (a) any cost or expense incurred by reason of the liquidation or redeployment in whole or in part of deposits or other funds required by the Lender to fund any Bankers' Acceptance or to fund or maintain any Loan as a result of the Borrower's failure to complete a Drawdown or to make any payment, repayment or prepayment on the date required hereunder or specified by it in any notice given hereunder;
- (b) subject to permitted or deemed Rollovers and Conversions, the Borrower's failure to provide for the payment to the Lender of the full principal amount of each Bankers' Acceptance on its maturity date;
- (c) the Borrower's failure to pay any other amount, including without limitation any interest or fee, due hereunder on its due date after the expiration of any applicable grace or notice periods (subject, however, to the interest obligations of the Borrower hereunder for overdue amounts);
- (d) the prepayment of any outstanding Bankers' Acceptance before the maturity date of such Bankers' Acceptance;
- (e) the Borrower's failure to give any notice required to be given by it to the Lender hereunder;
- (f) the failure of the Borrower to make any other payment due hereunder;
- (g) any inaccuracy or incompleteness of the representations and warranties contained in Article 9;

- 67 -

- (h) any failure of the Borrower or Powell to observe or fulfill its obligations under Article 10;
- (i) any failure of the Borrower to observe or fulfill any other Obligation not specifically referred to above; or
- (j) the occurrence of any Default or Event of Default in respect of the Borrower, any of its subsidiaries or Powell,

provided that this Section shall not apply to any losses, claims, costs, damages or liabilities that arise by reason of the gross negligence or willful misconduct of the Indemnified Party claiming indemnity hereunder. The provisions of this Section shall survive repayment of the Obligations.

14.3 Environmental Indemnity

The Borrower shall indemnify and hold harmless the Indemnified Parties forthwith on demand by the Lender from and against any and all claims, suits, actions, debts, damages, costs, losses, liabilities, penalties, obligations, judgments, charges, expenses and disbursements (including without limitation, all reasonable legal fees and disbursements on a solicitor and his own client basis) of any nature whatsoever, suffered or incurred by the Indemnified Parties or any of them in connection with the Credit Facilities, whether as beneficiaries under the Documents, as successors in interest of Lender, or voluntary transfer in lieu of foreclosure, or otherwise howsoever, with respect to any Environmental Claims relating to the property of the Borrower or any of its Subsidiaries arising under any Environmental Laws as a result of the past, present or future operations of the Borrower or any of its Subsidiaries) relating to the property of the Borrower or its Subsidiaries owned, operated or leased by the Borrower or its Subsidiaries (or any such predecessor in interest), including any liabilities arising as a result of any indemnity covering Environmental Claims given to any person by the Lender or a receiver, receiver manager or similar person appointed hereunder or under Applicable Law (collectively, the "**Indemnified Third Party**"); but excluding any Environmental Claims or liabilities arise by reason of the gross negligence or willful misconduct of the Indemnified Party or the Indemnified Third Party claiming indemnity hereunder. The provisions of this Section shall survive the repayment of the Obligations.

14.4 Judgment Currency

- (a) If for the purpose of obtaining or enforcing judgment against Powell, the Borrower or any of its Subsidiaries in any court in any jurisdiction, it becomes necessary to convert into any other currency (such other currency being hereinafter in this Section referred to as the "Judgment Currency") an amount due in Canadian Dollars under this Agreement, the conversion shall be made at the rate of exchange prevailing on the Banking Day immediately preceding:
 - (i) the date of actual payment of the amount due, in the case of any proceeding in the courts of any jurisdiction that will give effect to such conversion being made on such date; or
 - (ii) the date on which the judgment is given, in the case of any proceeding in the courts of any other jurisdiction (the date as of which such conversion is made pursuant to this Section being hereinafter in this Section referred to as the "**Judgment Conversion Date**").

- 68 -

- (b) If, in the case of any proceeding in the court of any jurisdiction referred to in Section 14.4(a)(ii), there is a change in the rate of exchange prevailing between the Judgment Conversion Date and the date of actual payment of the amount due, the Borrower shall pay such additional amount (if any) as may be necessary to ensure that the amount paid in the Judgment Currency, when converted at the rate of exchange prevailing on the date of payment, will produce the amount of Canadian Dollars which could have been purchased with the amount of Judgment Currency stipulated in the judgment or judicial order at the rate of exchange prevailing on the Judgment Conversion Date.
- (c) Any amount due from the Borrower under the provisions of Section 14.4(b) shall be due as a separate debt and shall not be affected by judgment being obtained for any other amounts due under or in respect of this Agreement.
- (d) The term "rate of exchange" in this Section 14.4 means the noon rate of exchange for Canadian interbank transactions in Canadian Dollars in the Judgment Currency published by the Bank of Canada for the day in question, or if such rate is not so published by the Bank of Canada, such term shall mean the equivalent amount of the Judgment Currency determined by the Lender, acting reasonably.

ARTICLE 15 GENERAL

15.1 Exchange and Confidentiality of Information

- (a) Powell and the Borrower each agrees that the Lender may provide any assignee or any *bona fide* prospective assignee pursuant to Section 15.5 with any information concerning the financial condition of Powell, the Borrower and its Subsidiaries provided such party agrees in writing with the Lender for the benefit of Powell and the Borrower to be bound by a like duty of confidentiality to that contained in this Section.
- (b) The Lender acknowledges the confidential nature of the financial, operational and other information and data provided and to be provided to it by Powell, the Borrower pursuant hereto (the **"Information")** and agrees to use all reasonable efforts to prevent the disclosure thereof provided, however, that:
 - (i) the Lender may disclose all or any part of the Information if, in its reasonable opinion, such disclosure is required in connection with any actual or threatened judicial, administrative or governmental proceedings including, without limitation, proceedings initiated under or in respect of this Agreement;
 - (ii) the Lender shall incur no liability in respect of any Information required to be disclosed by any Applicable Law or regulation, or by applicable order, policy or directive having the force of law, to the extent of such requirement;

- 69 -

- (iii) the Lender may provide Lender's legal counsel and its other agents and professional advisors with any Information; provided that such persons shall be under a like duty of confidentiality to that contained in this Section;
- (iv) the Lender shall incur no liability in respect of any Information: (A) which is or becomes readily available to the public (other than by a breach hereof) or which has been made readily available to the public by Powell, the Borrower or its Subsidiaries, (B) which the Lender can show was, prior to receipt thereof from Powell or the Borrower, lawfully in the Lender's possession and not then subject to any obligation on its part to Powell or the Borrower to maintain confidentiality, or (C) which the Lender received from a third party who was not, to the knowledge of the Lender, under a duty of confidentiality to Powell or the Borrower at the time the information was so received;
- (v) the Lender may disclose the Information to other financial institutions and other persons in connection with the syndication by the Lender of a Credit Facility or the granting by the Lender of a participation in the Credit Facility where such financial institution or other person agrees to be under a like duty of confidentiality to that contained in this Section; and
- (vi) the Lender may disclose all or any part of the Information so as to enable the Lender to initiate any lawsuit against Powell, the Borrower or any of its Subsidiaries or to defend any lawsuit commenced by Powell, the Borrower or any of its Subsidiaries the issues of which touch on the Information, but only to the extent such disclosure is necessary to the initiation or defense of such lawsuit.

15.2 Notices

(a) <u>Notices Generally.</u> Except in the case of notices and other communications expressly permitted to be given by telephone, all notices and other communications provided for herein shall be in writing and shall be delivered by hand or overnight courier service, mailed by registered mail or sent by telecopier to the addresses or telecopier numbers specified below or, if to a Subsidiary, in care of the Borrower. Notices sent by hand or overnight courier service, or mailed by registered mail, shall be deemed to have been given when received; notices sent by telecopier shall be deemed to have been given when sent (except that, if not given on a business day between 9:00 a.m. and 5:00 p.m. local time where the recipient is located, shall be deemed to have been given at 9:00 a.m. on the next business day for the recipient). The respective parties' addresses and contact information shall be as follows:

To the Borrower, Nextron and PPC Technical:

Powell Canada Inc. 6005 – 72A Avenue Edmonton, Alberta T6B 2J1 Attention: Don Madison Facsimile:

- 70 -

To Powell:

Powell Industries Inc. 8550 Mosley Drive Post Office Box 12818 Houston, Texas, USA 77075 Attention: Don Madison Facsimile:

To the Lender:

HSBC Bank Canada 9th Floor, 10250 – 101 Street NW Edmonton, Alberta, Canada T5J 3P4 Attention: Aaron Spells Fax: (780) 426-2660

(b) <u>Change of Address.</u> Any party hereto may change its address or telecopier number for notices and other communications hereunder by notice to the other parties hereto.

15.3 Governing Law

This Agreement shall be governed by and construed in accordance with the laws of the Province of Alberta and the laws of Canada applicable therein, without prejudice to or limitation of any other rights or remedies available under the laws of any jurisdiction where property or assets of the Borrower may be found.

15.4 Benefit of the Agreement

This Agreement shall enure to the benefit of and be binding upon Powell, the Borrower, the Lender and their respective successors and permitted assigns.

15.5 Assignments and Participations

The Lender may, without consent of Powell or the Borrower during the continuance of an Event of Default and at all other times with the prior written consent of Powell and the Borrower, which consents shall not be unreasonably withheld or delayed:

- (a) sell, assign, transfer and grant an interest in any or all of the Commitments under any or all of the Credit Facilities, the Loans under such Credit Facilities and its rights under the Documents to any person or financial institution.
- (b) INTENTIONALLY DELETED
- (c) INTENTIONALLY DELETED
- (d) INTENTIONALLY DELETED

- 71 -

Upon any such sale, assignment, transfer or grant, the Lender shall have no further obligation hereunder with respect to such interest. Upon any such sale, assignment, transfer or grant, the granting Lender, the new Lender, Powell and the Borrower shall execute and deliver an assignment agreement. The Borrower shall not assign its rights or obligations hereunder without the prior written consent of the Lender, such consent not to be unreasonably withheld. Powell shall not assign its rights or obligations hereunder without the prior written consent of the Lender.

15.6 <u>Severability</u>

Any provision of this Agreement which is prohibited or unenforceable in any jurisdiction shall not invalidate the remaining provisions hereof and any such prohibition or unenforceability in any jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction.

15.7 Whole Agreement

This Agreement and the other Documents constitute the whole and entire agreement between the parties hereto regarding the subject matter hereof and thereof and cancel and supersede any prior agreements, undertakings, declarations, commitments, representations, written or oral, in respect thereof.

15.8 Amendments and Waivers

Any provision of this Agreement may be amended or waived only if Powell, the Borrower and the Lender so agree in writing. Any waiver or consent shall be effective only in the instance and for the purpose for which it is given.

15.9 Further Assurances

Powell, the Borrower and the Lender shall promptly cure any default by it in the execution and delivery of this Agreement, the other Documents or any of the agreements provided for hereunder to which it is a party. Each of the Borrower and Powell, at its expense, shall promptly execute and deliver to the Lender, upon request by the Lender (acting reasonably), all such other and further deeds, agreements, opinions, certificates, instruments, affidavits, registration materials and other documents reasonably necessary for the Borrower's or Powell's compliance with, or accomplishment of the covenants and agreements of the Borrower or Powell hereunder or more fully to state the obligations of the Borrower or Powell as set out herein or to make any registration, recording, file any notice or obtain any consent, all as may be reasonably necessary or appropriate in connection therewith.

15.10 Attornment

The parties hereto each hereby attorn and submit to the jurisdiction of the courts of the Province of Alberta in regard to legal proceedings relating to the Documents. For the purpose of all such legal proceedings, this Agreement shall be deemed to have been performed in the Province of Alberta and the courts of the Province of Alberta shall have jurisdiction to entertain any action arising under this Agreement. Notwithstanding the foregoing, nothing in this Section shall be construed nor operate to limit the right of any party hereto to commence any action relating hereto in any other jurisdiction, nor to limit the right of the courts of any other jurisdiction to take jurisdiction over any action or matter relating hereto.

15.11 Time of the Essence

Time shall be of the essence of this Agreement.

15.12 Credit Agreement Governs

In the event of any conflict or inconsistency between the provisions of this Agreement and the provisions of the other Documents, the provisions of this Agreement, to the extent of the conflict or inconsistency, shall govern and prevail.

15.13 <u>Counterparts</u>

This Agreement may be executed in any number of counterparts, each of which shall be deemed to be an original and all of which taken together shall be deemed to constitute one and the same instrument, and it shall not be necessary in making proof of this Agreement to produce or account for more than one such counterpart. Delivery of an executed counterpart of a signature page of this Agreement by telecopy or by sending a scanned copy by electronic mail shall be effective as delivery of a manually executed counterpart of this Agreement.

IN WITNESS WHEREOF the parties hereto have executed this Agreement.

POWELL CANADA INC.

By: <u>/s/ Don R. Madison</u> Name: Don R. Madison Title: President

By: /s/ Milburn E. Honeycutt Name: Milburn E. Honeycutt Title: Vice President

NEXTRON LIMITED

- By: /s/ Don R. Madison Name: Don R. Madison Title: President
- By: /s/ Milburn E. Honeycutt Name: Milburn E. Honeycutt Title: Vice President

HSBC BANK CANADA

- By: /s/ Aaron Spells Name: Aaron Spells Title: Assistant Vice President
- By: /s/ Kam Thind Name: Kam Thind Title: Assistant Vice President

POWELL INDUSTRIES, INC.

- By: /s/ Don R. Madison Name: Don R. Madison Title: President
- By: /s/ Milburn E. Honeycutt Name: Milburn E. Honeycutt Title: Vice President

PPC TECHNICAL SERVICES INC.

- By: /s/ Don R. Madison Name: Don R. Madison Title: President
- By: /s/ Milburn E. Honeycutt

Name: Milburn E. Honeycutt Title: Vice President

- 73 -

SCHEDULE "A"

FORM OF COMPLIANCE CERTIFICATE

TO: HSBC BANK CANADA

DATE:

3.

- 1. Reference is made to the amended and restated credit agreement made as of between Powell Canada Inc., Powell Industries, Inc. and HSBC Bank Canada and relating to the establishment of certain credit facilities in favour of the Borrower (as amended, modified, supplemented or restated, the "**Credit Agreement**"). Unless otherwise expressly defined herein, capitalized terms set forth in this Conversion Notice shall have the respective meanings set forth in the Credit Agreement.
- 2. This Compliance Certificate is delivered to the Lender pursuant to Section 10.1(e)(iii) of the Credit Agreement.

The undersigned, , of Powell Industries, Inc. ("Powell"), hereby certifies that, as of the date of this Compliance Certificate, I have made or caused to be made such investigations as are necessary or appropriate for the purposes of this Compliance Certificate and:

- (a) the consolidated financial statements for the [fiscal quarter OR fiscal year] ending , , provided to the Lender pursuant to Section 10.1(e) of the Credit Agreement were prepared in accordance with GAAP and present fairly, in all material respects, the consolidated financial position of Powell and its Subsidiaries as at the date thereof;
- (b) the representations and warranties made by Powell and the Borrower in Section 9.1 of the Credit Agreement are true and accurate in all respects as at the date hereof, except as has heretofore been notified to the Lender by the Borrower in writing **[or except as described in Exhibit hereto]**;
- (c) no event has occurred or is continuing which would constitute a Default or Event of Default, except as has heretofore been notified to the Lender by the Borrower in writing **[or except as described in Exhibit hereto]**;
- (d) as at the Quarter End ending , , , the Debt to EBITDA Ratio of Powell was :1.0; attached hereto as Exhibit A is a determination of the Debt to EBITDA Ratio as at the end of the aforementioned Quarter End, together with particulars of each of the definitions and elements included in the determination thereof;
- (e) as at the aforementioned Quarter End, the Fixed Charge Coverage Ratio of Powell was :1.0; attached hereto as Exhibit B is a determination of the Fixed Charge Coverage Ratio as at the aforementioned Quarter End, together with particulars of each of the definitions and elements included in the determination thereof;

(f)	as at the aforementioned Quarter End, the Consolidated Tangible Net Worth of Powell was U.S.\$; attached hereto as Exhibit C is a determination of the Consolidated Tangible Net Worth as at the aforementioned Quarter End together with particulars of each of the definitions at elements included in the determination thereof;						
(g)	no Default or Event of Default has occurred and is continuing [except as described in Exhibit hereto].						
against or a	I give this Compliance Certificate on behalf of Powell and in my capacity as the of Powell and no personal liability is a against or assumed by me in the giving of this Certificate.						
	Dated at	,	this	day of	,		

- 2 -

Name: Title:

SCHEDULE "B"

CONVERSION NOTICE

TO: HSBC BANK CANADA

DATE:

- 1. This Conversion Notice is delivered to you pursuant to the terms and conditions of the amended and restated credit agreement made as of between Powell Canada Inc., Powell Industries, Inc. and HSBC Bank Canada and relating to the establishment of certain credit facilities in favour of the Borrower (as amended, modified, supplemented or restated, the "**Credit Agreement**"). Unless otherwise expressly defined herein, capitalized terms set forth in this Conversion Notice shall have the respective meanings set forth in the Credit Agreement.
- 2. The Borrower hereby requests a Conversion as follows:
 - (h) Conversion Date:
 - (i) Applicable Credit Facility: **Revolving Facility**
 - (j) Conversion of the following Loans:
 - (i) Type of Loan
 - (ii) Amount being converted:
 - (iii) Interest Period maturity (for Bankers' Acceptances):
 - INTO the following Loan:
 - (i) Type of Loan:
 - (ii) Interest Period (specify term of Bankers' Acceptances):
 - (k) Payment, delivery or issuance instructions (if any):

- 3. The undersigned certifies to the Lender that:
 - (a) each of the representations and warranties contained in Section 9.1 of the Amended and Restated Credit Agreement shall be true, correct and complete on and as of the Conversion Date to the same extent as though made on and as of such date (except to the extent such representations and warranties specifically relate to an earlier date, in which case such representations and warranties shall have been true, correct and complete on and as of such earlier date); and
 - (b) on the date hereof, no Default or Event of Default has occurred and is continuing and no Default or Event of Default will occur as a result of the making of the Conversion contemplated herein.

Yours very truly,

POWELL CANADA INC.

Per:

Name: Title:

Per:

Name: Title:

- 2 -

SCHEDULE "C"

DRAWDOWN NOTICE

TO: HSBC Bank Canada

DATE:

 This Drawdown Notice is delivered to you pursuant to the terms and conditions of the amended and restated credit agreement made as of between Powell Canada Inc., Powell Industries, Inc. and HSBC Bank Canada and relating to the establishment of certain credit facilities in favour of the Borrower (as amended, modified, supplemented or restated, the "Credit Agreement"). Unless otherwise expressly defined herein, capitalized

terms set forth in this Drawdown Notice shall have the respective meanings set forth in the Credit Agreement.

- 2. The Borrower hereby requests a Drawdown as follows:
 - (a) Drawdown Date:
 - (b) Applicable Credit Facility: Revolving Facility
 - (c) Amount of Drawdown:
 - (d) Type of Loan:
 - (e) Interest Period (specify term for Bankers' Acceptances):
 - (f) Payment, delivery or issuance instructions (if any):

- 3. The undersigned certifies to the Lender that:
 - (a) each of the representations and warranties contained in Section 9.1 of the Amended and Restated Credit Agreement shall be true, correct and complete on and as of the Drawdown Date to the same extent as though made on and as of such date (except to the extent such representations and warranties specifically relate to an earlier date, in which case such representations and warranties shall have been true, correct and complete on and as of such earlier date); and
 - (b) on the date hereof, no Default or Event of Default has occurred and is continuing and no Default or Event of Default will occur as a result of the making of the Drawdown contemplated herein; and
 - (c) after giving effect to the foregoing Advances, the Outstanding Principal of all Loans outstanding under the **Revolving Facility** will not exceed the maximum amount of such Credit Facility.

Yours very truly,

POWELL CANADA INC.

Per:

Name: Title:

Per:

Name: Title:

- 2 -

SCHEDULE "D"

ROLLOVER NOTICE

TO: HSBC Bank Canada

DATE:

- 1. This Rollover Notice is delivered to you pursuant to the terms and conditions of the amended and restated credit agreement made as of between Powell Canada Inc., Powell Industries, Inc. and HSBC Bank Canada and relating to the establishment of certain credit facilities in favour of the Borrower (as amended, modified, supplemented or restated, the "**Credit Agreement**"). Unless otherwise expressly defined herein, capitalized terms set forth in this Rollover Notice shall have the respective meanings set forth in the Credit Agreement.
- 2. The Borrower hereby requests a Rollover as follows:
- (a) Rollover Date:
- (b) Applicable Credit Facility: **Revolving Facility**
- (c) Type of Loan:
- (d) Amount of Rollover:
- (e) New Interest Period (specify term):
- (f) Payment, delivery or issuance instructions (if any):

- 3. The undersigned certifies to the Lender that:
 - (a) each of the representations and warranties contained in Section 9.1 of the Amended and Restated Credit Agreement shall be true, correct and complete on and as of the Rollover Date to the same extent as though made on and as of such date (except to the extent such representations and warranties specifically relate to an earlier date, in which case such representations and warranties shall have been true, correct and complete on and as of such earlier date); and
 - (b) on the date hereof, no Default or Event of Default has occurred and is continuing and no Default or Event of Default will occur as a result of the making of the Rollover contemplated herein; and

Yours very truly,

POWELL CANADA INC.

Per:

Name:

Title:

Per:

Name: Title:

SCHEDULE "E"

DISCLOSURE SCHEDULE

THIS DISCLOSURE SCHEDULE is made as of

1. <u>Corporate Organization Chart of Powell Industries, Inc. and the Subsidiaries</u>

• See Appendix "A" attached hereto.

2. <u>Powell Industries, Inc.</u>

A. Corporate Information:

- (i) Jurisdiction of Incorporation:
- (ii) Date of Incorporation:
 - (a) Certificate of Incorporation:
 - (b) Certificate of Amendment(s):
- (iii) Share Capital and Other Ownership Interests:
 - (a) Issues Shares:
 - (b) Shareholders:
- (iv) Directors and Officers:
 - (a) Directors:
 - (b) Officers:

- B. Business Information:
 - (i) Chief Executive Office Location:
 - (ii) Jurisdiction of Conduct of Business:
 - (iii) Locations where Inventory or other Assets (other than Real Property) are held:
- C. Real Property and Leases Relating to Real Property:
 - (i) Real Property:
 - (ii) Leases:
- D. Trademarks, Patents, and Other Material Intellectual Property:
 - (i) Trade Names:
 - (ii) Trademarks:
 - (iii) Patents:

3. Powell Canada Inc.

- A. Corporate Information:
 - (i) Jurisdiction of Incorporation:
 - Canada
 - (ii) Date of Incorporation:
 - (a) Certificate of Incorporation:
 - (b) Certificate of Amendment(s):
 - (iii) Share Capital and Other Ownership Interests:
 - (c) Issues Shares:
 - (d) Shareholders:
 - Powell Industries, Inc. 100%
 - (iv) Directors and Officers:
 - (c) Directors:
 - (d) Officers:
- B. Business Information:
 - (i) Chief Executive Office Location:
 - •
 - (ii) Jurisdiction of Conduct of Business:
 - Canada
 - (iii) Locations where Inventory or other Assets (other than Real Property) are held:

- 3 -

- C. Real Property and Leases Relating to Real Property:
 - (i) Real Property:
 - (ii) Leases:
- D. List of Material Agreement:

(i)

- E. Trademarks, Patents, and Other Material Intellectual Property:
 - (i) Trade Names:
 - (ii) Trademarks:
 - (iii) Patents:

- 4 -

CERTIFICATION

I, Thomas W. Powell, certify that:

1. I have reviewed this Quarterly Report on Form 10-Q of Powell Industries, Inc.;

2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;

3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;

4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:

a) designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;

b) designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;

c) evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and

d) disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and

5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):

a) all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and

b) any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

/s/ Thomas W. Powell

Thomas W. Powell Chairman of the Board President and Chief Executive Officer (Principal Executive Officer)

CERTIFICATION

I, Don R. Madison, certify that:

1. I have reviewed this Quarterly Report on Form 10-Q of Powell Industries, Inc.;

2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;

3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;

4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:

a) designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;

b) designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;

c) evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and

d) disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and

5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):

a) all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and

b) any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

/s/ Don R. Madison

Don R. Madison Executive Vice President Chief Financial and Administrative Officer (Principal Financial Officer)

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with this Quarterly Report (the "Report") on Form 10-Q of Powell Industries, Inc. (the "Company") for the quarter ended March 31, 2012, as filed with the Securities and Exchange Commission on the date hereof, I, Thomas W. Powell, Chief Executive Officer of the Company, certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that:

(1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and

(2) The information contained in the Report fairly represents, in all material respects, the financial condition and results of operations of the Company.

/s/ Thomas W. Powell

Thomas W. Powell Chairman of the Board President and Chief Executive Officer

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with this Quarterly Report (the "Report") on Form 10-Q of Powell Industries, Inc. (the "Company") for the quarter ended March 31, 2012, as filed with the Securities and Exchange Commission on the date hereof, I, Don R. Madison, Executive Vice President and Chief Financial Officer of the Company, certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that:

(1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and

(2) The information contained in the Report fairly represents, in all material respects, the financial condition and results of operations of the Company.

/s/ Don R. Madison

Don R. Madison Executive Vice President Chief Financial and Administrative Officer